

NATIONAL CENTER FOR EARTHQUAKE ENGINEERING RESEARCH

State University of New York at Buffalo



Pipeline Replacement Feasibility Study: A Methodology for Minimizing Seismic and Corrosion Risks to Underground Natural Gas Pipelines

by

R.T. Eguchi, H.A. Seligson and D.G. Honegger

EQE International Lakeshore Towers 18101 Von Karman Avenue, Suite 400 Irvine, California 92715

Technical Report NCEER-95-0005

March 2, 1995

This research was conducted at EQE International and was partially supported by the National Science Foundation under Grant No. BCS 90-25010 and the New York State Science and Technology Foundation under Grant No. NEC-91029.

NOTICE

This report was prepared by EQE International as a result of research sponsored by the National Center for Earthquake Engineering Research (NCEER) through grants from the National Science Foundation, the New York State Science and Technology Foundation, and other sponsors. Neither NCEER, associates of NCEER, its sponsors, EQE International, nor any person acting on their behalf:

- a. makes any warranty, express or implied, with respect to the use of any information, apparatus, method, or process disclosed in this report or that such use may not infringe upon privately owned rights; or
- b. assumes any liabilities of whatsoever kind with respect to the use of, or the damage resulting from the use of, any information, apparatus, method or process disclosed in this report.

Any opinions, findings, and conclusions or recommendations expressed in this publication are those of the author(s) and do not necessarily reflect the views of NCEER, the National Science Foundation, the New York State Science and Technology Foundation, or other sponsors.



NATIONAL CENTER FOR EARTHQUAKE ENGINEERING RESEARCH

State University of New York at Buffalo

PB95-252326

Pipeline Replacement Feasibility Study: A Methodology for Minimizing Seismic and Corrosion Risks to Underground Natural Gas Pipelines

by

R.T. Eguchi, H.A. Seligson and D.G. Honegger EQE International Lakeshore Towers 18101 Von Karman Avenue, Suite 400 Irvine, California 92715

Technical Report NCEER-95-0005

REPRODUCED BY U.S. DEPARTMENT OF COMMERCE NATIONAL TECHNICAL INFORMATION SERVICE SPRINGFIELD, VA 22161

March 2, 1995

This research was conducted at EQE International and was partially supported by the National Science Foundation under Grant No. BCS 90-25010 and the New York State Science and Technology Foundation under Grant No. NEC-91029.

NOTICE

This report was prepared by EQE International as a result of research sponsored by the National Center for Earthquake Engineering Research (NCEER) through grants from the National Science Foundation, the New York State Science and Technology Foundation, and other sponsors. Neither NCEER, associates of NCEER, its sponsors, EQE International, nor any person acting on their behalf:

- a. makes any warranty, express or implied, with respect to the use of any information, apparatus, method, or process disclosed in this report or that such use may not infringe upon privately owned rights; or
- b. assumes any liabilities of whatsoever kind with respect to the use of, or the damage resulting from the use of, any information, apparatus, method or process disclosed in this report.

Any opinions, findings, and conclusions or recommendations expressed in this publication are those of the author(s) and do not necessarily reflect the views of NCEER, the National Science Foundation, the New York State Science and Technology Foundation, or other sponsors.



Pipeline Replacement Feasibility Study:

A Methodology for Minimizing Seismic and Corrosion Risks to Underground Natural Gas Pipelines

by

R.T. Eguchi¹, H.A. Seligson² and D.G. Honegger³

March 2, 1995

Technical Report NCEER-95-0005

NCEER Task Numbers 93-7301B, 92-3601B and 91-3541B

NSF Master Contract Number BCS 90-25010 and NYSSTF Grant Number NEC-91029

> also prepared for Southern California Gas Company Los Angeles, California

- 1 Vice President, EQE International
- 2 Staff Engineer, EQE International
- 3 Principal Engineer, EQE International

NATIONAL CENTER FOR EARTHQUAKE ENGINEERING RESEARCH State University of New York at Buffalo Red Jacket Quadrangle, Buffalo, NY 14261

PREFACE

The National Center for Earthquake Engineering Research (NCEER) was established to expand and disseminate knowledge about earthquakes, improve earthquake-resistant design, and implement seismic hazard mitigation procedures to minimize loss of lives and property. The emphasis is on structures in the eastern and central United States and lifelines throughout the country that are found in zones of low, moderate, and high seismicity.

NCEER's research and implementation plan in years six through ten (1991-1996) comprises four interlocked elements, as shown in the figure below. Element I, Basic Research, is carried out to support projects in the Applied Research area. Element II, Applied Research, is the major focus of work for years six through ten. Element III, Demonstration Projects, have been planned to support Applied Research projects, and will be either case studies or regional studies. Element IV, Implementation, will result from activity in the four Applied Research projects, and from Demonstration Projects.



Research tasks in the Lifeline Project evaluate seismic performance of lifeline systems, and recommend and implement measures for mitigating the societal risk arising from their failures or disruption caused by earthquakes. Water delivery, crude oil transmission, gas pipelines, electric power and telecommunications systems are being studied. Regardless of the specific systems to be considered, research tasks focus on (1) seismic vulnerability and strengthening; (2) repair and restoration; (3) risk and reliability; (4) disaster planning; and (5) dissemination of research products.

The end products of the **Lifeline Project** will include technical reports, computer codes and manuals, design and retrofit guidelines, and recommended procedures for repair and restoration of seismically damaged systems. The **structures and systems program** constitutes one of the important areas of research in the **Lifelines Project**. Current tasks include the following:

- 1. Continued testing of lightly reinforced concrete external joints.
- 2. Continued development of analytical tools, such as system identification, idealization, and computer programs.
- 3. Perform parametric studies of building response.
- 4. Retrofit of lightly reinforced concrete frames, flat plates and unreinforced masonry.
- 5. Enhancement of the IDARC (inelastic damage analysis of reinforced concrete) computer program.
- 6. Research infilled frames, including the development of an experimental program, development of analytical models and response simulation.
- 7. Investigate the torsional response of symmetrical buildings.

This report presents a methodology which a utility can use to fold mitigation for seismic hazards into its ongoing repair and replacement program. The methodology was developed specifically for buried pipeline components within the Southern California Gas Company (SoCalGas) system. Both transmission and distribution pipeline systems are considered; however, suggested procedures differ, due in part to the importance and relative lack of redundancy (i.e., interconnectedness) for transmission pipe.

In the past, the SoCalGas repair and replacement program focused on corrosion damage. The new methodology incorporates potential seismic damage as characterized by areas of potential ground failure. As part of this effort, a new procedure for estimating corrosion leakage rates in "data-poor" areas is proposed.

The report describes realistic mitigation procedures for buried pipeline components which is one of the objectives of NCEER's Lifeline Project.

ABSTRACT

This report completes the review of procedures used by the Southern California Gas Company to optimize decisions on pipeline replacement and repair. In addition to discussions with the Engineering Design Department, meetings were also held with representatives from System Planning, Transmission and Distribution. This study was conducted as a joint effort between EQE International and Cornell University. Partial support for this effort was received from the National Center for Earthquake Engineering Research.

This report is comprised of two major parts: (1) a report that discusses a plan for consideration of seismic and corrosion risks under a common program, and (2) a report that summarizes the development of improved corrosion leakage models for the Southern California Gas Company (Appendix B).

The main conclusions of this report are three-fold:

- 1. It is possible to develop consistent, company-wide а pipe repair/replacement methodology based on minimizing expected costs from corrosion-related failures, and increasing the seismic resistance and safety of the system. For Transmission, this program is based on refining the delineation of areas of potential ground failure (i.e., liquefaction) that are responsible for the majority of the seismic risk. For Distribution, this program is based on using EPOCH (a computer program developed by Distribution to optimize economic decisions on pipe repair/replacement based on corrosion risks) as a major element; additional criteria are applied afterward to decide whether pipes initially identified for repair should be replaced for seismic hazard mitigation purposes.
- 2. Current methods for predicting pipeline leakage based on corrosion failures appear to be adequate when sufficient repair data are available. Improvements can be made by incorporating the "age dependent" model developed in this study that allows for prediction of leakage rates based on one or two data points. An analysis of repair data found that a key parameter in establishing future corrosion leak rates is the age at which the first leak is discovered. When spurious data were removed from the

۷

data set, it was observed that the rate of increase of leak rates increases with the age at first leak. By incorporating this new parameter, the assessment of future leakage can be expanded to include more pipe, such as an additional 19 percent in the pilot study. It must be cautioned, however, that the analysis performed in this study was for a small area of the total system. Further investigation of other areas would need to be performed in order to verify whether the trends observed in this study are general trends.

3. A number of recommendations are made regarding further steps for this study. The most important recommendation is to extend parts of the Feasibility study so that (1) a procedure for integrating seismic and corrosion risks for distribution pipelines can be tested for a small area of the system, and (2) the details of a more integrated, interdepartmental program can be developed and tested.

<u>Section</u>	Title			Page
1.	INTRO	NTRODUCTION		
2.	INTER	INTERACTION WITH SOUTHERN CALIFORNIA GAS		
	COMP	PANY PERSONNEL		
	2.1	Projec	t Meetings	2-1
	2.2	Inform	ation Supplied by SoCalGas	2-1
	2.3	Outco	me of Interaction	2-4
3.	PERCE	EIVED IS	SSUES AND CONCERNS	3-1
	3.1	Repair	vs. Replacement	3-1
		3.1.1	Distribution - "Repair vs. Replace"	3-2
		3.1.2	Transmission - Prioritized Replacement	3-2
	3.2	Seismi	ic vs. Non-Seismic Risks	3-3
		3.2.1	Distribution	3-3
		3.2.2	Transmission	3-4
	3.3	Plannir	ng Time Frames	3-6
4.	GENE	RAL ME	THODOLOGY FOR PIPELINE	
	REPLA		IT DECISIONS	4-1
	4.1	Pipelin	e Replacement Methodology for Transmission	4-1
		4.1.1	Review Pipeline Data	4-1
		4.1.2	Review System Planning Department's	
			Long Term Goals	4-3
		4.1.3	Assess Seismic Hazard and Impact of Outage	4-3
		4.1.4	Assess Other Non-Seismic Risks	4-5
		4.1.5	Select Replacement Strategy	4-5
		4.1.6	Calculate Replacement Costs	4-6
		4.1.7	Calculate Priority Factor	4-6

TABLE OF CONTENTS

TABLE OF CONTENTS (CONTINUED)

<u>Section</u>	Title		
	4.2	Pipeline Replacement Methodology for Distribution	4-6
		4.2.1 Review Planning Data	4-8
		4.2.2 Run EPOCH	4-8
		4.2.3 Post-EPOCH Analysis	4-9
5.	REFIN	ED PREDICTIVE MODELS FOR CORROSION LEAKAGE	5-1
	5.1	Pilot Study Data	5-1
	5.2	Assessment of Linear Regression Prediction	
		Technique	5-2
	5.3	Alternative Leak Prediction Methodology -	
		the Age Dependent Model	5-2
	5.4	Conclusions and Recommendations from the	
		Pilot Study	5-6
6.	RECO	MMENDED REFINEMENTS FOR IMPROVED DECISION-	
	MAKI	NG CAPABILITIES	6-1
	6.1	Mapping and Records	6-1
	6.2	Interaction with System Planning	6-2
	6.3	Incorporation of Seismic Risks	6-2
7.	REFEF	RENCES	7-1

APPENDICES

- A Explanation of Least Squares Method
- B Results of Distribution Study
- C Index Maps to Atlas Sheets and Pipeline Segment Database
- D Lead Database

LIST OF TABLES

		<u>Page</u>
2-1:	Summary of Project Meetings	2-2

LIST OF FIGURES

4-1:	General Methodology for Pipeline Replacement Decisions	4-2
4-2:	Procedure for Incorporating Seismic Hazard Data into Pipeline Repair/Replacement Decisions for Distribution	4-7
5-1:	Comparison of Leak Rate Prediction from Linear Regression of Data Through 1986 to Actual Post-1986 Performance (from Torrance Area Pilot Study)	5-4
5-2:	Age Dependent Leak Rate Model for Steel Pipe Installed in 1912 (from Torrance Area Pilot Study)	5-5
5-3:	Comparison of Leak Rates Predicted from Age Dependent Model to Actual Post-1986 Performance (from Torrance Area Pilot Study)	5-7
5-4:	Comparison of Predicted Leak Rates from the Dependent Model to Actual Post-1986 Performance for Pipes with Limited Leakage (from Torrance Area Pilot Study)	5-8

SECTION 1 INTRODUCTION

As the cost environment within which gas utilities operate becomes more competitive, issues such as pipe repair versus pipe replacement become more important. Decisions to automatically replace generalized categories of pipe without a cost/benefit analysis are rapidly becoming obsolete, primarily because of excessive costs. At the same time, the ability to assess future risks associated with pipeline failure are becoming more keenly developed. The use of wellmanaged databases has allowed utility companies to examine in detail the probability or likelihood of experiencing certain kinds of pipe failures. This information, combined with an assessment of the impact of these failures, has allowed gas utilities to compare the benefits and risks of adopting alternative strategies for pipe repair and replacement. In general, the actual strategy for repair or replacement will depend on the particular characteristics of the utility and the goals of their replacement program.

In June of 1992, the Southern California Gas Company (SoCalGas) authorized EQE International and Cornell University to perform an independent assessment of current SoCalGas pipe repair and replacement strategies. The primary focus of this program was on risk assessment, and economic and safety issues related to pipe repair/replacement. Because a large part of this effort dealt with an analysis of risks associated with earthquakes, partial funding for this project was provided by the National Center for Earthquake Engineering Research (NCEER).

The intent of this study is to help SoCalGas develop a company-wide strategy for replacement of current and future steel pipelines within its system. To accomplish this, the study has been divided into two phases: feasibility and implementation. This particular report summarizes the feasibility phase in which a general methodology for making decisions regarding steel pipeline integrity has been developed. In addition, a portion of this methodology has been applied to a small area within the SoCalGas system. Whereas the Cornell report emphasizes pipe repair/replacement for transmission and distribution supply lines, the EQE report concentrates on distribution pipelines. In the implementation phase, the general methodology will be refined, tested for additional portions of the system, and

implemented with SoCalGas engineers as a continuing program to assess pipeline conditions and set cost-effective priorities for steel pipeline repair and replacement.

The following tasks were performed to meet the objectives of the feasibility phase:

- 1. Review current programs being used by the Transmission and Distribution Departments of SoCalGas for prioritizing pipe repair and replacement.
- 2. Recommend improvements in existing methodologies, and suggest ways of integrating these improvements to form a more consistent, systematic approach to planning.
- 3. Review and refine risk models to reflect more accurately the risks associated with different pipe repair and replacement strategies.
- Suggest ways of utilizing more detailed information on the location of potential liquefaction areas in estimating future earthquake risks to pipelines.
- 5. Develop a framework for a comprehensive system integrity and pipe repair/replacement methodology.

In general, these tasks parallel those completed by Cornell University for transmission and distribution supply pipelines.

This report is organized into six major sections, including this introduction. Section 2 discusses the interaction of the project team with SoCalGas personnel. A number of meetings were held between EQE, Professor T.D. O'Rourke of Cornell University and SoCalGas personnel from the Engineering Design, System Planning, Transmission and Distribution Departments to discuss the objectives and status of the various tasks. Section 3 identifies important issues and concerns regarding the development of a system-wide pipe repair/replacement program. Issues such as pipe repair vs. replacement, seismic vs. non-seismic risks, and short-term vs. long-term planning are discussed. Section 4 discusses a general methodology for combining the risks from various hazards or effects. Separate discussions are given for transmission and distribution systems. Section 5 presents the major contribution of the feasibility study, i.e., the development of improved corrosion leakage models. Using data from a portion of the SoCalGas system, an analysis

was made of current techniques for predicting future leaks due to corrosion. As a result of this analysis, several recommendations were made with regard to better use of the data and new models for estimating corrosion related leakage. The results of this corrosion study are documented in detail in a separate report that is attached here as an appendix. Finally, Section 6 recommends future steps in this study. One important recommendation is to extend the feasibility study to allow for a more complete integration of risks (i.e., earthquake and corrosion) in establishing pipe repair and replacement priorities. This analysis would be applied to the same pilot area selected for the corrosion analysis.

SECTION 2

INTERACTION WITH SOUTHERN CALIFORNIA GAS COMPANY PERSONNEL

An important part of this project was the interaction between the project team and SoCalGas personnel. This interaction was important in understanding the priorities assigned by each SoCalGas department to each of the project tasks. Additionally, this interaction was used to refine the objectives and scope of critical tasks. The following subsections discuss the meetings that were held throughout the project, information that was received from SoCalGas, and the outcome of this interaction.

2.1 PROJECT MEETINGS

Numerous project meetings were held to solicit input from the various SoCalGas departments. Meetings held early in the project schedule (August 5 and September 10, 1992) to help define the focus of the study included representatives from Distribution, Engineering, Planning and Transmission. Several interim meetings (September 29, and October 12, 1992) emphasized the pilot distribution system corrosion leakage analysis. Later meetings summarized preliminary project results and functioned as project status updates (November 10 and December 23, 1992, and January 19, 1993). Table 2-1 summarizes the general purpose of each meeting, and lists EQE, SoCalGas, and Cornell personnel in attendance.

2.2 INFORMATION SUPPLIED BY SOCALGAS

During the course of the project meetings, various documents and studies prepared by SoCalGas were identified for our review. These documents described current programs and methodologies used by SoCalGas to prioritize pipeline replacement. These include:

- <u>Value Chain Analysis of the Pre-WWII Transmission Pipeline</u>
 <u>Replacement Program</u>, SoCalGas (Transmission), November,
 1991
- <u>Underground Piping System Replacement Assessment</u>, G.E. Strang, SoCalGas (Engineering), May, 1986

Table 2-1

SUMMARY OF PROJECT MANAGERS

DATE	GENERAL PURPOSE	MEETING LOCATION	EOE	CORNELL	SoCalGas*
8/5/92	Project Kick-Off Meeting Orientation to SoCalGas Issues	SoCalGas - LA	Eguchi Honegger	O'Rourke	Ackart (T) Becker-Castle (R) Butler (D) Conley (D) Constantine (E) Gailing (T) Haynes (E) Moore (E) Nose' (E) Saad (T) Sam (E) Stevens (E) Wellman (D)
9/10/92	Follow-up to Kick-Off Meeting - Refinement of Issues/Scope	SoCalGas - LA	Eguchi Honegger Seligson		Conley (D) Hammer (D) Mansdorfer (T) McNorgan (E) Stevens (E) Wellman (E)
9/29/92	Meeting to collect data for pilot study (corrosion leakage)	SoCalGas - Torrance Division	Eguchi Honegger Seligson Shu		Blood (D) Conley (D) Hammer (D) Jordan (D) Moore (E)
10/12/92	Follow-up meeting to refine study area data	SoCalGas - Torrance Division	Seligson Shu		Blood (D) Hammer (E)
11/10/92	Project Team Meeting - Status Report	EQE - Irvine	Eguchi Honegger Seligson Shu	O'Rourke	McNorgan (E) Moore (E)
12/23/92	Presentation of preliminary results for pilot study	SoCalGas - LA	Eguchi Seligson Shu		Becker-Castle (R) Conley (D) Dowell (E) Gailing (E) Haynes (E) Madariage (E) Moore (E)
1/19/93	Discussion for report outline, Confirmation of deliverables	EQE - Irvine	Eguchi Seligson Shu		Dowell (E)

Table 2-1 (Continued) SUMMARY OF PROJECT MANAGERS

DATE	GENERAL PURPOSE	MEETING LOCATION	EQE	CORNELL	SoCalGas*
7/8/93		SoCalGas - LA	Eguchi Seligson	O'Rourke	Becker-Castle (R) Conley (D) Gailing (E) Haynes (E) McNorgan (E) Stevens (E)

* Note:	(D) = Distribution	(E) = Engineering	(P) = Planning
	(R) = Research	(T) = Transmission	

- "Engineering Report, Special Pipeline Replacement Program, -1994 Rate Case", SoCalGas (Engineering), June, 1992.
- A sample data sheet and flow chart describing EPOCH (Efficient Pipeline Operation in a Competitive Habitat), the economic repair-replace decision-making program under development by Distribution.
- Annual report for calendar year 1992, Gas Distribution System.

In addition, other reference material identified by Transmission staff was provided:

- J.F. Kiefner and P.H. Vieth (1991), "Methods for Prioritizing Pipeline Maintenance and Rehabilitation", <u>Pipeline Risk</u> <u>Assessment, Rehabilitation and Repair Conference</u>.
- W.E. Martinsen and J.B. Cornwell (1991), "Use and Misuse of Historical Pipeline Failure Data", <u>Pipeline Risk Assessment</u>, <u>Rehabilitation and Repair Conference</u>.
- W.K. Muhlbauer (1991), Dow Chemical Company, "RIPS a Pipeline Safety Evaluation System", <u>Pipeline Risk Assessment</u>, <u>Rehabilitation and Repair Conference</u>.
- N.A. Townsend and G.B. Fearnehough (1986), British Gas Corporation, "Controlling Risk From U.K. Gas Transmission Pipelines", <u>7th Symposium on Line Pipe Research</u>, American Gas Association.

2.3 OUTCOME OF INTERACTION

As a result of the discussions with various SoCalGas personnel and review of relevant background material, the project team was able to:

 identify the general framework within which any SoCalGas pipeline replacement program must operate

- understand replacement programs and strategies currently in place at SoCalGas, and evaluate current risk assessment techniques
- identify issues of importance to the various departments (see Section 3.0)
- identify operational differences between distribution and transmission that might impact implementation of a uniform pipe replacement strategy (see Section 3.1)
- understand the linkages between Distribution, Transmission, and System Planning (see Section 3.3)
- identify areas where the project team might make significant contributions to existing procedures (see Section 5.0).

One of the more important outcomes of this interaction was the decision to refocus development efforts from economic modelling to risk assessment. During the initial stages of this project, it was pointed out that a significant internal effort was being undertaken by Distribution to develop a computer program capable of making decisions regarding pipe repair or replacement based on economic considerations. Because of the proprietary nature of that program, few details were provided to the project team on the algorithms used to forecast leaks caused by corrosion or the methods used to calculate costs and benefits. As a result, it was collectively decided that EQE's efforts should focus on an independent development of corrosion leakage models, and that recommendations be provided on how best to utilize the repair data available to SoCalGas engineers.

In order to provide some guidance to SoCalGas on how current methods for pipe repair/replacement can be integrated with methods that focus on seismic risks, a general methodology has been developed. This methodology, discussed in Section 4.0, emphasizes a prioritized replacement program for transmission and distribution supply lines, and an optimization program for pipe repair/replacement for Distribution built around the current Distribution corrosion program EPOCH.

ł ł. ł.

SECTION 3 PERCEIVED ISSUES AND CONCERNS

In the course of this project, several issues relevant to the development of a consistent replacement strategy were identified. It was clear from discussions with SoCalGas personnel, that the issue of pipe repair and replacement was perceived differently by the various departments. The departments that are directly impacted by the issue of pipe repair/replacement include:

- System Planning
- Transmission
- Distribution
- Engineering Design

Addressing the pipe repair/replacement problem from a company-wide basis requires an understanding of the relevant issues for each department. It is possible that issues affecting one department may, in fact, not be considered significant by the other departments. One reason for these differences may be the level of risk associated with pipeline failure. The risk resulting from failures on transmission lines, for example, may be considered more significant than a distribution main failure, thus necessitating a stronger safety component. Another reason for implementing different replacement strategies may relate to the number and frequency of repairs made on each system. Because numerous repairs are made to the distribution system each year, the cost to maintain the system becomes a critical factor. Therefore, strategies to reduce the overall economic cost of maintaining the system become more important.

The following discussions underscore some of the major issues that must be addressed to formulate a company-wide approach to pipe repair/replacement. In general, these issues fall into three categories: pipe repair versus pipe replacement, seismic versus non-seismic risks, and varying time frames for planning.

3.1 REPAIR VS. REPLACEMENT

In order to develop a single pipeline replacement methodology that would be applicable to both transmission and distribution, it is instructive to identify our

3-1

understanding of operational differences with respect to pipeline replacement that could impact implementation.

3.1.1 Distribution - "Repair vs. Replace"

Routine repair/replacement decisions for Distribution piping are generally reactive -Distribution responds when a leak is reported or found as part of scheduled surveys. Serious leaks (Codes 1 and 2) are repaired immediately, or within two weeks, while less serious leaks (Code 3) that require action within one year, become part of the "Repair vs. Replace" decision-making process. That is, pipeline replacement is only an option for pipeline segments with Code 3 leakage pending. Currently, SoCalGas is in the process of implementing EPOCH, a computer program which quantifies the cost of repair and replacement alternatives for Distribution piping. The results of EPOCH provide the repair/replace decision, as well as a basis for prioritization of projects.

In response to CPUC suggestions, a limited number of distribution pipeline classes have been identified for inclusion in a special pipeline replacement program.

> Attention by the CPUC Staff to "reportable incidents" involving main or service failures has increased since 1980. The Staff has suggested planned removal of "families" of gas facilities unless the company demonstrates that the "reportable" incident involved unusual conditions unlikely to be repeated in the future (Strang, 1986).

Classes slated for replacement include certain plastic services, copper mains and services, cast iron mains, bare steel main in conduit, and Pre-World War II supply lines in urban areas. Replacement priorities have been set according to safety concerns, continuity of service and certain economic factors, such as "... prevention of cost from incidents, judgements and assessments" (Strang, 1986)

3.1.2 Transmission - Prioritized Replacement

For high pressure transmission lines, any failure is a significant incident, and not a simple leak. Failures are promptly repaired, and replacement is not a viable option in response to this type of failure. In other words, there is no "routine" pipeline replacement program for transmission pipelines. Replacements are performed on a

systematic basis; certain vulnerable or mechanically deficient classes of pipe, or pipe in areas of perceived seismic hazard have been identified and are scheduled for long-term replacement. These replacement programs have been authorized by the CPUC. With regard to pre-World War II pipe, the Value Chain Analysis report stated,

As of the early 1980's... it was recognized that the condition of some of this pipe had deteriorated. Also, several reports by consultants indicated that the poor weld quality in pipelines built prior to WWII made them significantly more susceptible to failures during earthquakes. As a result, the Company sought and received authorization from the CPUC for capital expenditures over and above traditional levels to fund a special replacement program.

Priorities are set based on relative risk and the level of approved funding. Transmission's approach is, therefore, generally pro-active - replacements are made in anticipation of possible high cost, high impact failures. Other replacements are made in response to planning issues, such as anticipated or actual changing demands.

3.2 SEISMIC VS. NON-SEISMIC RISKS

Because of the difference in the nature of transmission and distribution systems, different risks will dominate, and failures will have different impacts. Pipeline failures are generally attributed to one of several causes: corrosion, third party damage, material failure, construction defects, or seismic loads. These causes may be grouped into predictable and non-predictable failures. Corrosion effects are generally predictable, while third party damage, construction defects and material failure are not. The unpredictable failure modes, are, for the most part, controllable. The damage caused by seismic loads is certainly quantifiable, but the probability of occurrence must be taken into consideration as well.

3.2.1 Distribution

Gas distribution systems are usually extensive, highly netted, highly redundant networks of mostly small diameter, medium pressure pipe. SoCalGas distribution mains are primarily steel (64.8% as of the end of 1992, according to the annual

report for calendar year 1992), with the remainder comprised of plastic (35.1%), and cast iron (0.01%). Distribution supply lines, which essentially function as transmission lines, are treated in this discussion as transmission lines. Because most repairs can be made to low pressure lines while under pressure, leakage or failure of an individual pipe will have little impact on supply to the surrounding area.

The majority of repairs required by the distribution system are caused by corrosion. For example, a detailed study of a small area within the City of Torrance revealed that 70% of the repairs made between 1970 and 1991 were attributed to corrosion. It is reported that

Company-initiated leakage surveys, routine inspection procedures and pipe replacements in advance of public improvements largely identify and control hazardous conditions which develop slowly over time. Serious leaks, which generate immediate hazards, are primarily related to: 1) materials defects which take years to cause failure; 2) accidental damage by outside forces; or 3) significant events initiated by earthquakes (Strang, 1986).

The majority of risks for Distribution i.e., corrosion risks, are being addressed through the routine pipe repair/replacement program. The remaining risks (material deficiencies and earthquake vulnerability) are currently being included qualitatively under the Special Pipeline Replacement Program.

3.2.2 Transmission

Gas transmission systems are generally non-netted, high pressure systems with limited redundancy. SoCalGas transmission pipelines are exclusively made of steel, and are typically large in diameter. The impact of failure of transmission pipelines, as well as distribution supply lines, may not be insignificant. The transmission system transports gas from out of state, to and from storage fields, and from local producers.

The distribution supply system, operating at higher pressures and larger diameters < than the remainder of the distribution system>, is operationally critical to continued supply during peak demands or emergency conditions. In many cases, these supply lines constitute single sources of supply to large areas, many within the central city, i.e. Hollywood, Beverly Hills (Strang, 1986).

Corrosion leakage is not expected to be a problem for SoCalGas transmission lines, because of the high priority placed on cathodic protection and monitoring. Between the years of 1983 and 1990, the transmission system averaged only 24 leaks per year, or 0.007 leaks per mile (or 0.004 leaks per km) of pipe per year (SoCalGas, 1991). This can be contrasted to figures for steel without cathodic protection in the distribution system. For the 12 year period between 1973 and 1984, bare steel in the distribution system averaged 1.18 repairs per mile (or 0.28 repairs per km) per year (estimated from data provided in Strang, 1986). The transmission system suffers corrosion leakage up to 170 times slower than the unprotected steel in the distribution system.

According to a study by U.K. Gas, rupture of high pressure transmission lines caused by corrosion is unlikely at stress levels below 58% of SMYS (SoCalGas, 1991), and most of the older SoCalGas lines are operated well below this threshold.

The more significant risk to the transmission system are the non-predictable failures - "sudden failure due to unusual loading conditions, usually earthquakes, related to poor construction techniques or materials" (SoCalGas, 1991) Certain pipe classes have been identified as having vulnerable welds. These pipes have failed in the past under unusual loads. Unusual loading conditions include (SoCalGas, 1991):

- pipeline exposure in cold temperatures causing contraction and weld failure,
- use of construction equipment over pipelines cracking welds,
- train derailment, and
- earthquake loads.

The risk and impact of failure due to earthquake loads has been addressed qualitatively by Transmission in the Value Chain Analysis (SoCalGas, 1991). Vulnerable classes of pipe have been identified for replacement based on location through identified seismic hazard zones in areas of population concentration, where the impact of failure would be most significant. Priorities are based on safety, cost, and reliability of delivery.

3.3 PLANNING TIME FRAMES

Because Distribution and Transmission are governed by different risks and replacement procedures, the time frame for replacement planning varies significantly.

Distribution is primarily concerned with the predictable effects of corrosion, and time limitations with respect to public works moratoriums. As a result, the time frame within which Distribution currently operates is five years, as reflected in EPOCH. Links to the System Planning Department are limited - projects require System Planning review only if the cost exceeds \$200,000 or if the project crosses divisional boundaries. It is presumed that Distribution consults the "Master Plan" in designing replacement projects.

On the other hand, because transmission replacement projects involve large capital outlays, it is more closely tied to System Planning. Replacements proceed in anticipation of infrequent earthquake events, in response to anticipated changing loads, and in response to construction projects impacting the pipeline right-of-way. The planning time frame is by necessity, significantly longer than that of Distribution.

SECTION 4

GENERAL METHODOLOGY FOR PIPELINE REPLACEMENT DECISIONS

In this section, a general methodology is outlined for making optimal decisions regarding pipeline replacement. Whereas other parts of this study have focused on individual methods of quantifying seismic or corrosion-related risks, this section begins to define how these risks can be balanced in an overall risk reduction program.

The methodology for optimizing pipeline replacement decisions can be described by seven basic steps, as diagrammed in Figure 4-1. The basic steps apply to both Transmission and Distribution, but the implementation will vary depending upon each department's operation. In each of the first four steps, a weighting factor (based on guidelines to be developed by EQE, Cornell University and SoCalGas) is used to facilitate project prioritization. These factors, designated as P (Pipe factor), D (Demand factor), S (Seismic hazard factor), and O (Other, non-seismic factor) are described in the following sections.

4.1 PIPELINE REPLACEMENT METHODOLOGY FOR TRANSMISSION

Various transmission pipelines have been identified for replacement through a systematic review of pipeline class and location. Other factors may be incorporated into the assessment in a consistent manner.

4.1.1 Review Pipeline Data

This task entails a review of pipeline data for the segment under consideration. Pipeline material (i.e., weld-type), age, diameter, pressure, and cathodic protection are all key factors. Additional emphasis would be placed on operational history. A review of each individual pipeline's operational history might allow for consideration of poor performers in otherwise acceptable pipe classes.

This step also includes determining whether the pipe segment is considered vulnerable or mechanically deficient, or falls into a class of pipe included in previously established long-term replacement programs. All of the information gathered is used to estimate the pipeline data weighting factor (the "P" factor).

4-1



2HD 400nb/SCG-FG41

Figure 4-1: General Methodology for Pipeline Replacement Decisions

4.1.2 Review System Planning Department's Long-Term Goals

This phase of the methodology entails checking with System Planning to determine whether the segment under consideration is within or serves an area of changing demand, and is scheduled for upsizing, downsizing, or abandonment. Currently, this data is contained in a hard-copy "Master Plan". This information is used to select replacement pressures or diameters, as well as to compute a "Demand" weighting factor ("D" Factor).

This process could be streamlined with the implementation of Geographic Information System (GIS) methods. A digital map could be developed by System Planning identifying areas of "Planning Concern". Such a map would be "dynamic" and reflect on-going system alterations, identify areas of expected growth or development, and include information on expected loads and service requirements. This map could be made available to Transmission such that a quick on-screen review would indicate future requirements for any pipe in question. Alternatively, if the pipe falls within an area of "Planning Concern", it could be a "flag" requiring a project review by the System Planning Department.

Transmission line replacement is more closely tied to the goals of Planning than Distribution line replacement. Transmission, by nature, must more directly address changing regional demands. System Planning Department input during review of existing lines helps identify the optimal size, pressure and location for a given pipeline. Possible links to distribution projects should also be taken into consideration. This may be accomplished through examination of hard copy plans, on-screen maps or direct System Planning Department input.

4.1.3 Assess Seismic Hazard and Impact of Outage

The main stimulus for transmission line replacement is anticipation of sudden failure due to seismic hazards. Detailed delineation of these hazards is essential for development of a multi-risk decision-making procedure. The implementation of this step presupposes the existence of seismic hazard maps at a scale appropriate for application to the transmission system. Some mapping has been performed in previous studies by consultants, but additional maps would be required. Justification of such expenditures might come in the form of savings gained by reducing the size of hazard areas as currently identified, and the corresponding reduction in the amount of pipe to be replaced. Because a consistent companywide approach is preferable, the most likely proponent of systematic hazard mapping would be the System Planning Department. While the areal extent of useful maps varies significantly between Transmission and Distribution, a program to identify hazard areas significant to both systems would, in the long run, save money as well as improve safety and system reliability.

Possible candidate hazards for mapping include surface fault rupture, liquefaction susceptibility, landslide, lateral spread, and strong ground shaking. Information is currently available on the topics of landslide, liquefaction and surface fault rupture. Consideration of these seismic hazards, if available in digital form, would be straightforward to implement. For example, the California Division of Mines and Geology (CDMG) has an on-going program to map active and potentially-active fault traces within the state at a scale of 1 inch = 2,000 feet. These maps are developed and published under the auspices of the Alquist-Priolo Special Studies Zone Act of 1972, and are readily available. Such maps could be consulted in hard-copy map form, or be digitized for automatic on-line overlay. Strong ground shaking maps, if required, could be developed in either probabilistic or scenario-based forms.

For this step in the methodology, available seismic hazard maps are consulted to determine if the pipe segment under study crosses any of the various hazard zones. In conjunction with relative pipeline vulnerability data, this information is used to identify optimal repair/replacement techniques, and replacement material. Alternatively, the presence of the seismic hazard may activate the requirement for a review by System Planning.

The impact of pipeline outage is also considered. If the pipeline in question is a critical transmission or supply line, whose outage would isolate numerous customers, consideration is given to possible relocation, additional redundancy, or placement of isolation valves to limit outage and speed restoration. Such an assessment requires information on service areas, supply, and redundancy. The seismic hazard information and impact assessment are utilized to develop the seismic hazard or "S" weighting factor.

4-4

4.1.4 Assess Other Non-Seismic Risks

Additional risks threatening transmission pipelines may be addressed in this step of the methodology. Although the VCA report dismissed the other identified "unusual load" risks for the system as a whole, there may be certain pipeline segments wherein these risks should be considered. Other, as yet unidentified risks, could be easily added into the assessment as well. Any such conditions would be incorporated here into the "Other" risk or "O" Factor.

4.1.5 Select Replacement Strategy

Based on the planning goals, seismic hazard and other risk exposure, an optimal replacement strategy for the transmission line under review is developed. Possible strategies include:

- Do not replace at this time
- Do not replace, but perhaps increase monitoring to track some operational deficiency
- Replace with a specified material, diameter and pressure to address planning concerns within a certain time frame
- Relocate pipeline to avoid seismic hazard (that is, install an alternate line and abandon the more hazardous route)
- Develop an alternate "creative" replacement solution including cooperative planning efforts with Distribution. If the pipeline was subject to a System Planning review, an alternate replacement solution may have been suggested. For example, a System Planning review may indicate that an upgrade of a distribution line would allow for the abandonment of the transmission line scheduled for replacement. A cooperative replacement program would allow for cost savings and system optimization.

4.1.6 Calculate Replacement Costs

This task includes all the cost calculations. Based on local cost data, as well as information on future public works projects, the cost of the replacement strategy as specified by the preceding needs and risk assessments is calculated. The results of this step allow for replacement prioritization that incorporates long-term planning goals, as well as risk reduction measures.

4.1.7 Calculate Priority Factor

The four weighting factors estimated in tasks 1 - 4 are combined to develop one overall weighting factor associated with the replacement strategy. This priority factor allows for the relative ranking of various replacement projects based on the needs and risk assessment. The development of criteria for the Transmission priority factor is a critical element in the application of this methodology. Because of the large expense associated with transmission line replacements, and limited capital budgets, the priority factor will essentially determine the sequence of pipeline replacements. For this reason, substantial attention should be given to the development of guidelines for priority factor calculation, with input and general approval from System Planning and Engineering Design, as well as Transmission and Distribution. While the guidelines are expected to vary between the two operational units, the general approach should be consistent.

4.2 PIPELINE REPLACEMENT METHODOLOGY FOR DISTRIBUTION

The methodology for pipeline replacement for distribution lines is presented in Figure 4-2. The methodology is built around the use of EPOCH, a computer program developed by Distribution to optimize decisions with respect to repair/replacement of pipelines affected by corrosion. Additional elements proposed in this study are steps to insure that opportunities to improve seismic safety through replacement are not lost. These steps are discussed in the following subsections.


2HD 400nb/SCG-Fg42

Figure 4-2:Procedure for Incorporating Seismic Hazard or System Enhancement Datainot Pipeline Repair/Replacement Decisions for Distribution

4.2.1 Review Planning Data

In this step, two types of planning data are reviewed. First, data identifying future changes to the system are reviewed. When changes are anticipated or planned, this information may be used by Distribution to help decide when and how replacement of a particular pipe segment should be accomplished.

In the case of seismic hazards, maps could be available for planning purposes to identify opportunities to improve the seismic safety of the system. Based on criteria that would be established by Engineering and Distribution, any significant benefit, with respect to seismic safety, resulting from replacement would be noted.

One possible basis for this criteria could be the degree of seismic hazard and the seismic vulnerability of the pipe segment. Certain combinations of these two parameters would lead to significant seismic safety benefits through replacement. The results of this review, combined with the results from EPOCH, could lead to replacement decisions that incorporate not only economic considerations but safety considerations as well.

4.2.2 Run EPOCH

In this step, the computer program EPOCH would be run to identify the economic benefits of repair versus replacement. This analysis would be run with the updated models for corrosion leakage. Based on the results of this analysis, pipelines would be classified into one of three categories according to economic considerations:

- 1. Pipelines that should be replaced.
- 2. Pipelines that should be repaired.
- Pipelines that are marginal, i.e., the cost difference between the two options is considered small, and the pipeline could either be replaced or repaired.

In general, the assignment to each of these categories will be based on the expected costs associated with mitigating the effects of corrosion. Categories 1 and 2 should reflect firm decisions based primarily on minimizing future costs. Category 3 can result in 1 or 2, with the addition of information on future growth plans or seismic hazard levels.

4.2.3 Post-EPOCH Analysis

If the results of EPOCH suggest that the best mitigation option is to replace the pipe, then replacement incorporates prudent decisions regarding seismic design or future growth.

If the results of EPOCH indicate that repair rather than replacement is, by far, the best option or strategy, then seismic design and other considerations should be postponed until that pipe segment is re-evaluated.

If, however, the results of EPOCH fall into category 3, that is, a borderline decision to repair or replace, then the results of the seismic and planning review could be used to encourage replacement, if significant benefits would result. If however, no benefits are identified, repair is warranted. By incorporating this added step, Distribution will realize the following benefits:

- 1. Maintain EPOCH as the primary decision tool for deciding the repair or replacement issue.
- 2. In marginal cases, decisions can be made to improve the safety and reliability of the system.
- The integration of this added step would insure that Distribution maintains a proactive program to balance all risks in their evaluations.

In order to test this strategy as an effective method for considering all risks, it is recommended that this procedure be tested in a small portion of the SoCalGas service area. In testing this procedure, the results from both the current seismic study and the corrosion analysis would be used. One possible area for this evaluation would be the Torrance area on which both the Cornell and EQE studies have focused.

SECTION 5

REFINED PREDICTIVE MODELS FOR CORROSION LEAKAGE

During the course of this study, a topic identified by Distribution as requiring further investigation was the development of a more comprehensive corrosion leakage model. Although current methods for predicting corrosion leakage appear to be effective when sufficient repair data are available, there are numerous cases where sufficient data do not exist. For these cases, alternative methods were sought.

To evaluate the current procedure utilized by Distribution to project future leakage due to corrosion, a detailed analysis was performed on a small portion of the Distribution system. The purpose of this pilot study was to determine if an alternate model to predict leakage could be developed from available information. This section presents a brief summary of the pilot study results - the full text of the report is contained in Appendix B.

5.1 PILOT STUDY DATA

The pilot area chosen was a small area within the City of Torrance. This area was selected because it is representative of somewhat older areas experiencing problems related to corrosion. Pipeline maps, including atlas sheets, leak history and leak detection survey maps were collected for the 5 atlas sheet study area. In addition, Distribution supplied a detailed data file, extracted from their Leak Repair Order (LRO) database, that contained repair information recorded for the period 1970-1992.

The majority of recorded main leaks were associated with individual homogeneous (with respect to age, material, diameter, cathodic protection) pipeline segments. A comparison of the leak history maps to the LRO data led to the conclusion that the history maps may provide an incomplete picture of actual segment leakage over time. Only 66% of noted main repairs from the LRO data were shown on the plotted history maps.

5.2 ASSESSMENT OF LINEAR REGRESSION PREDICTION TECHNIQUE

The linear regression technique currently used by Distribution to predict future pipeline leakage was tested (See Appendix A for an explanation of the Linear Regression or Least Squares Statistical Method). Cumulative leak rates (cumulative leaks per 1000 feet of pipe) for individual pipeline segments were plotted versus pipeline age. (Equivalent leaks per kilometer can be computed by multiplying leak rates in thousands of feet by 3.28). A "best-fit" line was developed and plotted, to predict subsequent leakage. Figure 5-1 provides sample results for pipe installed in 1912. As can be seen in the figure, linear regression techniques yield good predictions when sufficient data exists. For the test cases, half of the predictions were within 10% of actual, while 70% were within 27%. The major weakness of this approach is that future performance predictions cannot be made reliably when data are limited (i.e., one historic leak occurrence).

5.3 ALTERNATIVE LEAK PREDICTION METHODOLOGY - THE AGE DEPENDENT MODEL

To enhance prediction in those cases where limited historical leakage exists, an alternate "age dependent" predictive model was developed, based on the pipeline's age at instance of first leak¹. It was noted from an analysis of the Torrance data, that the rate at which leak rates increase tended to increase with age at first leak. That is, leak rates increased more quickly on older pipe segments than on newer ones. A regression model relating slope of the leak rate to age at first leak in log-linear space was developed for several sample classes of pipe. Figure 5-2 presents a sample model, for steel pipe installed in 1912.

It must be noted here that pipeline segments with high leak rates at younger ages that were subsequently replaced were removed from this data set. This action is explained by noting that the models being developed would be used on pipe with normal corrosion performance, i.e., not exhibiting excessive repairs in short performance periods. In essence, this led to the elimination of all pipe repair data

¹ Note that this leak is the first recorded leak listed in the Leak Repair Order (LRO) database, which begins after 1970. Leaks that may have been repaired prior to 1970 are not included in the present analysis.

associated with pipes that had been replaced between 1970 - 1992. Appendix A discusses more fully the rationale behind this action.

The age dependent model was tested for the same pipeline segments used in the assessment of linear regression techniques. This comparison is graphically





Figure 5-2: Age Dependent Leak Rate Model for Steel Pipe Installed in 1912

displayed in Figure 5-3. This alternative model was shown to be as good as the linear regression model for segments with extensive leakage histories. The model was also evaluated for pipeline segments with minimal leakage histories, as shown in Figure 5-4. This evaluation showed that the model is able to predict leakage within 15% of actual performance in the majority of cases (2/3) for segments with limited historical leakage.

5.4 CONCLUSIONS AND RECOMMENDATIONS FROM THE PILOT STUDY

Several conclusions and recommendations resulted from the pilot study assessment of leakage prediction techniques. These are repeated here:

Conclusions from the Torrance pilot study:

- The current method of leakage prediction is effective when sufficient leak data exists (i.e., at least three years with leakage)
- 2) Utilizing the leak history maps to represent historical leakage may underestimate leakage because the map updating procedure which entails physical patching often obscures data, and reporting practices may have varied over time. Other sources of data are available, including the Leak Repair Order database, which includes every pipeline repair made since 1970.
- An alternative method (the age dependent model) for verifying linear regression predictions and/or estimating pipeline leakage based on limited leakage data can be developed.

In addition to the above conclusions, the following recommendations are made:

 Since the pilot area used in this study was relatively small, a second analysis is recommended to confirm the trends developed in this first phase. This recommendation should only be implemented if the ability to estimate future leak rates based on limited data is important from the standpoint of the EPOCH program.



5-7



- 2) If further study is justified in this area, two areas are suggested for further study. The first area should be similar, although larger in size, to the Torrance area analyzed in the first phase. The purpose of this assessment will be to confirm the trends observed in the pilot study. In general, the models developed would have the most applicability to these older areas. A second, newer area should be selected to determine if similar models can be developed for other parts of the service area.
- 3) The same type of analysis could be applied to other pipe classes. Although most corrosion problems appear to affect pre-1936 bare pipe without cathodic protection, there are a limited number of other pipe classes that are also affected by corrosion. One such candidate pipe class is poorly coated steel pipe, without cathodic protection, installed between 1941 and 1957.

SECTION 6 RECOMMENDED REFINEMENTS FOR IMPROVED DECISION-MAKING CAPABILITIES

In this section, recommended refinements to improve SoCalGas's company-wide replacement decision-making capability are presented. Three areas are highlighted: Mapping and Records, Interaction with System Planning, and Incorporation of Seismic Risks.

6.1 MAPPING AND RECORDS

During the course of this study, possible improvements for future record keeping practices as well as several schemes to take advantage of the proposed GIS system were identified.

It was noted as a result of the Distribution pilot study that the utilization of leak history maps as the basis of future leakage predictions may be misleading due to possible incomplete recordation of leaks. Improvements in the record keeping process may be as simple as developing official guidelines for leak recordation, including a simple box to check on the leak repair order form. When this data is entered into the LRO database, it will then be possible to tell if the leak has been noted on the map, and a reminder generated if it has not.

With the implementation of a SoCalGas GIS, certain features could be incorporated into the GIS design to enhance operations of Distribution, Transmission and System Planning. For Distribution record-keeping purposes, the leak history map could be digitized, and be available for on-screen leak notation. The dynamic quality of the map would reflect daily activities, and be available to more than one user at a time. In addition, a database consisting of all information currently in the LRO database could be linked to the map file, allowing for inspection of detailed leak information simply by selecting the noted leak location. Eventually, such information could be tied to EPOCH to allow for retrieval of leakage information on specific pipeline segments automatically.

The GIS would also be an efficient platform to allow timely review of System Planning information during the replacement evaluation process. The development

of digital seismic hazard maps, and various planning maps such as "Master Plan" maps, load maps, or areas of "Planning Concern" would speed up the review process. These planning maps would be "dynamic", available on-line, and able to reflect on-going system alterations.

6.2 INTERACTION WITH SYSTEM PLANNING

A significant opportunity exists to link the efforts of Transmission and System Planning, and Distribution and System Planning. For the tasks of recording proposed changes to the system, or maintaining detailed data on seismic hazards, it seems appropriate for the System Planning Department to play a key role. Presumably, detailed data on proposed changes to the system are already being maintained by System Planning. This information, if available in a convenient format for all users, could be accessed by the other departments. This would insure that system information is used consistently by all departments.

With respect to seismic hazard conditions, if detailed maps are developed and maintained by Planning, decisions regarding opportunities for improving seismic safety can be made in a consistent manner by all departments. At this point, the following maps been further investigated:

- Potential Surface Fault Rupture Maps (Alquist-Priolo Fault Maps)
- Liquefaction Susceptibility or Potential Maps (Refinements made by Professor T. O'Rourke of Cornell University)
- Maps identifying areas of significant strong ground motion amplification (e.g., areas overlying soft soils, or deep alluvial deposits)

6.3 INCORPORATION OF SEISMIC RISKS

An area deserving special attention is the incorporation of seismic risks into Distribution planning efforts. As outlined in Section 4.2., a methodology for incorporating seismic risks into the current pipe/replacement format is recommended. Performing this task would help to insure that not only are economic considerations addressed in pipe replacement, but safety and reliability as well.

It was further recommended that this procedure be tested in the Torrance area where extensive work has already been performed by the Project Team on seismic and corrosion problems.

.

.

SECTION 7 REFERENCES

- O'Rourke, T.D. and M. C. Palmer (1994), "Feasibility Study of Replacement Procedures and Earthquake Performance Related to Gas Transmission Pipelines," NCEER Report No. 94-0012.
- Southern California Gas Company (1991), "Value Chain Analysis of the Pre-WWII Transmission Pipeline Replacement Program."
- Southern California Gas Company (1992), "Annual Report for Calendar Year 1992; Gas Distribution System."
- Strang, G.E. (1986), "Underground Piping System Replacement," Southern California Gas Company Engineering Report.

APPENDIX A EXPLANATION OF LEAST SQUARES METHOD

If a number of data point pairs (x,y) exist and are assumed to be linearly related, the "best-fit" line through these points can be determined using a "least squares" approach. According to the principle of least squares, "... a line provides a good fit to the data if the vertical distances (deviations) from the observed points are small. The measure of goodness of fit is the sum of the squares of these deviations. The best-fit line is then the one having the smallest possible sum of squared deviations" (Devore, 1982). In other words, a "least-squares" fit or a simple linear regression analysis is the process of defining the constants m and b to fill the equation of the line, y = mx + b, such that the variance between the actual (observed) value of y, and the value predicted by the equation is minimized.

In practice, for a given data set $\{(x_1, y_1)...(x_n, y_n)\}$, the constants defining the equation of the best-fit line may be determined as follows:

$$m = \frac{n \sum (x_i y_i) - (\sum x_i)(\sum y_i)}{n \sum x_i^2 - (\sum x_i)^2} , \text{ and}$$
$$b = \frac{\sum y_i - m \sum x_i}{n}$$

where;

$$n = number of (x,y) pairs$$

REF:

Devore, Jay L. (1982), <u>Probability and Statistics for Engineering and the Sciences</u>, Brooks/Cole Publishing Company, Belmont, California

APPENDIX B

RESULTS OF DISTRIBUTION STUDY: "AN ASSESSMENT OF CORROSION LEAKAGE MODELS-APPLICATION TO SOUTHERN CALIFORNIA GAS COMPANY DISTRIBUTION PIPELINES"

In June of 1992, the Southern California Gas Company (SoCalGas) entered into an agreement with EQE International and Cornell University to perform an independent assessment of current SoCalGas pipe repair and replacement strategies. The overall program focused on risk assessment, economic and safety issues. The purpose of this particular report is to summarize the project team's investigation of current SoCalGas leak prediction methods for distribution piping, and to suggest ways of improving these methods. Other topics related to earthquake risk and safety, and cost-benefit methods for determining pipe replacement are discussed in other project reports.

The estimation of future pipeline leaks is a concern shared by many owners and operators of natural gas distribution systems. The decision to replace or repair a damaged line often depends upon how the cost to replace compares to the anticipated costs of future repairs without replacement. Generally, the models that are used to estimate future leak rates are based on past pipeline repair data. Statistical models are developed that correlate expected leak rates with pipe material types, pipe age and corrosion protection. As with most models of this type, predictions are generally reliable as long as ample data is available.

Some of the factors that contribute to poor pipeline performance or damage are high soil corrosivities, high pipe-to-soil potentials, mechanical deficiencies, such as improperly screwed joints, and accidents caused by third party damage. In most cases, however, the only failures that can reasonably be predicted are corrosionrelated failures.

Modern procedures for corrosion control are generally quite effective. Physical application of pipe coatings to steel pipe extends the life of the pipe many years. Cathodic protection applied to bare and coated steel pipes can also mitigate the effect of corrosion. In addition, many operating companies are installing pipe whose materials are not susceptible to corrosion, i.e., plastic. In summary,

B-1

corrosion failures appear to be limited to older pipes that were installed without corrosion protection.

The Southern California Gas Company distribution system currently serves 4.65 million customers (P&GJ, 1992) throughout the southern California area. The company has been in existence since the early 1920s. In these early periods, it was common for bare steel pipe without cathodic protection to be installed.

In order to address the corrosion problem, Distribution initiated an aggressive program of cathodic protection and pipe repair and replacement. To determine whether particular pipe segments ought to be replaced or repaired, SoCalGas has developed an in-house computer program, Efficient Pipeline Operation in a Competitive Habitat (EPOCH), that forecasts future leaks and repair costs. The methodology considers three options using a five-year time frame; pipe replacement, repair and installation of cathodic protection, and repair without cathodic protection. An economic analysis compares the estimated costs associated with each of these options.

One of the areas that has concerned SoCalGas personnel is (1) whether the company is maximizing the use of all available pipe repair data, and (2) whether the statistical models that were being developed for use in the EPOCH program were reasonable predictors of future performance. These issues are addressed in the present report.

The rest of this appendix comprises five sections. Section B.1 describes the data received from SoCalGas Distribution. Included is a discussion of available data for the pilot study area where pipeline repair statistics were analyzed. Section B.2 discusses the development of the pipeline databases. The characteristics of exposed distribution pipelines in the study area are discussed in detail. Section B.3 reviews the current SoCalGas Distribution procedure for estimating future corrosion leaks and tests it with data collected in the pilot study area. Section B.4 presents an alternate corrosion prediction model that incorporates the age of first leak as a model parameter. The results show that the inclusion of this new parameter in the development of corrosion prediction models increases the number of cases in which corrosion predictions can be made. Finally, the major conclusions of this study are presented along with recommendations to further improve corrosion prediction modeling for all of SoCalGas's distribution pipelines.

B.1 DATA RECEIVED FROM SOCALGAS

Data received from SoCalGas may be divided into two categories: general system information, and information specific to the Torrance study area.

B.1.1 System Information

A review of general system information was performed to determine the relation of the small study area to the overall system. The primary source of piping information was a 1986 Engineering Department report entitled "Underground Piping System Replacement Assessment". Included in this report is a piping system breakdown by pipeline family, summarized for non-service facilities in Table B-1.

Table B-1 SoCalGas Pipe Families (Not Including Services)						
Pipe Material	Coating	Installation Era	Cathodic Protection	Length of Pipe (Miles)	Length of Pipe (Kms)	%
Steel	Bare	Pre-1936 Post-1936 Post-1936	Not Specified Unprotected Protected	2,285 2,897 173	1,420 1,801 108	6.3 8.0 0.5
	Coated	1936-1949 1949-1971 1949-1971 Post-1971	Not Specified Unprotected Protected Protected	4,585 4,830 9,780 3,625	2,850 3,002 6,078 2,253	12.6 13.3 26.9 10.0
Plastic				8,050	5,003	22.2.
Copper				60	37	0.2
Cast Iron				58	36	0.2
TOTAL MAINS				36,343	22,588	

B.1.2 Torrance Study Area Data

Detailed data for the selected study area within the City of Torrance was provided by Distribution and the South Coastal Division. The study area was limited to five SoCalGas atlas sheets - Torrance 9, 10, 15, 16 and 17. The boundary of the study area is shown on Figure B-1. For each of these atlas sheets, the following mapped information was collected:

- atlas sheets indicating pipeline material, diameter, date of installation, cathodic protection (CP) and pressure
- leak history maps (scale: 1 inch = 100 feet), which indicate the location and date of repairs attributed to corrosion, material failure, and construction defects. (The notation of mechanical leak repairs is optional.) This map set is updated upon pipe replacement by physically covering previous information by fastening a xerox of the revised atlas sheet pipe section directly to the history map.
 - records of recent leak detection surveys

In addition to the mapped information, the SoCalGas Leak Repair Order (LRO) database for the 5 listed atlas sheets was provided. The LRO database covers the years 1970 through 1992, and includes the repair location, leak code, leak cause, repair date, and miscellaneous pipeline information (diameter, material, year of installation, main/service, etc.). A total of 491 main leaks are listed for the 5 subject atlas sheets.





B.2 ANALYSIS OF TORRANCE AREA DATA

The analysis of the Torrance study area required the development of two complimentary databases; a pipeline segment database and a segment leak database. The correlation of the two data sets allows for the comparison of pipe leakage according to material, diameter, age, and unit length.

B.2.1 Development of Pipeline Segment Database

For each atlas sheet, individual pipeline segments were defined to be homogeneous with respect to pipeline diameter, material, cathodic protection, and age. That is, a change in material or diameter would be reflected by the definition of a separate segment. Only mains were included (i.e., no service pipe), and pipe coating was not part of the data available from the collected information. For each segment, the total length in feet was scaled from the leak history maps. Small index maps identifying pipeline segment names and locations, as well as a printout of the pipeline segment database are provided in Appendix C.

A total of approximately 20 miles (32.2 kms) of main are located within the study area - 26% plastic, 27% steel without CP, and 47% steel with CP. Detailed descriptions of each pipe group are as follows:

- The plastic pipe in use in this area is primarily 2 inches in diameter. All plastic piping was installed after 1978.
- Two-thirds of the steel pipe that is currently not under CP was installed prior to 1936. The remainder was installed prior to 1971, when it became the company standard that all steel pipe should be coated, and put under CP within one year of installation.
- Half of the steel pipe with CP was installed prior to 1971, but information detailing when the segment was actually put under CP was not among the available data, and may be some time after installation.
- While no specific coating information was available for the pipelines in the Torrance area, some generalizations may be

made according to historic installation practices (Strang, 1986). Pipe installed prior to 1936 may be assumed bare. All steel pipe installed after 1971 are required to be cathodically protected. According to SoCalGas data, approximately 82% of steel mains in place, as of the end of 1992, are coated, while the remaining 18% are bare (SoCalGas Annual Report for Calendar Year 1992). (The majority of this uncoated pipe is not cathodically protected.) For installation between the years of 1936 to 1941, and after 1958, the coated pipe may be assumed to have a "good" coating, while those put in place between the years of 1942-1957 may be considered to have "poor" coating (Strang, 1986).

The distribution of pipe types within the study area may be compared to systemwide statistics to determine if the selected study area is a representative sample.

- System-wide, 35% of all mains are plastic. Within the Torrance study area, 26% of mains are plastic.
- The Torrance study area has a higher percentage of steel pipe under cathodic protection than the system average.
 Approximately 56% of steel pipe is under cathodic protection system-wide (SoCalGas, 1992), while within the Torrance area, about 63% of the steel pipe has CP.
- The study area has a higher percentage of older and newer steel pipe than the system average. Overall, most of the system (79%) was installed between 1936 and 1970 (Strang, 1986), while that percentage for the study area is 48%. 8% of all SoCalGas steel mains were installed prior to 1936 vs. 26% of the steel pipe in the Torrance study area. Similarly, 13% of the system was installed in 1971 or after (Strang, 1986), vs. 26% for the study area.

B-7

B.2.2 Leak Assignments

Having identified pipeline segments and their characteristics, it was necessary to associate leak repairs as reported in the LRO database with individual pipeline segments. This required physically locating each repair in the LRO database on the leak history map or atlas sheet, according to address or location. This task was facilitated through the use of address map book pages from the City of Torrance, indicating street addresses for various lots. Verification of leak assignments made on the basis of location was possible, to some extent, through cross-referencing of pipeline data tabulated in the LRO database. For example, if the LRO location was an intersection containing two perpendicular pipes, assignment could usually be made by matching the pipe diameter or installation date in the LRO database with the same information in the pipeline segment database. In this way, virtually all main leaks were associated with an individual pipeline segment. A printout of this leak database is provided in Appendix D.

Of the 491 main leaks indicated in the LRO database, 32 were unlocatable. For some of these repairs, the address/location was a vague description, and the local pipes were similar enough to make a definitive assignment impossible. In other cases, repair addresses were physically on a different atlas sheet not included in the study area. Of the remaining 459 leaks, 70% (322 leaks) were attributed to corrosion, 1% (5 leaks) to material failure, and 2% (8 leaks) to construction defects. An additional 19% (88 leaks) were classified as "other" causes. This category includes mechanical leaks such as thread leaks. The remainder were classified as "outside or third party damage" (2%) or "unknown" causes (6%) which includes cases of repair or replacement when the leak itself was not physically exposed.

Because the LRO data included all leak repairs made between the years of 1970 and 1992, it was possible to determine pipe characteristics for pipeline segments that were subsequently replaced. For example, a pipeline segment replaced in 1986 will likely have its history concealed on the leak history map, but the LRO for years prior to 1986 will indicate the pipeline material, diameter and original installation date. This information was used to determine the "previous" pipe information for various pipeline segments, and allowed the "previous" pipeline segments and their leaks to be included in the leakage analysis. This information is included in the pipeline segment database in Appendix C.

B.2.3 Summary of Corrosion Leakage Data

Steel pipeline segments may be grouped into various categories, as follows:

- A) Steel pipe under CP
- B) Steel pipe without CP, installed pre-1936 (considered bare)
- C) Steel pipe without CP, installed between 1936 and 1971 (predominantly coated)
 - C1) 1936 1940, and 1958 1970 "good" coating
 - C2) 1941 1957 "poor" coating

For the steel pipe, The majority of the 322 corrosion repairs recorded in the LRO were assigned to Category B pipe (278 repairs, or 86%), making this data set the most suitable for detailed analysis. The remainder were distributed as follows;

- Category A 24 repairs (7%)
- Category C2 15 repairs (5%)
- Category C1 5 repairs (2%)

B.2.4 Comparison of LRO Data to the Leak History Maps

By physically plotting all listed repairs onto the leak history maps, it was possible to assess the completeness of the data as plotted by SoCalGas. According to SoCalGas personnel (personal communication with Mr. Ron Hammer, November, 1992), all repairs of leaks caused by corrosion, material failure, or construction defects should be posted on the leak history maps (leak causes 1 - 3, 9 and 10, respectively in the LRO database). Posting of mechanical or joint leaks is optional (these leaks are included in LRO leak cause 11 - "Other").

To make a comparison between the history maps and the LRO database, only those pipeline segments whose history for the years 1970 to 1992 is visible on the leak history maps may be used. That is, only segments that were not replaced during that time period, and have not had their leak history concealed are included. Of the 322 corrosion leaks included in the LRO database, 178 are on pipeline sections with visible history. Approximately 66% of these repairs were actually plotted on the maps by SoCalGas. Percentages according to leak code are as follows:

- approximately 58% of all Code 1 leaks on visible sections were noted on the map,
- approximately 69% of all Code 2 leaks, and
- approximately 68% of all Code 3 leaks.

While few of the repairs within the LRO database were caused by construction defects or material failure, virtually none of the these repairs were indicated on visible segments. In addition, it appears that only 17% of repairs due to "other" causes on visible segments are plotted (plotting of these repairs is optional).

It is possible, therefore, that using leak data from the leak history map may provide an incomplete picture of actual segment leakage over time. While the history maps do contain information prior to the 1970 start date of the LRO database, it does not appear that the mapped information is a comprehensive record.

B.2.5 General Approach for Calculating Leak Rates

Cumulative leak rates, or the total number of historic leaks developed by a pipeline segment per unit length, can form the basis for the prediction of future pipeline performance. Using pipeline characteristics and leak data, it was possible to develop cumulative leak rate histories for individual pipeline segments, and alternatively, to group all pipe of a particular category together to develop a "system" leak rate curve for that pipe class. Leak rates were developed according to pipe age at the time of leak occurrence. For the system analysis, pipe and associated leaks were aggregated according to the categories listed in Section B.2.3.

B.3 ASSESSMENT OF SIMPLE LINEAR REGRESSION APPROACH FOR LEAK PREDICTION

Currently, the repair/replace decision-making program utilized by SoCalGas includes the ability to predict future leakage from available pipeline data. Historical leakage data is taken from the leak history maps, and a standard least squares fit of the data is used to project future leakage on individual pipeline segments. This section discusses the method used to test the validity of this approach.

B.3.1 Test Method

To test the validity of a simple linear regression model for pipeline leakage, the following procedure was used:

- Historical leakage from the LRO database was used. This database is considered to be more comprehensive for post-1970 data because no data is lost during the map update process. The database was segregated into two data sets: data through 1986, and post-1986 data.
- A simple linear regression was performed on the data set through 1986. Cumulative leak rates (cumulative leaks per 1,000 feet of pipe) for individual pipeline segments were plotted versus pipeline age. A "best-fit" line was developed to predict subsequent leakage.
- Actual post-1986 performance was plotted and compared to predicted performance.

For the pre-1936 steel pipe data set (accounting for 86% of all corrosion leaks in the LRO database), 10 non-replaced segments had sufficient data to be included in the assessment. To be included, pipeline segments had to meet three critical requirements: 1) the segment must not have been replaced, 2) the segment must have sufficient data through 1986 to perform a linear regression (at least 2 years with reported leakage), and 3) actual post-1986 leakage data must exist to compare to the prediction. Figures B-2 through B-5 present the test results for the 10 segments, in three data sets; pipe installed in 1912 (Figures B-2 and



through 1986 to Actual Post-1986 Performance (Pipe Installed in 1912)

B-12

∠ ۱−د



2HD 400nb/16-16-L2

---- Least Squares Fit of Data Through 1986 * Post-1986 Leak Data 📙 📕 Leak Data Through 1986





B-14




Least Squares Fit of Data Through 1986

|

Post-1986 Leak Data

×

Legend: 🔳 Leak Data Through 1986

2HD 400nb/10-12CL2

B-3), 1913 (Figure B-4) and 1927 (Figure B-5). These three pipe categories represent roughly 60% of all non-replaced pre-1936 steel pipe. (The remaining 40% is divided among 14 different installation years.)

B.3.2 Results

As shown in the figures, the linear regression technique is a good predictor of actual performance when sufficient data exists (i.e., more than three data points), but the accuracy tends to decrease with decreasing data. For the 10 sample segments, 50% of the predictions at the age of last recorded post-86 leak were within 10% of the actual performance. Figure B-6 presents the distribution of the difference between predicted and actual performance for the straight line regression technique. It should be noted, however, that only pipeline segments with leakage in at least two years are included in this prediction assessment. No prediction can be made for segments with only one leak.

B.3.3 Strengths and Weaknesses of Simple Linear Regression Model

Strengths and weaknesses of the linear regression approach may be summarized as follows:

Strengths:

- With sufficient data, the linear regression model provides a good predictor of future pipeline leak rates.
- By incorporating each individual segment's leakage history in the analysis, it is possible to include localized factors that might influence leak rates.
- The method is simple to develop and easy to use.

Weaknesses:

- This method can not predict future performance based on limited data (i.e., 1 data point).
- Accuracy decreases with fewer data points.



Figure B-6: Percent Difference Between Predicted and Actual Leak Rates using Simple Linear Regression techniques

 A least squares regression model developed from data on the leak history maps may not include all available data, and accuracy of prediction may be reduced.

B.4 DEVELOPMENT OF ALTERNATIVE LEAK PREDICTION METHODOLOGY: AGE DEPENDENT MODEL

Because the simple linear regression approach had several weaknesses, the possibility of developing a supplemental approach to leak prediction based on available data was explored. A variety of approaches were taken, and an alternative leak prediction model was developed. This section describes the development of the supplemental "age dependent" model. In this particular context, age is defined as the age of the pipe at the first observed leak. Note, that the database used in this study begins in 1970; therefore, it is possible that earlier leaks (before 1970) occurred which were not included in this analysis.

B.4.1 Development of the System Curve

For all pre-1936 steel pipe, the total number of leaks, and the age at which they occurred were tabulated using the 1970 - 1992 LRO data. Similarly, the total amount of pipe of a particular age in the ground was determined. For example, a steel pipeline segment installed in 1913 would contribute to the appropriate steel family for ages 57 - 79. If listed repairs occurred in 1975, 1980 and 1984, these leaks would contribute to ages 62, 67, and 71. Pipe that was replaced at some point during the 22 years of detailed data must be treated somewhat differently. In order to reflect the fact that the pipe is removed from the ground, the corresponding data must also be removed from the database. If the example pipe referenced above was replaced in 1986, its length would only contribute through age 73. In addition, its leaks must also be removed following replacement.

Figure B-7 presents cumulative leak rates versus age for replaced and non-replaced pre-1936 steel pipe without catholic protection. As is evident in the figure, the trends reflected by the two categories of pipe are quite different. For the replaced pipe, the trend reflects a series of upward and downward slopes. In large part, the downward trends are attributed to pipeline replacement, and associated removal of data. In addition, the initial cumulative leak rates for replaced pipe (i.e., for ages 45 to 55) are higher than those for non-replaced pipe for the same age range.



Figure B-7: Cumulative Corrosion Leak Rates for Pre-1936 Steel Pipe without Cathodic Protection

According to discussions with SoCalGas personnel, some types of steel pipe experience rapid increases in repair rates, even at fairly young ages. Because of these high rates of initial repair, these pipelines were replaced even though the pipes were quite young. Since the focus of this particular study is to develop methods to identify when to repair and when to replace pipe based on long-term data, it was decided to exclude the replaced pipe information. The decision to replace pipe with extremely high repair rates usually does not require a sophisticated cost-benefit analysis. Where cost-benefit analysis can play a meaningful role is in deciding whether replacement will offset long-term repair costs.

It is important to note that the combination of the two curves in Figure B-7 reflect the total performance of pre-1936 steel pipe in this particular area. It is difficult to determine whether the higher slopes in the replaced pipe actually reflect leaks that occurred at earlier pipe ages since the data set that was used in this study only begins in 1970. In order to resolve this issue, i.e., higher slopes at earlier pipe ages, it would be prudent to investigate repair data beginning before 1970, and perhaps data from other parts of the system.

It may be possible that the trends observed in this study are unique to this particular service area. Therefore, before the models can be used for other areas, more analysis would have to be performed to justify the current trends and to determine whether they are system wide. This particular recommendation should be implemented only if SoCalGas personnel find the ability to estimate future leak rates based on one or two data points important.

Having plotted individual segment performance for the assessment of the linear regression model, it was noted that the slope of the best fit line appeared to increase with increasing age of first leak. That is, it appeared that the rate of increase in leak rate was higher for leaks striking older pipe. Figure B-8 displays individual pipeline segment leakage histories for the ten sample segments plotted with the non-replaced system curve. The general trend appears to be increasing slope with increasing age at first leak.



2HD 400nb/SYS-1912

Figure B-8: Plot of Individual Pipe Segment Leakage Histories on the Cumulative Corrosion Leak Rate Curve

B.4.2 Development of Age Dependent Model

It was clear that the predictive model should capture this variation with age at first leak, as well as reflect differences in installation practices over time. The resulting model relates slope of the leak rate to age at first leak, for pipe installed in various years. For this pilot application, pipe installed in the years 1912, 1913 and 1927 were used to represent pipe in the pre-1936 steel class. (As mentioned before, these sub-classes represent 60% of all non-replaced pre-1936 steel pipe without cathodic protection.)

For each pipeline segment installed in a given year, the slope of the best fit line for leak rate data through 1986 (developed for the linear regression analysis) was plotted against age at first leak. This data appeared to be linear in semi-log space, so a regression analysis was performed to determine the log-linear best fit line. For the 1912 and 1927 data sets, these lines were determined by 5 and 3 data points, respectively. Figure B-9 and B-10 present the resultant leak rate models for these installation years. The 1913 data set produced only 2 data points, which were deemed insufficient to develop a reliable best-fit line. In this case, the data from 1912, 1913 and 1927 were merged into one data set to develop a generalized leak rate model to be used in predictions for the 1913 data. This model is presented in Figure B-11.

B.4.3 Testing of Age Dependent Model

The testing of the age dependent model consisted of two basic tasks: 1) evaluate the performance of the age dependent model for segments with extensive leakage history (for which the linear regression model works well), and 2) evaluate its performance for segments with minimal leakage history (for which no linear regression model can be developed).

B.4.3.1 EVALUATION OF AGE DEPENDENT MODEL FOR SEGMENTS WITH EXTENSIVE LEAKAGE HISTORIES

To test the accuracy of the predictive model for segments with extensive leakage, the same 10 segments used for testing the linear regression model were utilized. For each pipeline segment, the age at first leak was used to determine the slope of the predictor line from the age dependent leak models (Figures B-9 through B-11).



Figure B-9: Age Dependent Leak Rate Model for Steel Pipe Installed in 1912



Figure B-10: Age Dependent leak Rate Model for Steel Pipe Installed in 1927



Figure B-11: General Age Dependent Leak Rate Model for Pre-1936 Steel Pipe

From the point of first leak, a line with this slope was drawn. Figures B-12 through B-15 present the results of this evaluation. On each figure, the solid line represents the age dependent model prediction, while the dashed line represents the linear regression prediction.

In most cases, the age dependent model is approximately as good or a better predictor than the linear regression model. The average difference between actual and predicted performance for these 10 pipeline segments is 25.7% for the age dependent model, versus 29.5% for the simple linear regression. Figure B-16 presents the distribution of the difference between predicted and actual performance for both the age dependent model (solid bars) and the straight line regression technique (hatched bars).

B.4.3.2 EVALUATION OF AGE DEPENDENT MODEL FOR SEGMENTS WITH MINIMAL LEAKAGE HISTORIES

While the previous comparison shows that using the age dependent model is roughly equivalent to using the linear regression model, it only applies to pipeline segments with leaks in at least two years. For pipeline segments that have just begun to leak, and have had only one recorded year with leakage, the linear regression technique can not be used to predict future performance, but the age dependent model can. Within the current database, the number of such segments is not insignificant. Figure B-17 shows the distribution of non-replaced pre-1936 steel pipeline segments according to the number of years with leakage (1970 - 1992). As shown in this figure, 45% of the segments could utilize the linear regression model (i.e., 2 or more years with leakage), and 36% have not yet suffered leakage. The remaining 19% have suffered only one leak to date, and could utilize the age dependent model for future leakage prediction.

Six sample segments (installed in 1912, 1913 and 1927) with one leak in the years through 1986, and at least one leak after 1986 have been identified. For these segments, the age dependent model was utilized to predict future leakage. These predictions are shown in Figures B-18 (pipe installed in 1912), B-19 (1913) and B-20 (1927). The results of these predictions are, for the most part, fairly accurate. In fact, in 2/3 of the cases, the predicted leakage is within 15% of actual performance.





Figure B-13: Comparison of Leak Rates Predicted from the Age Dependent Model to Actual Post-1986 Performance (Pipe Installed in 1912)

Age Dependent Model Prediction

— Least Squares Fit of Data Through 1986

Post-1986 Leak Data

×

Leak Data Through 1986

Legend:

2HD 400nb/16-16-L







to Actual Post-1986 Performance (Pipe Installed in 1927)

2HD 400nb/10-12C-L



Figure B-16: Percent Difference Between Predicted and Actual Number of Leaks



Figure B-17: Distribution of Pipe Segments by Number of Years with Leak Occurrences

2HD 400nb/15-3A-R



Age Dependent Model Prediction

ī

Post-1986 Leak Data

*

Legend: 🔳 Leak Data Through 1986







2HD 400nb/9-3-R



B.5 CONCLUSIONS AND RECOMMENDATIONS

The assessment of pipeline data in the Torrance study area has led to the following conclusions:

- The current method of leakage prediction is effective when sufficient leak data exists (i.e., at least three years with leakage)
- 2) Utilizing the leak history maps to represent historical leakage may underestimate leakage because the map updating procedure often obscures data, and reporting practices may have varied over time. Other sources of data are available, including the Leak Repair Order database, which includes every pipeline repair made sine 1970.
- 3) An alternative method (the age dependent model) for verifying linear regression predictions and/or estimating pipeline leakage based on limited leakage data can be developed.

In addition to the above conclusions, the following recommendations are made:

- Since the pilot area used in this study was relatively small, a second analysis is recommended to confirm the trends developed in this first phase. This recommendation is considered significant only if the ability to predict future leaks based on limited data is considered important.
- 2) Two areas are suggested for further study. The first area should be similar, although larger in size, to the Torrance area analyzed in the first phase. The purpose of this assessment will be to confirm the trends observed in the pilot study. In general, the models developed would have the most applicability to these older areas. A second, newer area should be selected to determine if similar models can be developed for other parts of the service area.

3) The same type of analysis could be applied to other pipe classes. Although most corrosion problems appear to affect pre-1936 bare pipe without cathodic protection, there are a limited number of other pipe classes that are also affected by corrosion. One such candidate pipe class is poorly coated steel pipe, without cathodic protection, installed between 1941 and 1957.

B.6 REFERENCES

- P&GJ (1992), "The 12th Annual Pipeline and Gas Journal 500 Report", <u>Pipeline and</u> <u>Gas Journal</u>, September.
- Southern California Gas Company (1992), "Annual Report for Calendar Year 1992; Gas Distribution System."
- Strang, G.E. (1986), "Underground Piping System Replacement", Southern California Gas Company Engineering Department Report.

APPENDIX C

- INDEX MAPS TO ATLAS SHEETS

and

- PIPELINE SEGMENT DATABASE



Z

Preceding page blank







Z









Schematic Index Map for Atlas Sheet Torrance 16

C-6



Seg #	P S	ipeline egment#	Material	Diameter	Year Installec	Pressure	СР	Length in feet	Material	Diameter	Year Installed	Age at Repl.	CP
1	Tor	09-1	s	2	71	M	Y	825	s	2	13	58	N
2	Тог	09-2	S	3	56	M	Ŷ	460					
5	Tor	09-2A	5	2	10	M	N	125					
5	Tor	09-3A	P	3	85	M	N	50					
6	Tor	09-4A	S	3	76	M	Y	195	S	3	12	64	N
7	Tor	09-4B	P	2	85	M	N	250	S	5 7	12	73	N
9	Tor	09-40	P	2	86	M	N	275	S	2	13	73	N
10	Tor	09-5A	P	2	86	M	N	75	-	-			
11	Tor	09-6	S	2	12	M	N	600					
12	Tor	09-6A	S	2	12	M	N	510					
14	Tor	09-7	s	2	49	M	N	310					
15	Tor	09-9	s	2	54	M	N	200					
16	Tor	09-10	S	3	49	M	N	570	_	-			
17	Тог	09-10A	S	3	62	M	Y	400	S	3	12	50	N
10	Tor	09-11	S P	2	86	M	N	335	3	2	12	50	N
źó	Тог	09-12A	Ś	2	73	M	Ŷ	80					
21	Tor	09-13	Р	2	90	M	N	300	S	2	27	63	N
22	Tor	09-13A	S	2	2/	M	N	185					
23	Тог	09-135	S	6	82	M	Ŷ	855	s	6	13	69	N
25	Тог	09-14A	s	6	12	M	Ň	810					
26	Tor	09-14B	S	6	72	M	Y	26					
27	Tor	09-140	S	6	15	M	N	400	c	6	25	48	N
20	Tor	09-140	S	2	45	M	N	37	3	0	23	40	n
30	Tor	09-15	s	ž	73	M	Ŷ	300	S	8	12	61	N
31	Tor	09-16	S	2	51	M	N	375					
32	Tor	09-16A	S	2	4/ 87	M	N	155					
34	Tor	09-18	S	8	22	M	Ň	400					
35	Тог	09-19	S	4	12	M	N	120					
36	Tor	09-20	P	1	80	M	N	80					
37	Tor	09-21	S	8 2	60 51	M	N N	285					
39	Тог	09-22A	S	2	40	M	N	130					
40	Tor	09-22B	P	2	83	м	N	100					
41	Tor	09-23	S	3	51	M	Y	260					
42	Tor	09-24	P	3 3	89	M	N N	1025					
45	Tor	09-26	S	8	45	M	N	260					
45	Tor	09-30A	S	8	45	M	N	50					
46	Tor	09-30B	S	8	56	M	N	10					
47 48	Tor	09-31	s c	2	36	M	N	20					
49	Tor	10-1	Š	ž	51	M	Ŷ	475					
50	Tor	10-2	S	2	51	M	Y	465					
51	Tor	10-3	S	2	51	M	Ŷ	630					
52	Tor	10-4	s	2	27	M	N	400					
54	Tor	10-5B	ŝ	2	35	M	N	85					
55	Tor	10-5C	S	2	37	M	N	100					
56	Tor	10-5D	S	2	39	M	Ŷ	105					
57 58	Tor	10-5E	s S	2	41 56	M	Ý	120					
59	Tor	10-6	P	2	83	M	N	500	S	3	24	59	N
60	Tor	10-7	S	2	27	M	N	330					
61	Tor	10-7A	S	2 2	26 87	M	א ע	190					
63	Tor	10-8	P	2	83	M	N	160	S	2	26	57	N

<----- CURRENT PIPELINE INFORMATION -----> <---- PREVIOUS PIPELINE INFORMATION ----->

<	CURRENT	PIPELINE	INFORMATION	>	<	PREVIOUS	PIPELINE	INFORMATION	>
---	---------	----------	-------------	---	-------------	----------	----------	-------------	---

Seg #	Pipeline Segment#	Material	Diameter I	Year nstallec	Pressure	СР	Length in feet	Material	Diameter	Year Installed	Age at Repl.	СР
64	Tor 10-8A	S	2	51	M	Y	405					
65	Tor 10-8B	S	2	24	M	N	75					
66	Tor 10-9	P	2	85	M	N	250	S	2	24	61	N
07 88	10F 10-10 Tor 10-11	5	8	83 78	н м	Y	2205	s	2	24	5/	LI I
69	Tor 10-12A	S	1	43	M	Ň	50	3	2	24	24	R
70	Tor 10-12B	S	3	28	М	N	100					
71	Tor 10-12C	S	3	27	M	N	850	_				
72	Tor 10-12D	P	37	82	M	N	400	S	2	24	58	N
74	Tor 10-12E	S	4	45	m M	Ý	45					
75	Tor 10-12G	s	3	56	M	Ý	385					
76	Tor 10-12H	S	4	45	M	N	50					
77	Tor 10-121	S	2	13	M	Y	500	<u> </u>	2	20		
70 70	Tor 10-15	P	2	78	M	N	550	s	2	20) C 27	N
80	Tor 10-14A	s	ž	39	M	Ň	185	5		27		п
81	Tor 10-15*	S	2	27	M	N	395					
82	Tor 10-15A	S	2	29	M	N	355					
83	Tor 10-16	S	2	24	M	N	375					
04 85	Tor 10-10A	s c	2	-25	M	N 10	300					
86	Tor 10-18A	S	2	17	M	N	860					
87	Тог 10-18В	S	2	74	м	Y	50					
88	Tor 10-18C	S	2	17	M	N	20					
89	Tor 10-19A	S	2	78	M	Ŷ	375	•	•	47		
90	100 10-198	5 6	2	74 56	M M	T Y	225	5	2	17	57	N
92	Tor 10-19D	S	2	13	M	Ň	1010					
93	Tor 10-20A	s	2	17	M	N	325					
94	Tor 10-208	NO RECORD)									
95	Tor 10-20C	S	2	17	M	N	370	_	-	• •		
96	Tor 10-21	e e	2	78 75	M	N Y	740	S	2	16	62	N
98	Tor 10-22A	s	2	73	M	Ŷ	270					
99	Tor 10-23	S	2	70	M	Ý	290					
100	Tor 10-23A	S	2	73	M	Y	430					
101	Tor 10-24	S	2	72	M	Y	375	~	~	27	F/	
102	Tor 10-24A	P	2	60 86	л м	N	1125	5 C	2	27	20 50	N
104	Tor 10-26	P	2	86	M	N	1125	S	2	27	59	N
105	Tor 10-27	P	2	86	M	N	1110	S	2	27	59	N
106	Tor 10-28	Ρ	2	85	M	N	70					
107	Tor 10-29	P	2	86	M	N	80					
100	Tor 10-31	P	2	00 86	M	N	95					
110	Tor 10-32	P	2	86	M	Ň	280					
111	Tor 10-33	Ρ	2	81	M	N	125					
112	Tor 10-34	P	2	86	M	N	120					
113	Tor 10-35	S	2	66	M	N	220	~	-	24		
114	Tor 10-35A	Р 5	2	00 71	л м	N Y	200	5	2	21	60 //0	N
116	Tor 10-37	s	ž	38	M	Ň	120	5	2	~~	47	n
117	Tor 10-37A	S	2	19	м	N	200					
118	Tor 10-378	S	2	18	M	N	135					
119	Tor 10-38	S	2	45	M	N	100	c	-			.,
120	Tor 10-39	۲ ۲	2	10	m M	N	204	2	5	1د	41	N
122	Tor 10-40A	S	3	45	M	Ŷ	465					
123	Tor 10-40D	Ρ	2	86	м	Ň	1475	S	2	21	65	N
124	Tor 10-40E	P	2	78	M	N	360	S	2	22	56	N
125	10r 10-41	S	4	39	M	Y N	425					
160	101 10-100	3	۷	12	m	n	100					

Seg #	Pi Se	peline gment#	Material	Diameter	Year Installed	Pressure I	СР	Length in feet	Material	Diameter	Year Installed	Age at Repl.	CP
127	Tor	10-101	s	4	39	M	Ŷ	150					
128	Тог	10-101B	S	4	27	M	Y	15					
129	Tor	10-102	P	2	84	M	N	120	S	2	21	63	N
130	IOF	15-1	5	4	/ l 83	H M	า ม	500 475					
132	Tor	15-3	Ś	2	39	M	N	210					
133	Tor	15-3A	S	2	27	M	N	310					
134	Tor	15-3B	Р	2	91	M	N	185					
135	Ior	15-3C 15-4	S	2 8	51	M 4	Y	390	c	R	20	50	N
137	Tor	15-4A	S	8	45	H	Ň	1320	5	U	20		N
138	Tor	15-4B	S	8	91	н	Y	225	S	8	24	67	N
139	Tor	15-4C	S	8	83	Н	Y	340					
140	Tor	15-5	S	8	85	M	Ŷ	290					
141	Tor	15-0	s	2	57	M	Y	325					
143	Tor	15-7A	s	2	78	M	Ý	310					
144	Tor	15-8	S	4	45	M	N	24					
145	Tor	15-8A	S	4	27	H	N	1325					
146	Tor	15-8B	S D	2	55	н	N	275	c	2	27	50	м
147	Tor	15-9	P P	2	84	M	N	50	3	2	21	78	n
149	Tor	15-10	P	2	86	M	N	260	S	2	27	59	N
150	Tor	15-11	S	1	73	M	Y	95					
151	Tor	15-12	S	2	27	M	N	225					
152	Tor	15-12A	S	2	41	M	N N	250					
154	Tor	15-120	s	3	12	M	N	275					
155	Tor	15-12D	P	2	84	M	N	320					
156	Tor	15-13	S	2	12	M	N	600					
157	Tor	15-14	Р	4	91	M	N	240					
158	Tor	15-14A	S	4	51	M	Y	150					
160	Tor	15-140	S	2	73	M	Ý	110					
161	Tor	15-14D	S	4	73	M	Ŷ	2810					
162	Tor	15-15	S	2	24	M	N	150					
163	Tor	15-15B	S	2	13	M	N	300					
164	IOR	15-16	S	2 7	51	M	T Y	1010					
166	Tor	15-16R	s	3	56	M	Ŷ	340					
167	Tor	15-16C	P	2	85	M	Ň	149					
168	Tor	15-17	S	2	69	M	Y	490					
169	Tor	15-17A	S	2	40	M	Ŷ	95					
170	Tor	15-178	5 c	2	40	m M	r V	100					
172	Tor	15-18A	S	2	50	M	'n	80					
173	Tor	15-18B	S	2	74	M	Y	50					
174	Tor	16-1	Р	2	86	M	N	625	S	2	12	74	N
175	Tor	16-2	P	2	86	M	N	250					
177	Tor	16-34	s	ן ד	59	M	N	55					
178	Tor	16-4	Š	3	12	M	Ň	495					
179	Tor	16-4A	S	3	67	M	N	35					
180	Tor	16-5	S	2	50	M	Ŷ	135					
181	Tor	16-5A	S	2	50 ∡7	M	۲ ۷	255					
183	Tor	16-6	5	2	67 48	ri M	Ŷ	105					
184	Tor	16-7	S	2	54	M	Ý	370					
185	Tor	16-7A	S	3	13	M	Y	50					
186	Tor	16-9	S	2	40	M	Y	160					
187	Tor	16-10	S	6	82	M	Ŷ	435	S	6	13	69	N
189	Тог	16-11A	s S	6	84	M	Ý	25					

<----- CURRENT PIPELINE INFORMATION -----> <----- PREVIOUS PIPELINE INFORMATION ----->

Seg #	Pi Se	peline gment#	Material	Diameter	Year Installed	Pressure 1	СР	Length in feet	Material	Diameter	Year Installed	Age at Repl.	CP
190	Tor	16-12	S	6	41	м	Y	125					
191	Tor	16-13	Р	2	86	м	N	275	S	4	18	68	N
192	Тог	16-14	S	6	73	M	Ŷ	3010	S	4	12	61	N
193	IOF	16-14A	S D	0	00 80	м м	T N	1380	s	4	13	76	N
195	Tor	16-15A	Ś	4	51	M	Ŷ	130	0	•			N
196	Tor	16-15B	S	6	70	M	Y	450					
197	Tor	16-16	S	3	12	M	N	920					
198	Tor	16-16A	s	2	70	M	Y	160					
200	Tor	16-16C	S	2	77	M	Ý	275	S	2	13	64	N
201	Тог	16-16D	S	2	13	M	N	210					
202	Тог	16-17	P	2	91	M	N	900	S	2	13	78	N
203	Tor	16-1/A	P	2	91 74	M	N	00 860	S	2	57	34 41	N
204	Тог	16-19	s	2	72	M	Ý	160	3	2	15	01	n
206	Tor	16-19A	S	2	91	M	Ý	70					
207	Tor	16-20	S	2	72	м	Y	275					
208	Tor	16-21	P	2	80	M	N	1025	S	2	12	68	N
209	Tor	16-21A	S	4	58	M	Ŷ	610					
210	Tor	16-22	5 c	2	00 50	т м	Ŷ	1635					
212	Tor	16-24	S	2	69	H	Ŷ	500					
213	Tor	16-25	P	2	86	M	Ň	85					
214	Tor	16-26	S	2	69	м	Y	880					
215	Tor	16-26A	S	2	58	M	Ŷ	295					
216	Tor	16-27	S	2	48	м	Ŷ	160	<u> </u>	-	47	4	
217	Tor	16-28	5	2	74	м м	v	275	5	2	15	01	N
219	Tor	17-1	S	2	68	M	Ý	240					
220	Tor	17-2	P	2	86	М	Ň	75					
221	Тог	17-3	Ρ	2	79	М	N	700					
222	Tor	17-4	S	2	50	M	Ŷ	60 170					
223	Tor	17-4A	ç	2	48	M	Ŷ	170					
225	Tor	17-5	P	2	79	M	Ň	300	S	2	23	56	N
226	Tor	17-5A	P	2	81	м	N	65	S	2	23	58	N
227	Tor	17-6	Р	4	81	M	N	920	S	4	22	59	N
228	Tor	17-6A	S	4	22	M	N	570	_		_ /		
229	IOF	17-68	S	4 2	(Y 70	M	T V	530	5	4	26	53	N
230	Tor	17-00	s	4	50	M	Ý	550					
232	Tor	17-8	s	2	47	M	Ň	120					
233	Tor	17-8B	S	3	30	м	N	220					
233	Tor	17-8A	S	3	34	M	N	250					
234	TOF	17-9	S	4	23	M	N	360					
235	Tor	17-96	P	2	84	M	N	485	s	2	47	37	ม
237	Tor	17-11	s	1	43	M	Ŷ	450	Ū	-		5,	N
238	Тог	17-12	S	2	48	м	Y	250					
239	Tor	17-12A	S	2	41	M	Y	430					
240	Тог	17-13	S	2	13	M	N	300					
241	Tor	17-154	s	2	58	M	Y	003					
243	Tor	17-14A	s	2	16	M	Ň	260					
244	Тог	17-15	S	2	65	м	Y	325					
245	Tor	17-16A	S	3	79	M	Y	15					
246	Tor	17-168	S	3	69	M	Ŷ	845					
247	101	17-165	5	1	/U 87	M M	រ ម	100					
240	Tor	17-17	P	2	78	et M	N	900 900	s	2	16	62	ม
250	Tor	17-18	S	2	41	м	Ŷ	415	5	2	10	02	n
251	Tor	17-19	S	2	42	м	Y	465					

<----- CURRENT PIPELINE INFORMATION ------> <----- PREVIOUS PIPELINE INFORMATION ----->

Seg #	Pipeline Segment#	Material	Diameter	Year Installe	Pressure d	CP	Length in feet	Material	Diameter	Year Installed	Age at Repl.	CP
252 T	or 17-20	S	2	48	м	Ŷ	540					
253	Tor 17-20A	ŝ	2	52	M	Ý	120					
254	Tor 17-20B	ŝ	2	49	M	Ŷ	100					
255 T	ог 17-21	P	2	82	M	Ň	240					
256 T	or 17-22	P	2	85	M	N	1200					
257 T	or 17-23	S	3	49	M	Ŷ	1200					
258 T	or 17-24	s	2	50	н	Y	350					
259 T	or 17-24A	S	3	49	м	Y	170					
260 T	or 17-24B	S	3	56	M	Y	350					
261 T	or 17-25	S	3	49	м	Y	250					
262 T	or 17-25A	S	2	62	м	Y	120					
263 T	ior 17-26	S	2	72	м	Y	260					
264 T	or 17-27*	Р	2	88	м	N	590	S	2	13	75	N
265 T	or 17-27A	S	2	65	м	N	50					
266	Tor 17-278	S	2	13	м	N	70					
267 T	or 17-28	S	1	65	м	Y	325					
268 T	or 17-29	S	2	56	M	Y	635					
269 I	for 17-30	S	6	70	м	Y	1165					
270 T	or 17-31	S	4	23	м	N	775					
271 T	or 17-31A	S	2	49	м	Y	75					
272 1	for 17-318	S	2	47	м	Y	70					
273 T	or 17-31C	S	4	81	м	Y	30	S	4	49	32	N
274 T	ог 17-32	S	8	49	м	N	2180					
275 1	or 17-32A	S	8	81	м	Y	150					
276 T	or 17-33	S	4	59	M	Y	710					
277 1	'or 17-34	S	3	59	м	Y	70					
278 1	for 17-35	S	2	52	м	N	165					
279 T	or 17-36A	S	2	49	м	N	90					
280 T	or 17-36B	S	2	47	м	N	50					
281 T	for 17-40	S	2	22	м	N	190					
							106241					

<----- CURRENT PIPELINE INFORMATION -----> <----- PREVIOUS PIPELINE INFORMATION ----->
APPENDIX D

LEAK DATABASE

MAIN REPAIR	- LEAKS	MODIFIED LEAK REPAIR ORDER DATABASE										binolino i	Pipeline
Leak Order Number	ATLAS 1 SHEET F	EAK PRIO Address or Location	D SIZE (IAMETER inches)Facilit	LEAK	Cause	Repair Type	Material	Year Installed	Detect Date	Action Date	Segment (C=CURRENT =PREVIOUS)
76 -683118	TOR0009	2 ALLEY W/D CRAVENS	200	с, N		۰	. ,	4.	1913	03/01/1976	03/01/1976	TOR09-001	U I
88 - 6UUZU		1 1318 CKAVENS	002	~ •	- •			• •		08/18/1988	08/18/1988	10k09-005	. (
70 - 51170		Z ZUI4 TUKKANLE BLVU 7 1717 DACT AVE		0 M				* ~	1012	01/01/19/8	01/01/19/0	TOP09-004B	20
80 - 2246	TOR0009	3 2014 TORRANCE BLVD	300	1.11		• •		- 1	1956	09/11/1980	04/11/1981	TOR09-004C	۵.
85 - 36725	TOR0009	2 1411 POST AV	200			• •	. .	4	1930	08/14/1985	05/14/1985	TOR09-005	. a.
87 - 12424	TOR0009	1 1444 POST AVE	200				·	4	1922	08/21/1987	08/21/1987	TOR09-006	. ບ
92 - 82157	TOR0009	3 1417 EL PRADO AV	200		-	• •	. .	4	1900	08/06/1992	08/06/1992	TOR09-006	U
82 - 27790	TOR0009	2 1269 SARTORI	200	5	-			4	1912	09/14/1982	09/14/1982	TOR09-006A	U
92 - 82235	TOR0009	3 1265 SARTORI	200	0	-	•	•	4	1912	08/12/1992	08/17/1992	TOR09-006A	U
77 -818772	TOR0009	3 ALLEY E/O CRAVENS & S/O POST	200	2	-	~	*-	4	1912	01/01/1977	01/01/1977	TOR09-006A	υ
75 -744457	TOR0009	1 1414 CEAVENS	200	2	-	•	*	4	1912	03/01/1975	03/01/1975	TOR09-006A	ы С
80 - 2070	TOR0009	2 1414 CRAVENS	200	~	-	-	-	4	1950	10/22/1980	10/22/1980	TOR09-009	ပ
89 - 60031	TOR0009	1 923 VAN NESS	200	2		-	-	4	1927	08/22/1989	08/22/1989	TOR09-013	۵.
77 -844488	TOR0009	3 921 VAN NESS	200	~	-	•	e	2	1972	12/01/1977	12/01/1977	TOR09-013	٩.
83 - 30134	TOR0009	3 1971 TORRANCE BLVD	200	2		-	•	4	1926	07/21/1983	07/21/1983	TOR09-013A	ပ
80 -095788	TOR0009	3 925 VAN NESS	200	5	-	-	-	4	1936	12/18/1980	07/18/1981	TOR09-013B	U
80 -09578A	TOR0009	1 925 VAN NESS	200	~	-	-	-	4	1936	12/18/1980	12/18/1980	TOR09-013B	U
84 - 51241	TOR0009	1 923 VAN NESS AVE	200	5	-	-	-	4	1939	03/07/1984	03/07/1984	TOR09-013B	U
84 - 38639	TOR0009	1 923 VAN NESS AVE	200	2	-	ſ,	•	4	1939	03/07/1984	03/07/1984	TOR09-0138	υ
77 -876992	TOR0016	3 1500 CABRILLO	009	9	-	-	-	4	1913	08/01/1977	08/01/1977	TOR09-014	ፈ
81 - 11919	TOR0009	3 1502 CABRILLO AVE	009	6	-	•	•	4	1925	09/28/1981	02/28/1982	TOR09-014	۵.
79 - 994782	TOR0009	1 908 VAN NESS	600	\$	-		-	4	1912	07/01/1979	07/01/1979	TOR09-014A	U
89 - 40448	TOR0009	3 N/E COR VAN NESS & MULLEN AV	600	6	-	*	•	4	1912	08/22/1989	08/22/1990	TOR09-014A	ы С
89 - 40451	TOR0009	3 1820 TORRANCE BL	200	2	-	•	•••	2	1947	08/24/1989	08/24/1990	TOR09-016A	U
76 -683120	TOR0009	1 1820 TORRANCE BLVD.	200	2	_	-	•	2	1947	02/01/1976	02/01/1976	TOR09-016A	U
86 - 90074	TOR0009	3 N/W I/S 213TH ST & BOW	300	m	-	-	4	2	1951	03/21/1986	03/21/1987	TOR09-023	ы
84 - 56813	TOR0009	3 N/W CORNER WESTERN & 213TH ST	800	8	•	-		5	1945	03/12/1984	02/12/1985	TOR09-026	ပ
91 - 15032	TOR0009	1 NW 213 & WESTERN	800	8	••	-	-	2	1945	08/19/1991	08/19/1991	TOR09-026	U
86 - 26051	TOR0010	2 2512 TORRANCE BL	200	2		-	-	5	1935	01/06/1986	01/06/1986	TOR10-004	ပ
82 - 21994	TOR0010	3 2558 TORRANCE BLVD	200	2	-	-	-	2	1935	06/11/1982	04/11/1983	TOR10-004	U
84 - 51787	TOR0010	3 1324 DATE AVE	200	2	-	-	-	2	1927	08/28/1984	08/28/1984	TOR10-005A	υ
85 - 36251	TOR0010	2 1326 DATE	200	2	-	-	-	4	1927	05/31/1985	06/14/1985	TOR10-005A	ပ
88 - 20573	TOR0010	3 2515 EL DORADO ST	200	2	-	*	-	4	1926	12/01/1988	02/01/1989	TOR10-007A	ပ
75 -686462	TOR0010	3 1606 CRENSHAW BLVD	200	2	-	*	-	4	1926	02/01/1975	02/01/1975	TOR10-008	٩.
84 - 39655	TOR0010	2 2555 SONCHA	200	2	-	•	•	M	1951	11/26/1984	11/26/1984	TOR10-008A	U
84 - 39583	TOR0010	Z Z555 SONOMA	200	2	.	4	•	M	1951	11/14/1984	11/14/1984	TOR10-008A	U
82 - 22002	TOR0010	3 1512 DATE AVE	200	2	-		*	4	1924	06/11/1982	04/11/1983	TOR10-009	٩
78 -936778	TOR0010	3 2555 SONOMA	200	2	-	-	-	4	1927	07/01/1978	07/01/1978	TOR10-009	۵.
84 - 51799	TOR0010	3 1508 DATE AV	200	2		-	-	2	1924	08/30/1984	08/30/1984	TOR10-009	٩
88 - 20569	TOR0010	3 2371 TORRANCE BLVD	300	m t-		-	-	4	1927	12/01/1988	08/01/1989	TOR10-012C	U
84 - 31444	TOR0010	1 1027 ACACIA	300	m L	-	-	-	4	1927	11/15/1984	11/15/1984	TOR10-012C	ပ
86 - 60325	TOR0010	3 2413 TORRANCE BLVD	300	m	•	-	•	4	1927	11/26/1986	10/26/1987	TOR10-012C	u
82 - 28741	TOR0010	3 1104 AMAPOLA	200	2	-	•	•	4	1924	06/09/1982	06/09/1982	TOR10-012D	۵.
78 -976417	TOR0010	1 2313 TORR BLVD	200	~	-	Ļ	•	4	1926	06/01/1978	06/01/1978	TOR10-012D	٩
78 -933750	TOR0010	2 2305 TORR BL.	200	2	•	-	-	4	1924	07/01/1978	07/01/1978	TOR10-012D	٩
78 -976416	TOR0010	1 2305 TORR BLVD.	200	2	-		• •	4.	1926	07/01/1978	07/01/1978	TOR10-012D	۹. ۱
78 -976419	TOR0010	1 2306-2308 TORR BLVD	200	~		- •	- •	4 (1921	06/01/1978	06/01/19/8	TOR10-012D	۵. ۱
10007 - 20		1 11U5 PUKIULA AVE Z 1010 BEERV	200	ი ი		- •	- •	とく	001	U6/U1/19/0	10/01/19/0	10K10-0126	ه د
34 - 1076		D IUIY BEELD	200	- v	-	~	-	t	0741	+0K1 /C1 /11	4041 /01 /71		L

Preceding page blank

MAIN REPA	IR LEAKS	- MODIFIED LEAK REPAIR ORDER DATABASE										binolino	Pipeline
Leak Ordé Number	PRICES SHEET	LEAK PRIO Address or Location S) 321s	IAMETER inches)Facilit	y In	Cause	Repair Type	Material	Year Instal led	Detect Date	Action Date	Segment	CC=CURRENT P=PREVIOUS)
78 - 99016 56 - 3220	E TOROUTC	1 811-815 BEECH AVE. 2 016 peecu			**		- •	~ t	19291	J6/U1/19/8	06/01/19/8	10K10-013	<u>م</u> د
200 - 200 78 - 00016		1 017 REFCH AVE		10				t 1	1028	06/01/01/02	06/01/1978	TOP10-013 TOP10-013	L 0
82 - 2865	9 TOROD10	1 812 CRENSHAW BLVD	200	10	• •	• •		• • •	1928	06/01/1982	06/01/1982	TOR10-013	. a.
82 - 2868	5 TOR0010	1 811 BEECH AVE	200	1			•	4	1928	05/28/1982	05/28/1982	TOR10-013	. a .
84 - 3138	5 TOR0010	3 1020 CRENSHAW BLVD	200	5	-	-	-	2	1928	10/26/1984	10/26/1984	TOR10-013	۵.
78 - 2526	6 TORO010	3 1007 BEECH AVE	200	2	-	-	-	4	1928	10/01/1978	10/01/1978	TOR10-013	۵.
78 -99017	'9 TOROO10	1 1015-1023 ACACIA	200	2	-	-	-	4	1929 (06/01/1978	06/01/1978	TOR10-014	٩
82 - 2195	15 TOR0010	3 0 ACACIA AVE	200	2	•	-	-	4	1927	06/11/1982	04/11/1983	TOR10-015	٩
82 - 2871	1 TOR0010	1 1103 AMAPOLA AVE	200	2	•	-	-	4	1924 (06/03/1982	06/03/1982	TOR10-016	ပ
78 - 2527	3 TOR0010	3 2310 SIERRA ST	200	2	-	-	-	4	1924	10/01/1978	10/01/1978	TOR10-016	J
82 - 2877	78 TOR0010	1 1103 AMAPOLA	200	2	-	-	-	4	1925 (06/17/1982	06/17/1982	TOR10-016	ပ
84 - 735	19 TOR0010	3 1006 SIERRA PL	200	2	-	-	-	4	*1925	12/05/1984	10/05/1985	TOR10-017	ပ
84 - 735	10 TOR0010	2 1012 SIERRA PL	200	2	-	-	-	4	1925	12/05/1984	12/19/1984	TOR10-017	പ
396 - 06	13 TOR0010	1 918 S COTA AV	200	2	-	-	~	4	1900	03/05/1990	03/05/1990	TOR10-018/	U J
90 - 261	8 TOR0010	3 I/S SIERRA & PORTOLA AV	200	2	-	-	-	4	1917	10/31/1990	07/31/1991	TOR10-018/	U
77 -86615	1 TOROO10	3 I/S COTA & SIERRA ST	200	2	-	-	-	4	1917	7791/10/10	01/01/1977	TOR10-018/	U
78 -93364	0 TOROO10	2 917-19 PORTOLA	200	2	,	-	-	4	1917	07/01/1978	07/01/1978	TOR10-018/	U
88 - 2057	0 TOROO10	3 SONOMA & ALLEY W/COTA	200	2	-	-	-	4	1913	12/01/1988	08/01/1989	TOR10-0191	U
88 - 2057	'2 TOR0010	3 S/E COR ELDORADO & ALLEY W/O COTA	200	2	-	-	-	4	1913	12/01/1988	08/01/1989	TOR10-019	U
86 - 3775	9 TOR0010	1 1417 COTA AVE	200	2	-	-	-	4	1913	01/06/1986	01/06/1986	TOR10-0191	U J
87 - 4031	8 TORO010	1 1507 COTA AVE	200	2	-	-	-	4	1913	01/26/1987	01/26/1987	TOR10-0191	U
87 - 842	7 TORO010	3 1421 COTA AVE	200	2	•	-	-	4	1913	01/27/1987	01/27/1988	TOR10-0190	ပ ၂
83 - 5013	4 TORO010	2 1411 COTA AVE	200	2	•	-	•	4	1913	02/14/1983	02/14/1983	TOR 10-0190	U
82 - 2874	9 TORO010	2 1417 COTA AVE	200	2	***	-	-	4	1913	06/09/1982	06/09/1982	TOR 10-0190	ပ ၂
82 - 2200	11 TOR0010	3 INT ALLEY W/O COTA & N/PL ELDOR	200	2	•	-	-	2	1913	06/11/1982	06/11/1983	TOR10-0190	ပ
86 - 342	16 TORO010	3 1421 COTA AVE	200	~		·	•	м	1966	03/21/1986	03/21/1987	TOR10-0190	ပ ၂
83 - 2995	5 TORO010	3 1225 COTA	200	2	Ļ	* -	•	4	1921	05/27/1983	05/27/1983	TOR10-0190	ပ ၂
85 - 3694	Z TOR0010	3 1403 COTA	200	2	-	*	.	4	1913 1	09/18/1985	09/18/1985	TOR10-0190	J
83 - 5014	1 TOR0010	2 1418 AMAPOLA AVE	200	2	4	~	•	4	1913 1	02/15/1983	02/15/1983	TOR 10-0190	ں ب
78 - 93383	2 TORD010	2 1103 PORTOLA AVE	200		-	•	• •	4	1917	06/01/1978	06/01/1978	TOR 10-020/	
84 - 3971	0 TOR0010	3 1103 PORTOLA AVE	200	. 6	-	•		4	1917	12/03/1984	12/03/1984	TOR 10-020/	U U
78 -90963	7 TOR0010	3 1103 PORTOLA AV	200	2	1	-	•	4	1978	09/01/1978	09/01/1978	TOR 10-020/	0
78 -97641	3 TOR0010	1 1007-1011 ARLINGTON	200	2	-	-		4	1916	06/01/1978	06/01/1978	TOR10-021	۵.
78 -97642	4 TOR0010	1 921 ARLINGTON	200	2	-	•		4	1916 1	06/01/1978	06/01/1978	TOR10-021	۵.
78 - 81873	5 TOR0010	2 824 PORTOLA ALLEY E/O	200	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	5	-	-	4	1919 (06/01/1978	06/01/1978	TOR10-021	۵.
82 - 2171	7 TORO010	3 1303 BEACH AVE	200	1		•	-	4	1927	06/16/1982	09/16/1982	TOR 10-024/	. a.
82 - 2745	6 TOR0010	3 1303 REFCH AVE	200	1	-	•	-	7	1927	06/15/1982	06/15/1982	TOR 10-0241	. a
82 - 5001	2 TOP0010	2 1230 CPENCHAU BI VN	200	10	• •	• •	. 4-	7	1027	12/17/1082	12/17/1082	TOP 10-0277	. a
82 - 2877	Z TOROD10	T 120 STERVISH STAT	200	10	• ←		• •-	. 4	1027 (74/16/1082	06/16/1982	TOR10-0241	. a
70 77207		U 1000 ULEVII 74E J 2347 AVAVIA AVE	200	- -	- •		• •	• ~	1017	10/ 10/ 10/ 10/ D	00/ 10/ 10/ 10/ 00 00 00 00 00 00 00 00 00 00 00 00 0		- 6
10000X- 0/		Z DZIT ALALIA AVE			- •			5 ~	1414	10/01/17/0	UO/UI/ 17/0	10410-067	1 C
1202 - 8/		5 2414 SUNUMA AVE		- •	- •	- •	- •	<i>t</i> ~	1741	10/01/17/01	10/01/19/0		ג נ
2006- 8/	6 TOKUU10	I IZIT-IZZI ACACIA AVE		~ •	- •		- •	• t	1761	8/61/10/90	0201/10/00	10K10-025	<u>م</u> ،
(6701 - 6/		T OF OOT DOTTON & ALACIA N/U SUNUMA	B	~ •	- •	- •	- •	t •	1761	6/61/10/01	6/61/10/01	C20-01 X01	. (
1262 - 81		5 SE CUK SUNUMA & ALT W/ACACIA		2	- •	- •		• t	1761	8/61/10/60	8/61/10/60	C2U-01 XUI	2
77 -91577	8 TOR0010	3 2414 SONOMA AVE	200	2	.		• •	4	1925 (22/01/1977	2261/10/20	TOR10-025	۵.
81 - 1132	4 TOROO10	3 1423 ACACIA AVE	200	~~~	r= 1	، - ،	- I	• •	1927	05/14/1981	07/14/1981	TOR 10-025	۵. ۱
78 -99016	8 TOR0010	1 2415 SONOMA AVE	200	2			2	4.	1927	06/01/19/8	06/01/19/8	TOR 10-025	۵. ۱
84 - 3966	6 TOR0010	1 1503 MADRID	200	2	-	•	•	4	1927	11/27/1984	11/27/1984	TOR10-026	۵.
84 - 5184	0 TOR0010	3 1503 MADRID AVE	200	2	-		•	2	1927 (09/11/1984	09/11/1984	TOR10-026	٩

MAIN REPAII	- SAEAKS	MODIFIED LEAK REPAIR ORDER DATABASE										Dinalina	Pipeline info
Leak Order Number	ATLAS L SHEET P	EAK RIO Address or Location) SIZE (IAMETER inches)Facili	LEAK ty In	Cause	Repair Type	Material	Year Installec	Detect I Date	Action Date	Segment Number	CC=CURRENT P=PREVIOUS)
		· · · · · · · · · · · · · · · · · · ·									02.012 102 00		
79 - 51442 78 - 51442	TORU010	5 1515 AMAPOLA AVE 7 1207 AMAPOLA AVE		20		- •		* *	1021	02/01/10/20	10/01/19/9	TOP 10-02/	<u> </u>
82 - 28757	TOROO10	1 1217 AMAPOLA AVE	200	5	- - -	• •	. –	• •	1921	06/11/1982	06/11/1982	TOR10-027	. œ
84 - 31233	TOR0010	3 1229 PORTOLA	200	2	-	-	~	4	1921	09/06/1984	09/06/1984	TOR10-035A	م
82 - 27434	TOR0010	1 1303 PORTOLA AVE	200	2	۴-	-	-	4	1921	06/10/1982	06/10/1982	TOR10-035A	٩.
82 - 27433	TOR0010	1 1215 PORTOLA AVE	200	2			•	4	1921	06/10/1982	06/10/1982	TOR10-035A	۹.
84 - 39705	TOR0010	1 1217 PORTOLA	200	2	- ·		•	4	1921	12/03/1984	12/03/1984	TOR10-055A	a . i
82 - 28759	TOR0010	3 1229 PORTOLA AVE	200	21	- ·	•- •		4.	1921	06/11/1982	06/11/1982	TOR10-055A	<u>م</u> ۱
84 - 39659	TOR0010	1 1217 PORTOLA	200	~ ~		· 1	- •	4.	1920	11/2//1984	11/2//1984	10R10-055A	.
78 - 990154	TOR0010	1 1307 PORTOLA AVE	200	2			- •	• •	1761	02/01/19/8	8/61/10/90	10410-01201	י נ
83 - 50544	TOR0010	S ALLEY W/O PORTOLA & ELDORADU AVE		N (- •		• •	1761	2861/10//0	2061/10/20	TOP10-020	י נ
84 - 59655		1 1228 AKLINGION AVE		vr	- •	- •		4 ~	9101	11/20/1904	11/20/1904	TOP10-012/	יכ
20 - 224 H4		2 ALLET N/U ENGRALIA S/U IURKANLE 2 AF77 PART ANT		4 6				* ~	10101	2041 / 11 / 20	101 / 107 / 20 / 20 / 20 / 20 / 20 / 20 / 20 /		ט נ
104000- 0/		2 1/10 EUCADEIA AVE		n r		- •		t ~	1017	10/21/10/00	06/30/1001	TOP10-01201	ە ر
72668 - 60		J 1416 CNUARLIA AV 7 1//7 s fucdarta av		א ר				r -1	10001	08/17/1002	08/17/1992	TOR10-039A) ن ا
12000 - 20	1000010	Z 1/// ENCDACIA AVE		א ר				17	1012	02 / 04 / 1988	05/04/1988	TOP10-0304	د
75396 - 84		T 1/17 ENGDAFTA AVE		א ר				r <	1012	12/01/1078	12/01/10/28	TOP 10-030) د ا
8CY02 - 78		1 7148 TODANUT AVE		<u>ہ</u> د					1021	11/20/1086	11/20/1984	TOP10-0400	۰ <u>م</u>
0704 - 40		1 2004 TORRANCE DL		J C				r ~	1021	11/20/108/	11/20/108/	TOP10-01/01	. ۵
84 - 39029		1 ZZUD TUKANLE BL		יר		- •		t ~	1721	11/20/1704	11/20/1708/		. 0
07070 - 70000		2 220/ TORRANCE BL		u r				~ t	0004	11/2/11/12/11	10/1/12/11		
5405 - 48 2002 - 48	10KUU10	Z ZSU4 TUKRANCE BLVU		טר		- •	- •	t ~	0761	4061/C2/11	4041/CZ/11	TOB10-040L	- 0
		1 2304 TUKK BLVU		יר		- •	- •	* ~	0761	11/2/1/1704	10/1/12/11		
84 - 59651	100010	1 2508 TOKK		2		- •		- t	8741	4961/07/11	4061/CZ/11		.
84 - 59655	TORU010	1 2212 TORRANCE BLVD		~ ~		- •		• •	1761	11/21/1984	11/21/1904		
18 -9/6418	10KUU10	1 1215 CUIA AVE		vr		- •		7 t	1241	10/01/19/0	10/01/17/0		- c
0//67 - 48		2 23UD TUKANCE BLVU		u r		- •		v ~	1016	12/ 12/ 1704/	11/20/108/		
10070 - 40		7 24// #ODRKANCE BL		ч r				t ~	(C)	11/20/1904	10,001,1904		. 0
(1707 - 8/		2 2300 TURKANCE BLVU 7 2300 TODAAVEF BLVD		יי				~ t	1022	08 /01 / 1075	08/01/10/01	TOP10-040E	. ۵
70 000 02		J 2473 TODA PI		<i>1</i> r		- •	- 1	t ~	1022	00/ 01/ 10/ 20 0/ /01/ 1028	02/01/10/00	TOP10-01/01	- 0
78 074070		1 21/2 TURK BL. 1 21/24 TODD DIVD		<i>u</i> r			۰ -	t 7	1022	10/01/10/8	10/01/1978	TOR10-040F	. 0.
10017 01		7 316/ TONDANCE DIM		ı c				* ~	1022	10/01/1078	10/01/1078	TOP 10-0405	. ۵
75 - 485003		J 2134 IUKKANCE BLVU 7 3308 TODDANFE BLVD		<i>4</i> C				7 4	1022	08/01/1075	08/01/1975	TOR 10-040F	. a
201570- 21		1 2172 TADD BI	200	10					1022	04/01/1978	04/01/1978	TOR 10-040F	. ല
78 -000155		1 221/ TODO BLVD	200	10	- -			1	1021	04/01/1078	06/01/1078	TOR 10-040F	. D
101044 - 01 0374 - 10		1 CELT TORN BLYD 1 om tobbande bi 137 11711 Amadon A		10			- •	1 ~	1021	00/11/1086	00/11/1086	TOP 10-102	۵.
0741 - 40 87 - 77/57		1 ON TOPPANCE BLID AL MUNICIPAL	200	10				7	1021	09/11/1086	09/11/1984	TOR10-102	۵.
80 - K0031	1000015	2 172/ DATE V CADEON	200	יי	• •			. 7	1027	08/17/1080	08/17/1989	TOR15-0034	. പ
77 - 84.445	1000015	1 1724 DATE AVE	200	10		• -	• •-	. 4	1027	08/01/1977	08/01/1977	TOR15-0034	0
104440- 11		1 1/20 UNIE AVE 1 receveral di 9 toiedo st		ıα					1020	05 /01 /1078	05/01/1978	TOR15-004	۰ م
10/20/ Q/		I CKENDRAW BL. & ICLEUC JI. 7 VII ODV IFFIRGOVICOFINGUAL					- •	1~	10/1	01/11/10/00	00/15/1000	10015-00/1	
89 - 40547		J N/E UKN JETTEKSUN/UKENSHAW 1 17/4 ITTTTTOTIL		0、	- *	~ ~		t <	10201	100/101/00	100/12/1081	100-11-001	، د
*cool - 10		I 2341 JETTERSON		, ,		- •	- •	t ~	1720	00/16/1701	01 /01 /1075	TO0 15-008	, c
20617 - 27		1 UAK SI & JEFFEKSUN		t -			- •	4 (1261	C141/10/10	01/10/10		. c
51955 - 58		1 Z54 TJEFFEKSUN		t •				N -	1261	COV1 /01 /CU	COVI /01 /CU		. נ
77 -915774	TOR0015	1 N/E COR OAK & JEFFERSON ST	400	4.		- •	1	4 (1761	//61/10//0	1161/10/10	10K15-0084	י נ
86 - 38/84	TORUUTS	Z 2341 JEFFERSON SI	004 4 0 0	t.			- •	~ ~	1771	U6/U4/1900	U0/ 10/ 1700	10K12-000A	י נ
cc25 - 58	TORUUN	3 2305 JEFFERSON AVE		4 (- •	- •	•	- t	1741	0061/11/20	10/11/11/20	10K12-000	ם נ
COUOSS- 11		5 1655 ALALIA 7 771111 1273 ALALIA		46				t <	1027	11/11/10/21	7101/06/70	TOP15-010	- 0
CK177 - 58		3 BEHIND 1022 ALALIA	500	J	_	-	-	r	1761	10/ 22/ 1701	1041 661 1201		-

.

MAIN REPAIR	- LEAKS	MODIFIED LEAK REPAIR ORDER DATABASE										Pineline	Pipeline info
Leak Order Number	ATLAS L	EAK RIO Address or Location (DIA SIZE (ir	METER Iches)Facility	LEAK	Cause	Repair Type	Material	Year Installec	Detect 1 Date	Action Date	Segment	(C=CURRENT P=PREVIOUS)
						1							
84 - 51085	TOR0015	1 1614 ACACIA AV	200	~ ,	- •			40	1927	01/04/1984	01/04/1984	TOR15-010	۵. ۱
160cl - 16	1000015	5 1621 S AMAPULA AVE		- .		- *		2	000	10/01/01/01	1661/00/01		ى ر
14// - 16		1 1342 PUST AVE		- -		- •		t v	1012	07/07/1081	00/07/1081	TOP 15-013	י נ
86 - 3132		1 1/2 ALEY W/PAPCON ST & ALLEY U	200	1 C				- 1	1024	01/20/1986	01/20/1086	TOR15-015	o د
122U2 - 00		T 1/3 ALLET N/ CANSON 31 & ALLET F	200	1 v 				- 7	1024	08/28/1080	02/28/1000	TOR15-015	, د
	TOPOOLS	2 1677 AMADRIA	200	1 C		• -		7	1074	06/23/1086	06/23/1086	TOR15-015	
00 - 20070 0 - 770ED		J 1UCI AMARULA J 1/6 ALLEV U/CABEON CT 8 ALLEV		- -				r ~	102/	01/20/1086	01/20/1086	TOP 15-015	ט נ
NC0/C - 00		Z 1/S ALLET N/CARSON SI & ALLET		- •	- •			, 1	1721	10011/07/10	0041/07/10		ז כ
84 - 39787		5 1512 EL PRADU AVE	007	- •	- •		- •	~ ~		+961/+1/21	12/14/1984	TOK 10-001	۲ ו
84 - 30764	TOR0016	Z 1515 ARLINGTON	007	~	- •	•) t	2161	4861/60/SU	02/09/1984	10K 16-001	a. 1
80 - 1869	TOR0016	5 1528 EL PRADO		- •	- •			N (1914	U0/18/1980	1961/91/10	TOK 10-001	a. 6
84 - 59/86		S 1512 EL PRADU AV		~ 1	- •		- •	2	004	12/ 14/ 1904	12/ 14/ 1904		2 (
83 - 50552	TOR0016	S 1615 CRAVENS AVE	200	ר י י ר	- (.	- •	4.	21.61	C041/47/00	00/24/1985		. ن
81 - 18595	TOR0016	2 1617 CRAVENS	300	m	2	~	-	4	1912	06/03/1981	06/03/1981	TOR 16-004	ပ
82 - 21742	TOR0016	2 1613 1/2 CRAVEN	300	с М	-	•	-	4	1912	06/07/1982	06/21/1982	TOR16-004	ы
90 - 10100	TOR0016	3 1613 CRAVENS AV	300	3	-	-	-	4	1912	12/27/1990	01/27/1991	TOR16-004	U
77 -876993	TOR0016	3 1600 CABRILLO .	600	6 1	-	•	-	4	1913	08/01/1977	08/01/1977	TOR16-010	٩.
83 - 50588	TOR0016	3 N/E CORNER BORDER AVE & CARSON	600	6	-		-		1941	07/13/1983	07/13/1983	TOR16-012	u
77 -908168	TOR0016	1 I/S CAZSON/BORDER AT RR CROSSING	600	6 1	-	-	-	2	1941	06/01/1977	06/01/1977	TOR16-012	U
83 - 37786	TOR0016	2 N/E CORNER BORDER AVE & CARSON AV	600	6 1	•		-		1941	06/29/1983	07/29/1983	TOR16-012	ပ
80 - 8376	TOR0016	1 1905 ABALONE	400	4	-		-	4	1913	03/18/1980	03/18/1980	TOR16-015	٩
84 - 56785	TOR0016	3 1740 ARALONE AV	400	4	ę	-	•	M	1913	05/15/1984	02/15/1985	TOR16-015	٩
86 - 3583	TOROD16	3 1805 ABALONE	400	4			•	14	1923	05/01/1986	05/01/1987	TOR16-015	. ۵
76 - 8/.8781	TOP0016	2 1720 ARAI OVE AVE	007	. 7	• ~	• -	• •	. 7	1013	09/01/1076	00/01/1076	TOP16-015	. 0
101010-01		Z 1770 ADALONE AVE		* ~	- •		- •	t ~	2101	08/01/10/0	08/01/10/00	TOP16-015	. 0
C130C1 - 21		2 1005 ADALONE 2 1005 ADALONE		• •		- •		t ~:	1012	07/01/10/20	07/01/10/20	TOP16-015	. 0
0010to-01		Z 1007 APALONE		- -		- •		t ~	101	05/01/1086	05 /01 /1087	TOP 16-015	
+011 - 00		Z DDDOFTER OF 4777 ADALONE		- •		- +		t ~	1014	00/1/10/00	C801/ 70/ 70		
/0011 - 10		3 UPPUSITE UP 1744 ABALUNE	004	- •	- •		- •	~ t		1061/10/00	10,04/1905		L 6
000011 - 18		2 1009 ABALUNE	4 C C		- (- 、	- t	01.61	1061 / 40 / 00	10/14/140		2.4
1/2051- 6/		3 182U ABALUNE	400	- •	2		• t	• t		06/01/10/20	06/01/10/80		a. 1
87 - 11771	TOR0016	2 1915 ABALONE AVE	400	4.	- •	- •	- •	4.	5191	05/08/198/	05/08/1987	TOR16-015	۵. ۱
77 -876989	TOR0016	3 1905 ABALONE AVE	400	4	1		- •	4	1915	1/61/10/80	1/61/10/80	TOR16-015	٩.
79 -150270	TOR0016	3 ABALONE AVE 60' N/N PL 220TH ST	400	4	, 1	•		4	1913	08/01/19/9	08/01/1979	TOR16-015	٩.
88 - 20159	TOR0016	3 1907 ABALONE AVENUE	400	4	•	-	•	4	1951	05/04/1988	06/04/1988	TOR16-015	U U
86 - 38623	TOR0016	2 1962 CARSON	300	м М	-	•	-	4	1968	05/08/1986	05/08/1986	TOR16-016	ပ
76 -844796	TOR0016	2 1720 GRAMERCY AVE	300	м Т	-	•	-	4	1912	07/01/1976	07/01/1976	TOR16-016	U
73 -570305	TOR0016	1 1719 GRAMERCY AVE	300	м Г	-	•	-	4	1912	07/01/1973	07/01/1973	TOR16-016	J
88 - 55465	TOR0016	1 1715 CABRILLO AVE	300	5	,	-	9	4	1912	05/02/1988	05/02/1988	TOR16-016	U
81 - 11368	TOR0016	3 1/S CARPILLO & ALLEY S/CARSON	300	M	,	•	•	4	1912	06/04/1981	07/04/1981	TOR16-016	
78 -035047	TOPOOLO	3 1/5 DE ALLEY & DE CARSON /ANDRED A		- - -	• •			7	1012	08/01/1078	08/01/1078	TOR16-016	, د
37/012 32		1 2020 POPRE			• -		• •		1012	07/01/1075	07/01/1075	TOP 16-016	
10+X11- CJ		1 2027 BURVER		- -	- •		- •	* ~	1012	01/01/10/10	11/11/10 14/10/10	1010-21004	J (
- 0000 - 00		2 2113 BURUEK AVE	200	- • • •	- *		- •	t ~	1014	10/ 10/ 1700	10/ 10/ 1700	101 10 - 01 101 7 10 - 21 10 - 4	ז נ
04001 - 04		2 1023 BURNEK AV		- •		- •	- •	• t	1 2 2 2	0461/41/21	1661/61/21		×. (
86 - 58510		5 1820 CABKILLO		- •		- •		t •		00/1 /07 /CD	0061/07/00		.
90 - 10088	100016	S ZIBIH SI & BUKDEK AV	200	 	- ,		- ,	t •	CI 61	0661/61/21	0661/61/21		-
90 - 2300	TOR0016	3 ALLEY W/O BORDER AV	200		- 1	• ·		4	1913	03/12/1990	11/12/1990	TOR16-017	۵.
78 -989851	TOR0016	2 1914 U 218TH ST.	200	~	, 1	1	•	4	1968	10/01/19/8	8/61/10/01	TOR16-021	۹. ۱
76 -800623	TOR0017	Z 2063 LINCOLN AVE	200	~ `	- •	•	•	• •	9761	9/61/10/90	0/01/19/0	TOR17-005	d. 1
86927 - 67	TOR001/	3 2409 ARLINGTON AVE	200		- •			•t	0771	V/VI/IV/SU	V/VI/IU/SU	TOK1/-005	<u>р</u> с
75 - 744 104	TORUUTY	3 2409 ARLINGTON AVE	200	7	-		-	t	C741	C141/10/01	C/41/10/01	CUU-11X01	2

MAIN REPI	AIR LEAKS	- MODIFIED LEAK REPAIR ORDER DATABASE										Pineline	Pipeline info
Leak Ord: Number	er ATLAS SHEET	LEAK PRIO Address or Location	SIZE	DIAMETER (inches)Facility	y In	Cause	Repair Type	Material	Year Installed	Detect Date	Action Date	Segment Number	(C=CURRENT P=PREVIOUS)
7264	31 TOR0017	3 S/E COR SANTA FE & ARLINGTON	400	- +	- •		+	4 ~	1922	04/01/19/9	04/01/19/9	TOP 17-006	בנ
30482 - 78	X0 10KUU17	7 2 N/E LINGION & FLAZA UEL AMU 7 2 N/E FOD OF ITUFOIN 2 ADITUFTON AV		t 4	- 7			* 4	1022	03/23/1986	03/23/1984	TOR17-0064	، د
88 - 788 88 - 788	33 TOROD17	7 3 2410 ARI INCTON AVE	400		• •			4	1922	06/20/1986	06/20/1986	TOR17-006/	ں ا
70 -10570	7 TOROD17	2 2409 ARLINGTON	400	4				4	1922	10/01/1979	10/01/1979	TOR17-006/	U
78 -97604	7 TOR0017	1 2202 ARLINGTON AVE	400	4	-		-	4	1956	11/01/1978	11/01/1978	TOR17-006E	۵.
90 - 235	58 TOR0017	7 3 2030 SANTA FE AVE	300	1		*	-	4	1930	05/11/1990	02/11/1991	TOR17-0085	ပ
83 - 299	58 TOR0017	2 1966 1/2 PLAZA DEL AMO	200	- 2	-		-	4	1941	05/13/1983	05/13/1983	TOR17-012/	U
78 -0074	55 TOROU17	7 2 1912 PI A7A DFI AMO	200	10			-	4	1916	09/01/1978	09/01/1978	TOR17-014/	ပ ျ
70 - 776	59 TOROD17	7 3 1012 PLAZA DEL AMO	2002	1		• •	• •	4	1916	03/01/1979	03/01/1979	TOR17-014/	U
85 - 100	55 TOROD17	7 3 1912 PLAZA DEL AMO	200			· -	• –	4	1965	05/15/1985	01/15/1986	TOR17-015	ں ا
83 - 2265	71 TOROD17	7 3 2303 ANDRED	200	2	-	-	-	4	1965	05/17/1983	01/17/1984	TOR17-0168	ں
78 -93384	45 TORO017	7 1 ALLEY S/O PLAZA DEL AMO	200	2	-	-		4	1916	07/01/1978	07/01/1978	TOR17-017	٩
78 - 9904(12 TOR0017	7 1 ALLEY S/O PLAZA DEL AMO &	200	5	-		-	4	1916	07/01/1978	07/01/1978	TOR17-017	٩
78 -91404	46 TOR0017	7 1 2408 CABRILLO	200	2	-	-	-	4	1916	06/01/1978	06/01/1978	TOR17-017	٩
78 - 9902	56 TOR0017	7 1 1880 PLAZA DEL AMO	200	5	-	-	-	4	1916	06/01/1978	06/01/1978	TOR17-017	۵.
83 - 226	27 TOR0017	7 3 835 MARINETTE AVE	300	∾	-	-	*	2	1949	05/17/1983	02/17/1984	TOR17-023	ပ
78 - 9902	74 TOR0017	7 2 2219 BORDER AVE	200	2		•	•	4	1913	06/01/1978	06/01/1978	TOR17-027	٩
77 -8769	24 TOR0017	7 3 2213 BORDER AVE	200	2		-	•	4	1913	08/01/1977	08/01/1977	TOR17-027	٩.
78 - 9902	57 TOR0017	7 Z Z230 CABRILLO	200	2	÷	÷	-	4	1913	06/01/1978	06/01/1978	TOR17-027	٩.
78 -91404	.9 TOR0017	7 1 2207 BOLDER AVE	200	2		,	•	4	1913	06/01/1978	06/01/1978	TOR17-027	۵.
84 - 517	75 TOP0017	7 3 1780 PLAZA DEL AMO	400	- 7			•	4	1923	07/31/1984	08/31/1984	TOR17-031	U
79 -15024	4 TOR0017	7 3 22500 BLOCK WESTERN	200	. 6	. 4		-	· N	1949	09/01/1979	09/01/1979	TOR17-031	U
77 -90815	20 TOR0017	7 1 22501 VESTERN AVE	200			. .	-	2	1949	7701/1977	7791/10/70	TOR17-031/	U J
83 - 2260	15 TOR0017	7 3 22501 WESTERN	400	- 1			-	2	1981	06/15/1983	07/15/1983	TOR17-031	ບ
70 -1502/	55 TORO017	7 3 1700 BLOCK PLAZA DFL AMO	800	8	•	÷	•	2	1949	09/01/1979	09/01/1979	TOR 17-032/	U
81 - 1858	31 TOR0017	7 3 NORTHWEST CORNER PLAZA DEL AMO &	200	. ~		•	-		1952	06/02/1981	06/02/1981	TOR17-035	υ
85 - 10/	58 TOR0017	7 3 1921 222ND ST	200	2	•	•	•	2	1922	05/15/1985	02/15/1986	TOR17-040	U
71 - 7560	14 TORODO	2 1 1 FUELLYN & TORBANCE BL	2002			· ~	•	4	1931	05/01/1971	05/01/1971	TOR09-001	٩
77 - 46019	77 TOROOOS	2 AILEY U/CRAVENS AVE	2002					4	1913	06/01/1972	06/01/1972	TOR09-001	U
72 - 706.1	17 TOPOOO	D 3 ALLEY RETUREN FUCPACIA AND POST	200	1 14	. 7	1		4	1912	12/01/1974	12/01/1974	TOR09-004/	٩
790 - 78	15 TOP0000	D 1 1423 DDST AV	2002			1	,	· M	1976	08/18/1986	08/18/1986	TOR09-004/	U
8/ - 517.	OULDAUT OC	D Z 2014 TOPPANCE RI VN YST SARTORI					• •	4	1040	08/10/1984	08/10/1984	TOR09-004(4
70 -1771	17 TORODOG	2 2 ND ALLEY S/TOR RI VD AND U/POST A	2002) ~		- ~		4	1913	09/01/1970	09/01/1970	TOR09-005	٩
82 - 2205	PR TORODO	2 1330 FL PRADO	200		. 4		· 10	•	1949	09/16/1982	09/16/1983	TOR09-008	U
RN - 209	27 TORODO	7 1414 CRAVENS	200	- -			-	4	1954	09/05/1980	03/05/1981	TOR09-009	U
74 - 4864	28 TOPOOO	7 1407 SAPTOPI	2002	. •-	. 7		-	~	0701	08/01/1974	08/01/1974	TOR09-010	U
70 -1270	76 TOPOOO	0 7 1434 MADEFI TNA					•	4	1913	09/01/1970	09/01/1970	TOR09-011	٩
70 - 12707	77 TOP0000	D D D D D D D D D D D D D D D D D D D		1 14		10	• •	4	1012	00/01/10/00	09/01/1970	TOR09-011	۵.
70 - 12705		D D D D D D D D D D D D D D D D D D D		- د -		17	• •	. 4	1012	00/01/1970	00/01/10/00	TOR09-011	. a.
202 - 201		D Z N/C PADALELAN D Z N/C PADALELAN		10		10	• •	• -	1926	07/20/1983	07/20/1983	TOR09-013/	. ບ
100 - 00	10000001 to	V D N/E CURREN ALLET E/U DARIUNI		- -	- •	10	- •	- M	1027	00/10/1080	08 / 22 / 1000	TOP/0-013/	
3404 - 604	40 IUKUUU	J ALLET W/U VAN NESS AL IUKKANCE DL J TATA DI EL ILEILEU VU			- 4	90	- 4	י ג י	1010	00/ 44/ 1707	00/ 66/ 17/0	TOP00-015	<u>ہ</u> د
	00 IURUUU	Z IUK DL E/ LLEWELLIN Z ZZZZ ZZZZ DIWE		- -) ~	J C	، د	* ~	1010	110101010101	10/01/10/1	TOP00-015	. ۵
0802- 1/	10 1080005	1 1855 1UKK BLVD		0.	* t	v r	- •	t •	71 41	10/01/19/1	10/01/19/1	TO010-010	
30c5 - C8	24 TORUUTU	Z 1515 DAIE		- 1		2			YCY1	COAL /01 /10	COVI /UI /IU		
71-1111	19 TORO010	1 1/S CRENSHAW BL ELDORADO ST	202	~~ ·		20	- •	1 5	1924	1/61/10//0	1/61/10//0		. د
90 - 26	16 TOR0010	0 3 2515 EL DORADO ST		~~~~	- •	N C		~ 1 ~	1761	0%1/0c/01	UV21/U2/21	TOR 10-00/0	ي د م
72 -1904	57 TORUUTU	1 2555 SONOMA ST		2	- •	νr	- •	3、	1075	U6/U1/1712	10/01/19/20	10K10-000	<u>م د</u>
12416- 87	38 TORUU10	1 2555 SONOMA			- •	л с		t ~	2021	0//01/19/0	U1/U1/19/0	10610-000	L 0
72-45	JZ TORUUIU	0 3 1512 DAIE AVE	nu2	-	-	v	-	t	1764	10/01/ 171 C	7141/10/00	10410-001	L

D-7

MAIN REPAIR	LEAKS -	MODIFIED LEAK REPAIR ORDER DATABASE										Dinalina	Pipeline
Leak Order Number	ATLAS I SHEET F	EAK PRIO Address or Location	SIZE ()IAMETER (inches)Facilit)	/ In	Cause	Repair Type	Material	Year Installe	Detect d Date	Action Date	Segment	(C=CURRENT P=PREVIOUS)
. 122077		1 4/42 CDENCUAL DIVIN			-	ſ		7	1027	701/10/20	07 /01 /1072	TOP10-011	4
73 -503780	TORO010	3 2371 TORR BLVD.	300	л м 		101	- •	* *	1927	06/01/1973	06/01/1973	TOR10-012C	υ
82 - 21720	TOR0010	3 1104 AMAPOLA	200	2	-	2	.	m	1924	06/09/1982	09/09/1982	TOR10-012D	٩.
73 -214711	TOR0010	3 2325-23-TORR. BL. ALLEY N/O TORR.	200	2	m.	2	\$	41	1922	07/01/1973	07/01/1973	TOR10-012D	٩
82 - 21427	TOR0010	3 1104 AMAPOLA	200	2	- (2	, ,	M ·	1924	06/09/1982	07/09/1982	TOR10-012D	a . i
84 - 31450	TOR0010	Z 811 BEECH AVE	200	2	N •	N	r	4 -	2701	11/16/1984	11/50/1984	TOR10-015	۵. ۵
86 - 60326	TOR0010	S 1020 ACACIA AVE		~ •	- •	2	N •	• •	1261	11/20/1980	1961/07/11		n. (
73 -503935	TORUUTU	Z ZZ65 TURR BLVU		- •	- •	v r		* *	2007	C/61/10/00	C/61/10/00	10K10-019B	<i>ع</i> د
71 -1908/8	TOR0010	1 121/ BEECH AVE			- •	2	- •	* +	1261	1/61/10/60	1/61/10/60	10k10-024A	. (
71 -1908/6	TORUU10	1 1217 BEECH AVE			~ ~	ч с		~ t	1241	13/11/10/20	19/11/19/1	TOP10-024A	.
0/177 - 78		2 1230 CKENSHAW BLVU			- +	9 0		* ~	1021	70//11/20/	COV1 / 11 / 20	TOB10-024A	. . c
. YEEE/7 - CO		Z 1464 BEEUN AVE Z 2/1/ Soudma St		10	• •	10	- •-	4 7	1027	12/01/10/21	12/01/1073	TOR10-025	. 0
, ATREE 0- 87		1 1221 ACAPTA AVE	202	1 C		יע		1	1027	06/01/1078	06/01/1978	TOR10-025	۵.
71 - 100225	TOR0010	3 FIDRADO SF & ALIFY F/D ACACIA	2002	. .		1	• ••••	. 4	1927	10/01/1971	10/01/1971	TOR10-026	۰.
R4 - 51893	TORD010	3 1315 - R MADRID	200	10	- 4	10	· M	· ~	1927	09/20/1984	09/20/1984	TOR10-026	. a.
70 -188101	TOP0010	1 1310 ARI INGTON	200	۱ <i>۲</i>		ı ۸			1922	10/01/1970	10/01/1970	TOR10-036	۵.
92 - 82210	TOR0010	3 1422 FUGRACIA AV	300	. •			-	4	1900	08/12/1992	08/12/1992	TOR10-039A	. u
78 -976080	TOR0010	2 2166 TORR BLVD	200	- ~	-	2	-	4	1922	10/01/1978	10/01/1978	TOR10-040D	۵.
73 -503939	TORO010	2 1213 COTA AVE	200	- 2	-	2	-	4	1921	06/01/1973	06/01/1973	TOR10-040D	. ⊶
74 -686329	TOR0010	3 2208 TORRANCE BLVD	200	2	ſ	2	-	4	1922	07/01/1974	07/01/1974	TOR10-040E	۵
84 - 7451	TOR0010	1 ON TORRANCE BLVD 27' W/W AMAPOLA	200	2	-	2	-	4	1921	09/11/1984	09/11/1984	TOR10-102	۵.
84 - 7453	TOR0010	1 ON TORRANCE BLVD 64 W/W AMAPOLA	200	2	4	2	-	4	1921	09/11/1984	09/11/1984	TOR10-102	۵.
27 - 844466	TOR0015	1 1726 DATE AVE	200		*		•	4	1927	7791/1977	08/01/1977	TOR15-003A	. U
72 -208364	TOR0015	1 2303 JEFFERSON	400	4	-	2	ę	4	1927	05/01/1972	05/01/1972	TOR15-008A	U
89 - 40348	TOR0015	3 2303 JEFFERSON	400	4	-	2	m	4	1927	08/15/1989	08/15/1990	TOR15-008A	U
79 -105354	TOR0015	1 2414 SONOMA ST	200	5	-	2	Ļ	4	1972	08/01/1979	08/01/1979	TOR15-009	۵.
86 - 37998	TOR0015	3 1631 AMAPOLA AVE	300	м Ч	-	2	-	4	1921	02/10/1986	02/10/1986	TOR15-012B	ပ
71 -179615	TOR0015	3 1614 COTA AVE	300	м Г	9	2	-	4	1912	05/01/1971	05/01/1971	TOR15-012C	U
02 - 80532	TOR0015	2 CARSON ST & COTA AV	200	2	-	2	•	2	1913	03/18/1992	03/18/1992	TOR15-015	J
78 - 909798	TOR0016	1 1452 EL PRADO	200	2	*	2	m	4	1912	06/01/1978	06/01/1978	TOR16-001	۵.
78 -935946	TOR0016	3 N/E CORNER/CARSON & ALLEY E CABRI	600	6 1		2	9	۴.	1913	08/01/1978	08/01/1978	TOR16-010	م
72 -190415	TOR0016	1 1871 CARSON ST	600	6 1	-	2	4	4	1941	02/01/1972	02/01/1972	TOR16-011	ပ
71 -180366	TOR0016	3 BORDER AV AND CARSON STREET	400	4 1	-	2	*	4	1918	: 03/01/1971	03/01/1971	TOR16-013	٩
72 -459961	TOR0016	2 1645 ARLINGTON	400	4	-	2	-	4	1922	10/01/1972	10/01/1972	TOR16-014	۵.
71 -180357	TOR0016	3 2118 CARSON STREET	400	4	4	2	4	4	1912	03/01/1971	03/01/1971	TOR16-014	۵.
71 -180358	TOR0016	3 1645 ARLINGTON AVENUE	400	4	•	2	-	4	1912	03/01/1971	03/01/1971	TOR16-014	۵.
85 - 35245	TOR0016	3 1746 ABALONE AVE	400	4	•	2	-	m	1913	02/06/1985	02/06/1985	TOR16-015	۵.
73 -684156	TOR0016	3 1740 ABALONE AV	400	4 1	-	2	-	4	1913	12/01/1973	12/01/1973	TOR16-015	۵.
80 - 1867	TOR0016	3 1740 ABALONE AVE	400	4	*	2	•	м	1913	06/18/1980	07/18/1980	TOR16-015	٩
71 -188370	TOR0016	3 1720 CABRILLO	300	3	•	2	-	4	1919	01/01/1971	01/01/1971	TOR16-016	U
71 -190956	TOR0016	2 1718 ANDRED	300	м Г	-	2	-	4	1912	1791/1971	09/01/1971	TOR16-016	U
74 -684175	TOR0016	3 2012 CABRILLO	200	2 1	-	2	-	4	1913	05/01/1974	05/01/1974	TOR16-016C	۵.
74 -684174	TOR0016	3 2025 BORDER AV	200	2	-	2	-	4	1913	05/01/1974	05/01/1974	TOR16-016C	۵.
73 -502783	TOR0016	3 2104 CABRELO AVE	200		*		•	4	1913	05/01/1973	05/01/1973	TOR16-016D	പ
71 - 180377	TOR0016	3 2100 BIK/BORDER AVENUE	400				ę	4	1918	04/01/1971	04/01/1971	TOR16-016D	J
71 - 180371	TOR0016	3 2117 BORDER AVENUE	400	4	-		• • • •	4	1918	04/01/1971	04/01/1971	TOR16-016D	U
90 - 11358	TOR0016	1 1817 BORDER AV	200	~~~	-		-	4	1913	12/19/1990	12/19/1990	TOR16-017	۵.
71 -188206	TOR0016	2 1876 BORDER	200	5	-	2	-	4	1923	02/01/1971	02/01/1971	TOR16-017	۵.
91 - 10111	TOR0016	3 ALLEY EAST OF CABRILLO	200	2	-	2	м	4	1913	01/08/1991	01/08/1991	TOR16-017	٩

MAIN REPAL	LEAKS	- MODIFIED LEAK REPAIR ORDER DATABASE										Dinalina	Pipeline info
Leak Order Number	ATLAS SHEET	LEAK PRIO Address or Location	sıze (JIAMETER (inches)Facility	y In	Cause	Repair Type	Material	Year Installec	Detect d Date	Action Date	Segment	C=CURRENT >=PREVIOUS)
00 - 10002	TOP0016	3 1007 RODAED AV	000		7		-	7	1913	12/19/1990	12/19/1991	TOR16-017	
90 - 10091	TORD016	3 1720 CABRILLO AV	200	1 CI	• •	1		.4	1957	12/19/1990	01/19/1991	TOR16-017A	. م
74 -686192	TOR0016	1 1720 MARTINA	300	ς Γ	•	~	-	4	1913	04/01/1974	04/01/1974	TOR16-018	٩
72 -460196	TOR0017	2 ALLEY S/CARSON ST	300	3	•	2	-	4	1913	06/01/1972	06/01/1972	TOR16-018	٩
73 -513287	TOR0016	1 1743 CABRILLO	200	2	*	2	-	4	1912	03/01/1973	03/01/1973	TOR16-021	٩
71 -192899	TOR0016	2 1917 19 4 218TH STREET	200	2	-	2	-	4	1912	04/01/1971	04/01/1971	TOR16-021	٩
71 -180361	TOR0016	2 1753 CABRILLO AVENUE	200	2	4	2	-	4	1912	05/01/1971	05/01/1971	TOR16-021	۵.
72 -460194	TOR0016	2 1723 MARTINA AVE	200	2	-	2	-	4	1913	06/01/1972	06/01/1972	TOR16-028	۵.
74 - 704011	TOR0017	3 2067 LINCOLN AVE	200	2	-	2	-	4	1923	08/01/1974	08/01/1974	TOR17-005	ፈ
71 -190782	TOR0017	2 2415 ARLINGTON	200	2	-	2	•	4	1926	11/01/1971	11/01/1971	TOR17-005	ፈ
74 -643175	TOR0017	3 2503 ARLINGTON AVE	200	2	-	2	۰-	4	1923	07/01/1974	07/01/1974	TOR17-005A	٩
76 -853719	TOR0017	3 ARLINGTON & PLAZA DEL AMO	400	4	4	2	9	4	1922	09/01/1976	09/01/1976	TOR17-006A	ပ
73 -570307	TOR0017	3 EAST SIDE OF ARLINGTON&PLAZADELAM	400	4	-	2	-	4	1922	08/01/1973	08/01/1973	TOR17-006A	U
85 - 1066	TOR0017	3 2732 ANDRED	400	4	-	~	9	4	1923	05/15/1985	04/15/1986	TOR17-009	U
85 - 1064	TOR0017	3 2407 GRAMERCY AVE	100	۰-		2	m		1943	05/15/1985	01/15/1986	TOR17-011	U
78 -992421	TOR0017	3 1912 PLAZA DEL AMO	200	2	*	2	-	4	1916	11/01/1978	11/01/1978	TOR17-014A	U
74 -704013	TOR0017	3 1912 PLAZA DEL AMO	200	2	-	2	-	4	1965	10/01/1974	10/01/1974	TOR17-015	ပ
91 - 7617	TOR0017	1 2413 CABRILLO ALLEY BEHIND	200	-	-	2	-	•	1941	05/14/1991	05/14/1991	TOR17-018	ပ
79 - 150267	TOR0017	3 2213 BORDER	200	2	-	2	-	м	1913	08/01/1979	08/01/1979	TOR17-027	۵.
81 - 18650	TOR0017	3 N/U CORN O/PLA7A DF AMO/UFSTERN	400	4		2	2	2	1949	06/17/1981	06/17/1981	TOR17-031C	۵.
78 -01/0/5	T00017	1 22501 UECTEDN AVE	007		7	1	7	10	1040	07/01/1978	07/01/1978	TOR17-031C	. a.
72 - 20/150	2100001	7 1400 334TU CT		- +	• •	10	•	10	7.01	12/01/10/21	12/01/1073	TOP17-0368	. د
80 - 151/		J 1055 TODANCE BI		 		7 4	~ **	1 -1	1000	05/05/1080	05/05/1080	TOR09-0148	<u>ں</u> د
10 - 10 - 10 - 10 - 10 - 10 - 10 - 10 -	100001	1 3800 DI A7A DEL AUO				• ~	- ư	t ~	1088	07/20/1088	07/20/1088	TOP15-016C	ه د
1/3CF - 00		1 2600 FLAZA VEL ANU 1 2672 di 474 dei 440		- -		t ~	אר	2 4	1087	00/16/1087	00/16/1087	TOP15-0160	י נ
100022 - 10		1 DIATA DEL AND 8 CADON		- -		t ~	א ר	2 4	1081	10/17/01/01	10/16/1085	TOP15-0160	ە د
177155 - 68 77155 - 88		I FEALA VEL ANU & CARSON 1 3800 di A7A dei Anu		1 C		t ~	v 1	2.4	1085	02/04/1088	02/04/1088	TOP15-0160	ے د
44100 - DO	7100001	1 2800 DI A7A DEI ANO	200	1 C		• •	> <		1080	06/13/1080	06/13/1080	TOP15-016C	، د
2101 - 40 22 05 1000		T DOCT AVE 9 ALEVAND	2004	- -	- 14	r a	4 C	0~	1012	7201/10/10	7201/10/10		00
00000- 11	TOP0010	J PUSI AVE & ALLET W/U SAKIUKI 1 1/0 FOEMOUALI & COMPUA		- •	، ر	cα	י ר י	t ~	1082	00/21/1086	00/21/108/	TOP10-0080	. د
					- 14	0 0	1 1	0~		08/01/1072	08/01/1072		ە ر
12 -40U200		J 14 14 CKAVENS			<u> </u>	> c	0 4	- t	7001	02/01/17/07/	00/01/19/0	TOP10-005	ם נ
		1 1/3 ELUOKAUU CKENSHAW			4 6	~ C	0 ~	<i>t</i> ~	1201	07/01/10/10	07/01/10/10	TOP10-011	
C04C1 - 14		1 1/S ELUDKADU SI UKENSHAW			0 -	> c	0 +	- t	1724	12/01/19/1	12/01/19/1	TOR10-0111	∟ د
12/AC - 40		1 212 PUKIULA			- 1	~ 0		t ~	10.01	12/01/1204	1201/10/21	TOR10-020A	ם נ
110041 - 11		1 IZIN BEECH AVE		- -	<u>-</u>	۰ (- c	t ~	1761	02/11/10/20	1111110/00	T0000-010	. د
		1 1304 EL PRAUO		- •	± ~	2	+ -	- t	A+4-	03/11/10/60	02/11/10/20	TO10-010	ינ
C2000 - 48		2 22/1 IUKKANCE BLVU		 1 0	* (2;	1 1	• t	1721	00/1//1/00	10011/11/00	1010-01V01	. נ
90 - 7035	TORU010	1 1222 CRENSHAW BLVD	200	י ה י	~	2;	Υ.	0.	1985	0661/60/20	0661/60/20	10K10-024A	، د
75 -686098	TORUUIU	Z 1504 ENGRACIA AVE	300	. ر. 		2:	4.	• t	1221	C/41/10/90	C141/10/90		a, (
71 -179616	TOR0015	1 1623 COTA AVE	200	2	4	10	9	4	1913	02/01/1971	02/01/1971	TOR15-015	u
87 - 11864	TOR0016	3 2061 - 220TH ST	200	2	2	10	-	-	1958	05/20/1987	05/20/1987	TOR16-026A	പ
87 - 11125	TOR0016	3 1717 MARTINA AVE	200	2	4	10	9	-	1974	02/10/1987	02/10/1987	TOR16-028	۵.
83 - 22622	TOR0017	3 2409 ARLINGTON AVE	400	4	4	10	м	9	1922	05/17/1983	02/17/1984	TOR17-006A	U
76 -844861	TOR0009	1 1335 POST	200	2	9	1	ŝ	4	1913	06/01/1976	06/01/1976	TOR09-003	₽.
90 - 2469	TOR0009	3 1318 S CRAVENS	200	2		=	9	м	1936	08/03/1990	08/03/1990	TOR09-003	٩.
79 - 51171	TOR0009	3 1326 1/2 ENGRACIA AVE	300	۲ ۲	4	1	9	m	1976	01/01/1979	01/01/1979	TOR09-004A	ပ
81 - 11917	TOR0009	3 1331 POST AVE	300	۲ ۲	4	11	-	4	1912	09/28/1981	12/28/1981	TOR09-004C	٩
81 - 11915	TOR0009	3 1330 EL PRADO	200	2	m	:	9	5	1949	09/28/1981	10/28/1981	TOR09-008	υ
80 - 2098	TOR0009	3 1330 EL PRADO	200	2	m	1	9	S	1949	09/05/1980	03/05/1981	TOR09-008	U
89 - 40450	TOR0009	3 ALLEY W/O VAN NESS AT ENGRACIA AV	200	2	м	1	6	•	1927	08/22/1989	08/22/1990	TOR09-013A	U

MAIN REPAIF	۲ LEAKS	MODIFIED LEAK REPAIR ORDER DATABASE										Dinolino	Pipeline
Leak Order Number	ATLAS L SHEET P	.EAK RIO Address or Location S	SIZE)IAMETER (inches)Facilit	LEAK	Cause	Repair Type	Material	Year Installe	Detect J Date	Action Date	Segment	C=CURRENT
80 - 9579	TOR0009	1 1920 TORRANCE	200	2	. 4	1	9	4	1936	11/07/1980	11/07/1980	TOR09-013B	
84 - 56812 84 - 3827/	TOR0009	3 900 VAN NESS 2 1010 VAN NESS	600	۰ م ر	4 *		\$ \$	~ ~	1912	03/12/1984	01/12/1985	TOR09-014A	ن ن
91 - 2929	TOR0009	3 923 VAN NESS AV	800	0 v0	- 4		o~0	t- 1	1912	08/15/1991	11/15/1991	TOR09-014A	ງ ເມ
77 -856088	TOR0009	3 MULLIN & CABRILLO	600	6	N	11	-	4	1912	03/01/1977	03/01/1977	TOR09-014A	υ
72 -412553	T OR 0009	3 W/S MULLIN AVE	600	۰ م		11	\$ V	4	1953	02/01/1972	02/01/1972	TOR09-014A	U
86 - 38275	TOR0009	2 1010 VAN NESS	600	9		=	-	4	1913	03/20/1986	03/20/1986	TOR09-014A	U
71 -190349	TOR0009	3 VAN NESS AND TORR BL	800	80 ·	4		۰ ور	4	1912	04/01/1971	04/01/1971	TOR09-014A	U
76 -856051	TOR0009	3 MULLIN & TORR BLVD	600	0 ·		=:	۰ و.	4	1971	06/01/1976	06/01/1976	TOR09-0148	U
82 - 22030	TOR0009	3 1200 CABRILLO	600	~ •	- 1	=:	••	4.	1913	09/16/1982	05/16/1983	TOR09-0140	U I
92 - 82159	TORU009	1 1250 S CABRILLO AV		Q 7	- 1	=:	- 1	4 (1900	U8/U6/1992	U8/U6/1992	TORU9-014C	ن ں
702021 - 71		2 1320 LABKILLU 7 VAN NESS 8 MILLIN AVE		- •	E		0 4	v ~	1961	11/01/19/9	11/01/19/9		، د
7011 - 10 7010 - 10		2 VAN NESS & MULLIN AVE 2 todaaret 0 hunded et					0 4	t u	1001	1041/01/1000	1061/01/60	10KUY-014E	י ר
00 - 212/ 872 - 8876	10K0009	J UKKANGE & MUBER SI 7 J17TU & FADDILLA		0 M	אר 	= =	2 4	.	1051	08/10/10/11	02/10/1901	10KUY-010 TOBD0-072	י נ
85 - 35500		3 1770 213 ST		רייי ריייר		= =	~	- ^	1051	03/08/1085	03/08/1085	TOP00-023	ى د
84 - 56814	TOR0009	3 1739 213TH ST	300			: =	9.9	1	1951	03/12/1984	01/12/1985	TOR09-023	
81 - 11918	TOR0009	3 I/S 213ST & CABRILLO AVE	300	. M		:=	9		1951	09/28/1981	12/28/1981	TOR09-023	0
87 - 12392	TOR0009	3 9' S/N TORRANCE BLD & 272' W/W W	800	8		1	9	4	1945	08/11/1987	08/11/1987	TOR09-030A	U U
88 - 61465	TOR0009	3 1619 GRAMERCY AVE	200	2	-	1		4	1936	08/18/1988	09/18/1988	TOR09-032	ر د
70 -127075	TOR0009	2 1628 CRAVENS	200	2	~	=	•	4	1913	02/01/1970	09/01/1970	TOR09-032	ں ۲
86 - 37927	TOR0010	3 1313 DATE AVE	200	2	~	1	~	4	1935	01/31/1986	01/31/1986	TOR10-0058	ں
71 -111118	TOR0010	1 1/S ELDORADO CRENSHAW	300	м Г	<u>~</u>	÷	-	4	1924	07/01/1971	07/01/1971	TOR10-006	۵.
73 -643335	TOR0010	3 I/S ALLEY S/TORR BL & E/CRENSHAU	200	2	<u> </u>	=	9	4	1927	12/01/1973	12/01/1973	TOR10-007	ပ
84 - 7091	TOR0010	3 2515 ELDORADO	200	2		=	-	4	1926	11/19/1984	12/19/1984	TOR10-007A	പ
78 -992401	TOR0010	1 ALLEY N/TORR E/CRENSHAW	200	2		=	9	4	1928	08/01/1978	08/01/1978	TOR10-013	م
86 - 60290	TOR0010	3 1103 BEECH AVE	200	2	· [~	=	9	4	1927	12/03/1986	05/03/1987	TOR10-013	۵.
90 - 10021	TOR0010	1 812 BEECH AVE	200	2	-	=	9	2	1939	10/30/1990	10/30/1990	TOR10-014A	ပ
82 - 28681	TOR0010	1 2310 SIERRA ST	200	2	<u>v</u>	=	-	4	1917	05/28/1982	05/28/1982	TOR10-016A	ပ
90 - 2619	T0R0010	3 2309 SIERRA AV	200	2	_	11	9	4	1917	10/30/1990	06/30/1991	TOR10-018A	υ
89 - 1798	TOR0010	3 1011 PORTOLA	200	2	-	:	~ '	4	1917	07/17/1989	07/17/1989	TOR10-020A	ပ
84 - 39690	TOR0010	1 1103 PORTOLA AVE	200	2	_	5	-	4	1917	11/30/1984	11/30/1984	TOR10-020A	ы С
78 -933840	TOR0010	2 1217 ACACIA AVE	200	~	· - •	=	9	4	1917	06/01/1978	06/01/1978	TOR10-025	۹.
84 - 51914	TOR0010	3 1313 MADRID AVE	200	2	ч ·	::	Ŷ	4	1921	09/25/1984	09/25/1984	TOR10-026	٩.
82 - 22000	TORUUTU	Z 1432 ENGRACIA AVE	200	~ ı	- T	Ξ;	•	N ·	21.61	06/11/1982	1982/1982	TOR 10-059A	U
84 - 7093	TOR0010	3 1310 MANUEL AVE		י הי י	4.	=:	\$	• •	2161	11/21/1984	12/21/1984	TOR10-039A	с (
82 - 21999	1 OK 00 10	3 W/SIDE ALLEY S/O ENGRACIA AVE	202	י רי י	-1	= :	۰ o	4	21.61	06/11/1982	06/11/1985	TOR 10-059A	U
78 -933843	TOR0010	Z 1432 ENGRACIA AVE	200	m N		F	-	4	1912	07/01/1978	07/01/1978	TOR10-039A	പ
72 -460469	TOR0010	3 CRENSHAU BLVD. & ALLEY S/O TORRAN	300	м М		=	9	4	1945	09/01/1972	09/01/1972	TOR'10-040A	ပ
83 - 50595	TOR0010	2 2260 TORRANCE BLVD	200	2	<u> </u>	=	-	4	1922	07/15/1983	07/15/1983	TOR10-040D	٩.
86 - 38081	TOR0015	3 CRENSHAW BLVD/CARSON ST	800	8	<u> </u>	:	6	2	1924	02/24/1986	02/24/1986	TOR15-004B	۵.
86 - 37946	TOR0015	3 I/S CRENSHAW BL & JEFFERSON ST	400	4	m 	:-	9	4	1945	02/03/1986	04/03/1986	TOR15-008	U
81 - 18786	TOR0015	3 2341 JEFFERSON	400	4	.	11	-	4	1923	07/22/1981	07/22/1981	TOR15-008A	പ
77 -800962	TOR0015	3 29 W/E CRENSHAW BL	400	4	~	=	9	4	1927	01/01/1977	01/01/1977	TOR15-008A	ပ
86 - 37869	TOR0015	3 N/W CORNER ALLEY N/CARSON &W/COTA	300	м Г	m	=	9	4	1924	01/22/1986	01/22/1986	TOR15-012B	υ
77 -908347	TOR0015	1 2265 CARSON APT B	200	2	<u>-</u>	=	9	4	1924	06/01/1977	06/01/1977	TOR15-015	ပ
79 - 51151	TOR0015	3 1736 WATSON AVE	200	2	-		6	2	1946	01/01/1979	01/01/1979	TOR15-017B	U
88 - 20400	TOR0015	3 1729 MANUEL AVE	200	2	4	.1	9	2	1950	08/30/1988	05/30/1989	TOR15-018A	ں
85 - 39847	TOR0016	2 1532 EL PRADO AVE	200	2	4	11	2	4	1912	01/02/1985	01/16/1985	TOR16-001	۵.
83 - 50533	TOR0016	3 1613 CRAVENS	300	n L	~	=	-	4	1912	06/24/1983	06/24/1983	TOR16-004	ပ

ipeline o scurrent		υ L	ບບ	۵.	۵. ۱	ں <i>د</i>	۲ с	- 0	. ഫ	υ	υ	۵. ۱	a. 6	ט נ	ں ن	ы	ы	٩	U I	0	י נ	י פ	. ሲ	ပ	υ I	م. د	ى ر	<u>ہ</u> م	ы С	۹.	. c	. ۵	. U	പ	٩	٩	۵. ۵	<i>ع</i> د	ה נ	م	ል. ነ	م م	. a .
P ipeline inf Segment (C= Number P=P		R16-004	R16-007	R16-010	R16-010	R16-016	1410-010 014-017	B16-021	R17-005	R17-006A	JR17-006A	117-0068	0817-006B	017-0008	R17-008B	R17-008B	R17-008B	R17-010	R17-012A	0817-012A	020-7130	R17-020	R17-027	R17-028	R17-031	0817-051C	0217-036A	R09-005	R09-013A	0R09-014	0KU9-U14	BU9-014	R10-005B	R10-008B	R10-013	R10-013	R10-014		R10-025	R10-027	R10-035A	R10-102	R10-102
Action S Date		1/27/1991 T0	3/17/1989 10	9/01/1978 10	8/01/1977 TO	7/01/1976 TO	14/01/19/1 10	DI 1661/00/01	0/01/1979 TO	9/01/1978 TO	1/01/1976 TO	18/01/1977 TC	01 4/61/10/6	12/11/1900 10	14/15/1990 TC	14/01/1979 TO	12/17/1984 TC	11/04/1984 TC	0/14/1991 TC	12/11/1988 TC	UI 1001/21/21/21	01 0041/11/61	12/15/1984 TO	14/01/1979 TC	02/01/1979 TC	01/19// 10/00	2/07/1991 TC	19/16/1983 TO	11/12/1985 TC	12/19/1982 TO	01 1861/c0/6	0/05/1081 TO	04/05/1985 TO	4/17/1985 T0	8/20/1985 TO	1/29/1984 TO	7/01/1978 TO	7/26/198/ 10	2/05/1985 TD	2/05/1985 TO	2/05/1985 10	9/11/1985 TO	9/11/1985 10
Detect Date	uate	12/27/1990 0	03/17/1989 0	09/01/1978 0	08/01/1977 0	07/01/1976 0	04/01/10/10	0 1441/C0/C0	10/01/1979 1	09/01/1978 0	11/01/1976 1	08/01/1977 0	0 4/61/10/60	7 701/11/CD	05/15/1989 0	04/01/1979 0	05/17/1983 0	01/04/1984 0	05/14/1991 1	05/11/1987 0	05/11/10/07	06/09/1982 0	06/15/1983 0	04/01/1979 0	02/01/1979 0	06/01/19// C	05/07/1991 1	09/16/1982 0	03/12/1984 0	03/19/1981 0	09/10/20/20	00/05/1980 0	12/05/1984 0	01/17/1985 0	11/20/1984 0	11/15/1984 1	07/01/1978 0	11/26/1986 1	12/05/1984 1	12/05/1984 1	12/05/1984 1	09/11/1984 0	09/11/1984 0
Year		1912	1974	1913	1913	1912	51.61		1023	1922	1922	1926	1956	10201	1930	1930	1930	1947	1941	1941	CI VI	1013	1913	1965	1923	1949	1040	1937	1927	1913	2191	1012	1935	1924	1928	1928	1929	192/	1027	1927	1921	1921	1925
aterial Ir		40	1 - 1	• • •	ۍ ب	4.	4 -	t v	יער	t	4	4	4.	4 ~	• •	14	2	4	2	 、	t +	- 7	2	5	ю (ς η τ	<i>ч</i> ८	1 CJ	м	υı	л I	ייר	n Irr	ŝ	2	4	4.	4 •	r 4	r • 7	4	~ r	101
tepair Tvne v	I ype	، ۵	- •-	· •0	9	• •	- •	- ~	. .	0.0	ŝ	9	، – ،	0 4	~	o v	9	9	-	 (2 4	0 +		9	\$	~ o	0 ~	2 14	5	~ `	2 10	<i>1</i> 0	1 U	1	2	2	2	~ ~	20	5	2	~ r	1 2
	cause	::		Ę	5	::		= =		:5	:	=	=:	=:			:	1	-	=:	= ;		=	:=			==	12	12	21	22	4¢	1 7	15	12	12	21	22	20	12	12	24 24 24	15
LEAK		، ۵۰		• •	m	9	N •	- 、	* M	4	м	м	• P	~ ~	א ר	n M	м	м	9	4 •	0~	4 N	0 I	m	m	M 1	0 M	0 V I	\$	Ś	ò	0 <	o ~c	o	9	-	،	vo v	0 <	o	9	~	00
ni i i tv	10111TY	÷- •		•	-	e e	- •		- •-		-		- •		- *-	• •	-	-	-		~ ~		• •		-						• •				~	-	• •	• •	-	• •	-	- •	
01AMETER Stochocycs	(1ncnes)ra	Mr	<u> </u>	ס~ו	9	m i	2	ч r	10	1-4	ю	4	41	~) M	א ר ייי ר	n M	ю	2	2	~ ~	л r	20	1 ~I	I •	4	4	<i>~</i> ~	1 01	N N	• و	Ŷ	0 4	2 ~	2	0	2	2	N 1	2 0	5	2	20	5
217E	S12E (300		600	600	300	200			400	300	400	400	2002		300	300	200	200	200			200	100	400	400		200	200	600	600			200	200	200	200	200		200	200	200	200
LEAK REPAIR ORDER DATABASE	Address or Location	IS CRAVENS AV	28 MAKUELINA AV 10 seamepry	52 CABRILLO	EY E/O CABRILLO N/O CARSON	18 ANDREO	15 BORDER AVENUE	JS CABKILLU AVE	10 ADI TNGTON	32 ARLINGTON AVE	S PLAZA DEL AMO & ARLINGTON	20 ARLINGTON	D2 ARLINGTON AVE	J SANTA FE & ANDREO	J COD ANDER AVE & SANTA FF AV	S SANTA FE AVE & ANDREO AVE	VIA FE & ANDRED	S GRAMMERCY & SEPULVEDA	LEY W/O ANDREO & S/O PLAZA DEL	43 ANDREO AVE	U S/S PL OF 222ND ST	SU LINCULN AVE	23 BORDER	BRILLO AVE 140' W/W BORDER	501 WESTERN AVE	501 S. WESTERN AVE	DIM & WESIEKN VE M/E FOD 224IU & WESTEDN	DP CRAVENS AV	20 SARTONI ALLEY R/O	26 CABRILLO AVE	DZ CABRILLO	E CURNEK UP DOUBLE SI IN THE	CU CABRILLU 26 nate ave	55 SONOMA AVC/DATE ST	5 BEECH ST	7 BEECH AVE	16 BEECH AVE	EY E/OF ACACIA	/ PORTOLA AVE	25 SONOMA	J7 PORTOLA	TORRANCE BLVD 2 V/W AMAPOLA	IUKKANLE BLVU 24 W/W AMARULA LEY W/AMAPOLA 15 S/MAIN TIE IN
- MODIF) LEAK	PRIO	5 9	2 2 2 2 2 2 7 7	2 - 2	3 ALI	2 17	02 G M	2	27.1	572 n M	2 1/	1 22	32	N N N		2 n m	3 SAI	м 1/	3 ALI	77 N N	5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	3 K 1 M	N N	3 22	- 1	22		₽ m	314	5 N 1	2 2	ч с м с		2 M	2 91	2 10	3 ALI	2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	2 <u>7</u> 2 2 2 2 3	м 1	NON	a cu ALI
<pre>LEAKS ATLAS Cueft</pre>	SHEET	TOR0016	TORUUIC	TOR0016	TOR0016	TOR0016	TOR0016			TOR0017	TOR0017	TOR0017	TOR0017	TOR0017	TOPD012	TORDO17	TOR0017	TOR0017	TOR0017	TOR0017	TOR001/	10KUU1/	TOR0017	TOR0017	TOR0017	TOR0017	TORUU1	TOROD00	TOR0005	TOR:0005	TOR0005			TOROO10	TORODIO	TOR0010	TOR0010	TOR0010	TOR001C	TOR0010	TOR0010	TOR0010	TOR0010
MAIN REPAIF Leak Order	Number	90 - 10099	90 - 2299 80 - 1350	78 - 990285	77 -876991	76 -844795	71 -180363	79101 - 1016/	70 - 105706	78 - 51056	76 -685841	77 -908377	74 - 704012	87 - 80358 7/ 70/015	R0 - 204015	70 - 72656	83 - 22626	84 - 30613	91 - 2890	87 - 80362	86 - 3589	8/ - 80361	83 - 22604	79 - 72660	79 -990314	77 -908172	90 - 2359 01 - 2887	82 - 22029	84 - 56811	81 - 11114	80 - 2096	<pre><!--!!! - 18</pre--></pre>	7877 - 78	85 - 35069	84 - 39630	84 - 39601	78 - 25333	86 - 60327	84 - 7388 84 - 7388	84 - 7393	84 - 7394	84 - 7449	84 - 7455 84 - 7455

MAIN REPAIR LEAKS	- MODIFIED	LEAK REPAIR ORDER DATABASE										Dinolino :	Pipeline
Leak Order ATLAS	LEAK			DIAMETER	LEAK		tepair		Year	Detect	Action	Segment (C=CURRENT
Number SHEET	PRIO	Address or Location	SIZE	(inches)Facility	IJ	Cause	Type Ma	aterial Ir	stalled	Date	Date	Number P	=PREVIOUS)
83 - 22724 TOR0017	3 2740 6	SRAMERCY AVE	200	2 1	9	12	2	ъ	1947 0	7/06/1983 (01/06/1984	TOR17-010	4
83 - 22728 TOR0017	3 2756 6	SRAMERCY AVE	200	2	9	12	-	'n	1947 0	7/06/1983 (01/06/1984	TOR17-010	م
83 - 22727 TOR0017	3 IN GRA	AMERCY 150' N/NPL SEPULVEDA	200	2	9	12	2	'n	1947 0	7/06/1983 (01/06/1984	TOR17-010	٩
83 - 22725 TOR0017	3 2741 0	SRAMERCY AVE	200	2	9	12	2	ы	1947 0	7/06/1983 (01/06/1984	TOR17-010	٩
83 - 22726 TOR0017	3 2748 6	SRAMERCY AVE	200	2	9	12	2	ъ	1947 0	7/06/1983 (01/06/1984	TOR17-010	٩
83 - 22729 TOR0017	3 IN GRA	AMERCY 90'N/NPL 0/SEPULVEDA	200	2	9	12	2	ъ	1947 0	7/06/1983 (01/06/1984	TOR17-010	٩
83 - 22723 TOR0017	3 2732 0	SRAMERCY AVE	200	2	9	12	2	Ś	1947 0	7/06/1983 (01/06/1984	TOR17-010	٩
88 - 20210 TOR0017	3 2213 E	30RDEN AVE	200	2	9	12	2	4	1913 0	5/18/1988 (07/18/1988	TOR17-027	۵.
81 - 11347 TOR0017	3 22501	WESTEREN AVE TORRANCE	200	2	9	12	2	ъ	1949 0	6/10/1981 (7/10/1981	TOR17-031A	U

NATIONAL CENTER FOR EARTHQUAKE ENGINEERING RESEARCH LIST OF TECHNICAL REPORTS

The National Center for Earthquake Engineering Research (NCEER) publishes technical reports on a variety of subjects related to earthquake engineering written by authors funded through NCEER. These reports are available from both NCEER's Publications Department and the National Technical Information Service (NTIS). Requests for reports should be directed to the Publications Department, National Center for Earthquake Engineering Research, State University of New York at Buffalo, Red Jacket Quadrangle, Buffalo, New York 14261. Reports can also be requested through NTIS, 5285 Port Royal Road, Springfield, Virginia 22161. NTIS accession numbers are shown in parenthesis, if available.

- NCEER-87-0001 "First-Year Program in Research, Education and Technology Transfer," 3/5/87, (PB88-134275).
- NCEER-87-0002 "Experimental Evaluation of Instantaneous Optimal Algorithms for Structural Control," by R.C. Lin, T.T. Soong and A.M. Reinhorn, 4/20/87, (PB88-134341).
- NCEER-87-0003 "Experimentation Using the Earthquake Simulation Facilities at University at Buffalo," by A.M. Reinhorn and R.L. Ketter, to be published.
- NCEER-87-0004 "The System Characteristics and Performance of a Shaking Table," by J.S. Hwang, K.C. Chang and G.C. Lee, 6/1/87, (PB88-134259). This report is available only through NTIS (see address given above).
- NCEER-87-0005 "A Finite Element Formulation for Nonlinear Viscoplastic Material Using a Q Model," by O. Gyebi and G. Dasgupta, 11/2/87, (PB88-213764).
- NCEER-87-0006 "Symbolic Manipulation Program (SMP) Algebraic Codes for Two and Three Dimensional Finite Element Formulations," by X. Lee and G. Dasgupta, 11/9/87, (PB88-218522).
- NCEER-87-0007 "Instantaneous Optimal Control Laws for Tall Buildings Under Seismic Excitations," by J.N. Yang, A. Akbarpour and P. Ghaemmaghami, 6/10/87, (PB88-134333). This report is only available through NTIS (see address given above).
- NCEER-87-0008 "IDARC: Inelastic Damage Analysis of Reinforced Concrete Frame Shear-Wall Structures," by Y.J. Park, A.M. Reinhorn and S.K. Kunnath, 7/20/87, (PB88-134325).
- NCEER-87-0009 "Liquefaction Potential for New York State: A Preliminary Report on Sites in Manhattan and Buffalo," by M. Budhu, V. Vijayakumar, R.F. Giese and L. Baumgras, 8/31/87, (PB88-163704). This report is available only through NTIS (see address given above).
- NCEER-87-0010 "Vertical and Torsional Vibration of Foundations in Inhomogeneous Media," by A.S. Veletsos and K.W. Dotson, 6/1/87, (PB88-134291).
- NCEER-87-0011 "Seismic Probabilistic Risk Assessment and Seismic Margins Studies for Nuclear Power Plants," by Howard H.M. Hwang, 6/15/87, (PB88-134267).
- NCEER-87-0012 "Parametric Studies of Frequency Response of Secondary Systems Under Ground-Acceleration Excitations," by Y. Yong and Y.K. Lin, 6/10/87, (PB88-134309).
- NCEER-87-0013 "Frequency Response of Secondary Systems Under Seismic Excitation," by J.A. HoLung, J. Cai and Y.K. Lin, 7/31/87, (PB88-134317).
- NCEER-87-0014 "Modelling Earthquake Ground Motions in Seismically Active Regions Using Parametric Time Series Methods," by G.W. Ellis and A.S. Cakmak, 8/25/87, (PB88-134283).
- NCEER-87-0015 "Detection and Assessment of Seismic Structural Damage," by E. DiPasquale and A.S. Cakmak, 8/25/87, (PB88-163712).

- NCEER-87-0016 "Pipeline Experiment at Parkfield, California," by J. Isenberg and E. Richardson, 9/15/87, (PB88-163720). This report is available only through NTIS (see address given above).
- NCEER-87-0017 "Digital Simulation of Seismic Ground Motion," by M. Shinozuka, G. Deodatis and T. Harada, 8/31/87, (PB88-155197). This report is available only through NTIS (see address given above).
- NCEER-87-0018 "Practical Considerations for Structural Control: System Uncertainty, System Time Delay and Truncation of Small Control Forces," J.N. Yang and A. Akbarpour, 8/10/87, (PB88-163738).
- NCEER-87-0019 "Modal Analysis of Nonclassically Damped Structural Systems Using Canonical Transformation," by J.N. Yang, S. Sarkani and F.X. Long, 9/27/87, (PB88-187851).
- NCEER-87-0020 "A Nonstationary Solution in Random Vibration Theory," by J.R. Red-Horse and P.D. Spanos, 11/3/87, (PB88-163746).
- NCEER-87-0021 "Horizontal Impedances for Radially Inhomogeneous Viscoelastic Soil Layers," by A.S. Veletsos and K.W. Dotson, 10/15/87, (PB88-150859).
- NCEER-87-0022 "Seismic Damage Assessment of Reinforced Concrete Members," by Y.S. Chung, C. Meyer and M. Shinozuka, 10/9/87, (PB88-150867). This report is available only through NTIS (see address given above).
- NCEER-87-0023 "Active Structural Control in Civil Engineering," by T.T. Soong, 11/11/87, (PB88-187778).
- NCEER-87-0024 "Vertical and Torsional Impedances for Radially Inhomogeneous Viscoelastic Soil Layers," by K.W. Dotson and A.S. Veletsos, 12/87, (PB88-187786).
- NCEER-87-0025 "Proceedings from the Symposium on Seismic Hazards, Ground Motions, Soil-Liquefaction and Engineering Practice in Eastern North America," October 20-22, 1987, edited by K.H. Jacob, 12/87, (PB88-188115).
- NCEER-87-0026 "Report on the Whittier-Narrows, California, Earthquake of October 1, 1987," by J. Pantelic and A. Reinhorn, 11/87, (PB88-187752). This report is available only through NTIS (see address given above).
- NCEER-87-0027 "Design of a Modular Program for Transient Nonlinear Analysis of Large 3-D Building Structures," by S. Srivastav and J.F. Abel, 12/30/87, (PB88-187950).
- NCEER-87-0028 "Second-Year Program in Research, Education and Technology Transfer," 3/8/88, (PB88-219480).
- NCEER-88-0001 "Workshop on Seismic Computer Analysis and Design of Buildings With Interactive Graphics," by W. McGuire, J.F. Abel and C.H. Conley, 1/18/88, (PB88-187760).
- NCEER-88-0002 "Optimal Control of Nonlinear Flexible Structures," by J.N. Yang, F.X. Long and D. Wong, 1/22/88, (PB88-213772).
- NCEER-88-0003 "Substructuring Techniques in the Time Domain for Primary-Secondary Structural Systems," by G.D. Manolis and G. Juhn, 2/10/88, (PB88-213780).
- NCEER-88-0004 "Iterative Seismic Analysis of Primary-Secondary Systems," by A. Singhal, L.D. Lutes and P.D. Spanos, 2/23/88, (PB88-213798).
- NCEER-88-0005 "Stochastic Finite Element Expansion for Random Media," by P.D. Spanos and R. Ghanem, 3/14/88, (PB88-213806).
- NCEER-88-0006 "Combining Structural Optimization and Structural Control," by F.Y. Cheng and C.P. Pantelides, 1/10/88, (PB88-213814).

- NCEER-88-0007 "Seismic Performance Assessment of Code-Designed Structures," by H.H-M. Hwang, J-W. Jaw and H-J. Shau, 3/20/88, (PB88-219423).
- NCEER-88-0008 "Reliability Analysis of Code-Designed Structures Under Natural Hazards," by H.H-M. Hwang, H. Ushiba and M. Shinozuka, 2/29/88, (PB88-229471).
- NCEER-88-0009 "Seismic Fragility Analysis of Shear Wall Structures," by J-W Jaw and H.H-M. Hwang, 4/30/88, (PB89-102867).
- NCEER-88-0010 "Base Isolation of a Multi-Story Building Under a Harmonic Ground Motion A Comparison of Performances of Various Systems," by F-G Fan, G. Ahmadi and I.G. Tadjbakhsh, 5/18/88, (PB89-122238).
- NCEER-88-0011 "Seismic Floor Response Spectra for a Combined System by Green's Functions," by F.M. Lavelle, L.A. Bergman and P.D. Spanos, 5/1/88, (PB89-102875).
- NCEER-88-0012 "A New Solution Technique for Randomly Excited Hysteretic Structures," by G.Q. Cai and Y.K. Lin, 5/16/88, (PB89-102883).
- NCEER-88-0013 "A Study of Radiation Damping and Soil-Structure Interaction Effects in the Centrifuge," by K. Weissman, supervised by J.H. Prevost, 5/24/88, (PB89-144703).
- NCEER-88-0014 "Parameter Identification and Implementation of a Kinematic Plasticity Model for Frictional Soils," by J.H. Prevost and D.V. Griffiths, to be published.
- NCEER-88-0015 "Two- and Three- Dimensional Dynamic Finite Element Analyses of the Long Valley Dam," by D.V. Griffiths and J.H. Prevost, 6/17/88, (PB89-144711).
- NCEER-88-0016 "Damage Assessment of Reinforced Concrete Structures in Eastern United States," by A.M. Reinhorn, M.J. Seidel, S.K. Kunnath and Y.J. Park, 6/15/88, (PB89-122220).
- NCEER-88-0017 "Dynamic Compliance of Vertically Loaded Strip Foundations in Multilayered Viscoelastic Soils," by S. Ahmad and A.S.M. Israil, 6/17/88, (PB89-102891).
- NCEER-88-0018 "An Experimental Study of Seismic Structural Response With Added Viscoelastic Dampers," by R.C. Lin, Z. Liang, T.T. Soong and R.H. Zhang, 6/30/88, (PB89-122212). This report is available only through NTIS (see address given above).
- NCEER-88-0019 "Experimental Investigation of Primary Secondary System Interaction," by G.D. Manolis, G. Juhn and A.M. Reinhorn, 5/27/88, (PB89-122204).
- NCEER-88-0020 "A Response Spectrum Approach For Analysis of Nonclassically Damped Structures," by J.N. Yang, S. Sarkani and F.X. Long, 4/22/88, (PB89-102909).
- NCEER-88-0021 "Seismic Interaction of Structures and Soils: Stochastic Approach," by A.S. Veletsos and A.M. Prasad, 7/21/88, (PB89-122196).
- NCEER-88-0022 "Identification of the Serviceability Limit State and Detection of Seismic Structural Damage," by E. DiPasquale and A.S. Cakmak, 6/15/88, (PB89-122188). This report is available only through NTIS (see address given above).
- NCEER-88-0023 "Multi-Hazard Risk Analysis: Case of a Simple Offshore Structure," by B.K. Bhartia and E.H. Vanmarcke, 7/21/88, (PB89-145213).
- NCEER-88-0024 "Automated Seismic Design of Reinforced Concrete Buildings," by Y.S. Chung, C. Meyer and M. Shinozuka, 7/5/88, (PB89-122170). This report is available only through NTIS (see address given above).

- NCEER-88-0025 "Experimental Study of Active Control of MDOF Structures Under Seismic Excitations," by L.L. Chung, R.C. Lin, T.T. Soong and A.M. Reinhorn, 7/10/88, (PB89-122600).
- NCEER-88-0026 "Earthquake Simulation Tests of a Low-Rise Metal Structure," by J.S. Hwang, K.C. Chang, G.C. Lee and R.L. Ketter, 8/1/88, (PB89-102917).
- NCEER-88-0027 "Systems Study of Urban Response and Reconstruction Due to Catastrophic Earthquakes," by F. Kozin and H.K. Zhou, 9/22/88, (PB90-162348).
- NCEER-88-0028 "Seismic Fragility Analysis of Plane Frame Structures," by H.H-M. Hwang and Y.K. Low, 7/31/88, (PB89-131445).
- NCEER-88-0029 "Response Analysis of Stochastic Structures," by A. Kardara, C. Bucher and M. Shinozuka, 9/22/88, (PB89-174429).
- NCEER-88-0030 "Nonnormal Accelerations Due to Yielding in a Primary Structure," by D.C.K. Chen and L.D. Lutes, 9/19/88, (PB89-131437).
- NCEER-88-0031 "Design Approaches for Soil-Structure Interaction," by A.S. Veletsos, A.M. Prasad and Y. Tang, 12/30/88, (PB89-174437). This report is available only through NTIS (see address given above).
- NCEER-88-0032 "A Re-evaluation of Design Spectra for Seismic Damage Control," by C.J. Turkstra and A.G. Tallin, 11/7/88, (PB89-145221).
- NCEER-88-0033 "The Behavior and Design of Noncontact Lap Splices Subjected to Repeated Inelastic Tensile Loading," by V.E. Sagan, P. Gergely and R.N. White, 12/8/88, (PB89-163737).
- NCEER-88-0034 "Seismic Response of Pile Foundations," by S.M. Mamoon, P.K. Banerjee and S. Ahmad, 11/1/88, (PB89-145239).
- NCEER-88-0035 "Modeling of R/C Building Structures With Flexible Floor Diaphragms (IDARC2)," by A.M. Reinhorn, S.K. Kunnath and N. Panahshahi, 9/7/88, (PB89-207153).
- NCEER-88-0036 "Solution of the Dam-Reservoir Interaction Problem Using a Combination of FEM, BEM with Particular Integrals, Modal Analysis, and Substructuring," by C-S. Tsai, G.C. Lee and R.L. Ketter, 12/31/88, (PB89-207146).
- NCEER-88-0037 "Optimal Placement of Actuators for Structural Control," by F.Y. Cheng and C.P. Pantelides, 8/15/88, (PB89-162846).
- NCEER-88-0038 "Teflon Bearings in Aseismic Base Isolation: Experimental Studies and Mathematical Modeling," by A. Mokha, M.C. Constantinou and A.M. Reinhorn, 12/5/88, (PB89-218457). This report is available only through NTIS (see address given above).
- NCEER-88-0039 "Seismic Behavior of Flat Slab High-Rise Buildings in the New York City Area," by P. Weidlinger and M. Ettouney, 10/15/88, (PB90-145681).
- NCEER-88-0040 "Evaluation of the Earthquake Resistance of Existing Buildings in New York City," by P. Weidlinger and M. Ettouney, 10/15/88, to be published.
- NCEER-88-0041 "Small-Scale Modeling Techniques for Reinforced Concrete Structures Subjected to Seismic Loads," by W. Kim, A. El-Attar and R.N. White, 11/22/88, (PB89-189625).
- NCEER-88-0042 "Modeling Strong Ground Motion from Multiple Event Earthquakes," by G.W. Ellis and A.S. Cakmak, 10/15/88, (PB89-174445).

- NCEER-88-0043 "Nonstationary Models of Seismic Ground Acceleration," by M. Grigoriu, S.E. Ruiz and E. Rosenblueth, 7/15/88, (PB89-189617).
- NCEER-88-0044 "SARCF User's Guide: Seismic Analysis of Reinforced Concrete Frames," by Y.S. Chung, C. Meyer and M. Shinozuka, 11/9/88, (PB89-174452).
- NCEER-88-0045 "First Expert Panel Meeting on Disaster Research and Planning," edited by J. Pantelic and J. Stoyle, 9/15/88, (PB89-174460).
- NCEER-88-0046 "Preliminary Studies of the Effect of Degrading Infill Walls on the Nonlinear Seismic Response of Steel Frames," by C.Z. Chrysostomou, P. Gergely and J.F. Abel, 12/19/88, (PB89-208383).
- NCEER-88-0047 "Reinforced Concrete Frame Component Testing Facility Design, Construction, Instrumentation and Operation," by S.P. Pessiki, C. Conley, T. Bond, P. Gergely and R.N. White, 12/16/88, (PB89-174478).
- NCEER-89-0001 "Effects of Protective Cushion and Soil Compliancy on the Response of Equipment Within a Seismically Excited Building," by J.A. HoLung, 2/16/89, (PB89-207179).
- NCEER-89-0002 "Statistical Evaluation of Response Modification Factors for Reinforced Concrete Structures," by H.H-M. Hwang and J-W. Jaw, 2/17/89, (PB89-207187).
- NCEER-89-0003 "Hysteretic Columns Under Random Excitation," by G-Q. Cai and Y.K. Lin, 1/9/89, (PB89-196513).
- NCEER-89-0004 "Experimental Study of `Elephant Foot Bulge' Instability of Thin-Walled Metal Tanks," by Z-H. Jia and R.L. Ketter, 2/22/89, (PB89-207195).
- NCEER-89-0005 "Experiment on Performance of Buried Pipelines Across San Andreas Fault," by J. Isenberg, E. Richardson and T.D. O'Rourke, 3/10/89, (PB89-218440). This report is available only through NTIS (see address given above).
- NCEER-89-0006 "A Knowledge-Based Approach to Structural Design of Earthquake-Resistant Buildings," by M. Subramani, P. Gergely, C.H. Conley, J.F. Abel and A.H. Zaghw, 1/15/89, (PB89-218465).
- NCEER-89-0007 "Liquefaction Hazards and Their Effects on Buried Pipelines," by T.D. O'Rourke and P.A. Lane, 2/1/89, (PB89-218481).
- NCEER-89-0008 "Fundamentals of System Identification in Structural Dynamics," by H. Imai, C-B. Yun, O. Maruyama and M. Shinozuka, 1/26/89, (PB89-207211).
- NCEER-89-0009 "Effects of the 1985 Michoacan Earthquake on Water Systems and Other Buried Lifelines in Mexico," by A.G. Ayala and M.J. O'Rourke, 3/8/89, (PB89-207229).
- NCEER-89-R010 "NCEER Bibliography of Earthquake Education Materials," by K.E.K. Ross, Second Revision, 9/1/89, (PB90-125352).
- NCEER-89-0011 "Inelastic Three-Dimensional Response Analysis of Reinforced Concrete Building Structures (IDARC-3D), Part I - Modeling," by S.K. Kunnath and A.M. Reinhorn, 4/17/89, (PB90-114612).
- NCEER-89-0012 "Recommended Modifications to ATC-14," by C.D. Poland and J.O. Malley, 4/12/89, (PB90-108648).
- NCEER-89-0013 "Repair and Strengthening of Beam-to-Column Connections Subjected to Earthquake Loading," by M. Corazao and A.J. Durrani, 2/28/89, (PB90-109885).
- NCEER-89-0014 "Program EXKAL2 for Identification of Structural Dynamic Systems," by O. Maruyama, C-B. Yun, M. Hoshiya and M. Shinozuka, 5/19/89, (PB90-109877).

- NCEER-89-0015 "Response of Frames With Bolted Semi-Rigid Connections, Part I Experimental Study and Analytical Predictions," by P.J. DiCorso, A.M. Reinhorn, J.R. Dickerson, J.B. Radziminski and W.L. Harper, 6/1/89, to be published.
- NCEER-89-0016 "ARMA Monte Carlo Simulation in Probabilistic Structural Analysis," by P.D. Spanos and M.P. Mignolet, 7/10/89, (PB90-109893).
- NCEER-89-P017 "Preliminary Proceedings from the Conference on Disaster Preparedness The Place of Earthquake Education in Our Schools," Edited by K.E.K. Ross, 6/23/89, (PB90-108606).
- NCEER-89-0017 "Proceedings from the Conference on Disaster Preparedness The Place of Earthquake Education in Our Schools," Edited by K.E.K. Ross, 12/31/89, (PB90-207895). This report is available only through NTIS (see address given above).
- NCEER-89-0018 "Multidimensional Models of Hysteretic Material Behavior for Vibration Analysis of Shape Memory Energy Absorbing Devices, by E.J. Graesser and F.A. Cozzarelli, 6/7/89, (PB90-164146).
- NCEER-89-0019 "Nonlinear Dynamic Analysis of Three-Dimensional Base Isolated Structures (3D-BASIS)," by S. Nagarajaiah, A.M. Reinhorn and M.C. Constantinou, 8/3/89, (PB90-161936). This report is available only through NTIS (see address given above).
- NCEER-89-0020 "Structural Control Considering Time-Rate of Control Forces and Control Rate Constraints," by F.Y. Cheng and C.P. Pantelides, 8/3/89, (PB90-120445).
- NCEER-89-0021 "Subsurface Conditions of Memphis and Shelby County," by K.W. Ng, T-S. Chang and H-H.M. Hwang, 7/26/89, (PB90-120437).
- NCEER-89-0022 "Seismic Wave Propagation Effects on Straight Jointed Buried Pipelines," by K. Elhmadi and M.J. O'Rourke, 8/24/89, (PB90-162322).
- NCEER-89-0023 "Workshop on Serviceability Analysis of Water Delivery Systems," edited by M. Grigoriu, 3/6/89, (PB90-127424).
- NCEER-89-0024 "Shaking Table Study of a 1/5 Scale Steel Frame Composed of Tapered Members," by K.C. Chang, J.S. Hwang and G.C. Lee, 9/18/89, (PB90-160169).
- NCEER-89-0025 "DYNA1D: A Computer Program for Nonlinear Seismic Site Response Analysis Technical Documentation," by Jean H. Prevost, 9/14/89, (PB90-161944). This report is available only through NTIS (see address given above).
- NCEER-89-0026 "1:4 Scale Model Studies of Active Tendon Systems and Active Mass Dampers for Aseismic Protection," by A.M. Reinhorn, T.T. Soong, R.C. Lin, Y.P. Yang, Y. Fukao, H. Abe and M. Nakai, 9/15/89, (PB90-173246).
- NCEER-89-0027 "Scattering of Waves by Inclusions in a Nonhomogeneous Elastic Half Space Solved by Boundary Element Methods," by P.K. Hadley, A. Askar and A.S. Cakmak, 6/15/89, (PB90-145699).
- NCEER-89-0028 "Statistical Evaluation of Deflection Amplification Factors for Reinforced Concrete Structures," by H.H.M. Hwang, J-W. Jaw and A.L. Ch'ng, 8/31/89, (PB90-164633).
- NCEER-89-0029 "Bedrock Accelerations in Memphis Area Due to Large New Madrid Earthquakes," by H.H.M. Hwang, C.H.S. Chen and G. Yu, 11/7/89, (PB90-162330).
- NCEER-89-0030 "Seismic Behavior and Response Sensitivity of Secondary Structural Systems," by Y.Q. Chen and T.T. Soong, 10/23/89, (PB90-164658).

- NCEER-89-0031 "Random Vibration and Reliability Analysis of Primary-Secondary Structural Systems," by Y. Ibrahim, M. Grigoriu and T.T. Soong, 11/10/89, (PB90-161951).
- NCEER-89-0032 "Proceedings from the Second U.S. Japan Workshop on Liquefaction, Large Ground Deformation and Their Effects on Lifelines, September 26-29, 1989," Edited by T.D. O'Rourke and M. Hamada, 12/1/89, (PB90-209388).
- NCEER-89-0033 "Deterministic Model for Seismic Damage Evaluation of Reinforced Concrete Structures," by J.M. Bracci, A.M. Reinhorn, J.B. Mander and S.K. Kunnath, 9/27/89.
- NCEER-89-0034 "On the Relation Between Local and Global Damage Indices," by E. DiPasquale and A.S. Cakmak, 8/15/89, (PB90-173865).
- NCEER-89-0035 "Cyclic Undrained Behavior of Nonplastic and Low Plasticity Silts," by A.J. Walker and H.E. Stewart, 7/26/89, (PB90-183518).
- NCEER-89-0036 "Liquefaction Potential of Surficial Deposits in the City of Buffalo, New York," by M. Budhu, R. Giese and L. Baumgrass, 1/17/89, (PB90-208455).
- NCEER-89-0037 "A Deterministic Assessment of Effects of Ground Motion Incoherence," by A.S. Veletsos and Y. Tang, 7/15/89, (PB90-164294).
- NCEER-89-0038 "Workshop on Ground Motion Parameters for Seismic Hazard Mapping," July 17-18, 1989, edited by R.V. Whitman, 12/1/89, (PB90-173923).
- NCEER-89-0039 "Seismic Effects on Elevated Transit Lines of the New York City Transit Authority," by C.J. Costantino, C.A. Miller and E. Heymsfield, 12/26/89, (PB90-207887).
- NCEER-89-0040 "Centrifugal Modeling of Dynamic Soil-Structure Interaction," by K. Weissman, Supervised by J.H. Prevost, 5/10/89, (PB90-207879).
- NCEER-89-0041 "Linearized Identification of Buildings With Cores for Seismic Vulnerability Assessment," by I-K. Ho and A.E. Aktan, 11/1/89, (PB90-251943).
- NCEER-90-0001 "Geotechnical and Lifeline Aspects of the October 17, 1989 Loma Prieta Earthquake in San Francisco," by T.D. O'Rourke, H.E. Stewart, F.T. Blackburn and T.S. Dickerman, 1/90, (PB90-208596).
- NCEER-90-0002 "Nonnormal Secondary Response Due to Yielding in a Primary Structure," by D.C.K. Chen and L.D. Lutes, 2/28/90, (PB90-251976).
- NCEER-90-0003 "Earthquake Education Materials for Grades K-12," by K.E.K. Ross, 4/16/90, (PB91-251984).
- NCEER-90-0004 "Catalog of Strong Motion Stations in Eastern North America," by R.W. Busby, 4/3/90, (PB90-251984).
- NCEER-90-0005 "NCEER Strong-Motion Data Base: A User Manual for the GeoBase Release (Version 1.0 for the Sun3)," by P. Friberg and K. Jacob, 3/31/90 (PB90-258062).
- NCEER-90-0006 "Seismic Hazard Along a Crude Oil Pipeline in the Event of an 1811-1812 Type New Madrid Earthquake," by H.H.M. Hwang and C-H.S. Chen, 4/16/90(PB90-258054).
- NCEER-90-0007 "Site-Specific Response Spectra for Memphis Sheahan Pumping Station," by H.H.M. Hwang and C.S. Lee, 5/15/90, (PB91-108811).
- NCEER-90-0008 "Pilot Study on Seismic Vulnerability of Crude Oil Transmission Systems," by T. Ariman, R. Dobry, M. Grigoriu, F. Kozin, M. O'Rourke, T. O'Rourke and M. Shinozuka, 5/25/90, (PB91-108837).

- NCEER-90-0009 "A Program to Generate Site Dependent Time Histories: EQGEN," by G.W. Ellis, M. Srinivasan and A.S. Cakmak, 1/30/90, (PB91-108829).
- NCEER-90-0010 "Active Isolation for Seismic Protection of Operating Rooms," by M.E. Talbott, Supervised by M. Shinozuka, 6/8/9, (PB91-110205).
- NCEER-90-0011 "Program LINEARID for Identification of Linear Structural Dynamic Systems," by C-B. Yun and M. Shinozuka, 6/25/90, (PB91-110312).
- NCEER-90-0012 "Two-Dimensional Two-Phase Elasto-Plastic Seismic Response of Earth Dams," by A.N. Yiagos, Supervised by J.H. Prevost, 6/20/90, (PB91-110197).
- NCEER-90-0013 "Secondary Systems in Base-Isolated Structures: Experimental Investigation, Stochastic Response and Stochastic Sensitivity," by G.D. Manolis, G. Juhn, M.C. Constantinou and A.M. Reinhorn, 7/1/90, (PB91-110320).
- NCEER-90-0014 "Seismic Behavior of Lightly-Reinforced Concrete Column and Beam-Column Joint Details," by S.P. Pessiki, C.H. Conley, P. Gergely and R.N. White, 8/22/90, (PB91-108795).
- NCEER-90-0015 "Two Hybrid Control Systems for Building Structures Under Strong Earthquakes," by J.N. Yang and A. Danielians, 6/29/90, (PB91-125393).
- NCEER-90-0016 "Instantaneous Optimal Control with Acceleration and Velocity Feedback," by J.N. Yang and Z. Li, 6/29/90, (PB91-125401).
- NCEER-90-0017 "Reconnaissance Report on the Northern Iran Earthquake of June 21, 1990," by M. Mehrain, 10/4/90, (PB91-125377).
- NCEER-90-0018 "Evaluation of Liquefaction Potential in Memphis and Shelby County," by T.S. Chang, P.S. Tang, C.S. Lee and H. Hwang, 8/10/90, (PB91-125427).
- NCEER-90-0019 "Experimental and Analytical Study of a Combined Sliding Disc Bearing and Helical Steel Spring Isolation System," by M.C. Constantinou, A.S. Mokha and A.M. Reinhorn, 10/4/90, (PB91-125385).
- NCEER-90-0020 "Experimental Study and Analytical Prediction of Earthquake Response of a Sliding Isolation System with a Spherical Surface," by A.S. Mokha, M.C. Constantinou and A.M. Reinhorn, 10/11/90, (PB91-125419).
- NCEER-90-0021 "Dynamic Interaction Factors for Floating Pile Groups," by G. Gazetas, K. Fan, A. Kaynia and E. Kausel, 9/10/90, (PB91-170381).
- NCEER-90-0022 "Evaluation of Seismic Damage Indices for Reinforced Concrete Structures," by S. Rodriguez-Gomez and A.S. Cakmak, 9/30/90, PB91-171322).
- NCEER-90-0023 "Study of Site Response at a Selected Memphis Site," by H. Desai, S. Ahmad, E.S. Gazetas and M.R. Oh, 10/11/90, (PB91-196857).
- NCEER-90-0024 "A User's Guide to Strongmo: Version 1.0 of NCEER's Strong-Motion Data Access Tool for PCs and Terminals," by P.A. Friberg and C.A.T. Susch, 11/15/90, (PB91-171272).
- NCEER-90-0025 "A Three-Dimensional Analytical Study of Spatial Variability of Seismic Ground Motions," by L-L. Hong and A.H.-S. Ang, 10/30/90, (PB91-170399).
- NCEER-90-0026 "MUMOID User's Guide A Program for the Identification of Modal Parameters," by S. Rodriguez-Gomez and E. DiPasquale, 9/30/90, (PB91-171298).
- NCEER-90-0027 "SARCF-II User's Guide Seismic Analysis of Reinforced Concrete Frames," by S. Rodriguez-Gomez, Y.S. Chung and C. Meyer, 9/30/90, (PB91-171280).

- NCEER-90-0028 "Viscous Dampers: Testing, Modeling and Application in Vibration and Seismic Isolation," by N. Makris and M.C. Constantinou, 12/20/90 (PB91-190561).
- NCEER-90-0029 "Soil Effects on Earthquake Ground Motions in the Memphis Area," by H. Hwang, C.S. Lee, K.W. Ng and T.S. Chang, 8/2/90, (PB91-190751).
- NCEER-91-0001 "Proceedings from the Third Japan-U.S. Workshop on Earthquake Resistant Design of Lifeline Facilities and Countermeasures for Soil Liquefaction, December 17-19, 1990," edited by T.D. O'Rourke and M. Hamada, 2/1/91, (PB91-179259).
- NCEER-91-0002 "Physical Space Solutions of Non-Proportionally Damped Systems," by M. Tong, Z. Liang and G.C. Lee, 1/15/91, (PB91-179242).
- NCEER-91-0003 "Seismic Response of Single Piles and Pile Groups," by K. Fan and G. Gazetas, 1/10/91, (PB92-174994).
- NCEER-91-0004 "Damping of Structures: Part 1 Theory of Complex Damping," by Z. Liang and G. Lee, 10/10/91, (PB92-197235).
- NCEER-91-0005 "3D-BASIS Nonlinear Dynamic Analysis of Three Dimensional Base Isolated Structures: Part II," by S. Nagarajaiah, A.M. Reinhorn and M.C. Constantinou, 2/28/91, (PB91-190553).
- NCEER-91-0006 "A Multidimensional Hysteretic Model for Plasticity Deforming Metals in Energy Absorbing Devices," by E.J. Graesser and F.A. Cozzarelli, 4/9/91, (PB92-108364).
- NCEER-91-0007 "A Framework for Customizable Knowledge-Based Expert Systems with an Application to a KBES for Evaluating the Seismic Resistance of Existing Buildings," by E.G. Ibarra-Anaya and S.J. Fenves, 4/9/91, (PB91-210930).
- NCEER-91-0008 "Nonlinear Analysis of Steel Frames with Semi-Rigid Connections Using the Capacity Spectrum Method," by G.G. Deierlein, S-H. Hsieh, Y-J. Shen and J.F. Abel, 7/2/91, (PB92-113828).
- NCEER-91-0009 "Earthquake Education Materials for Grades K-12," by K.E.K. Ross, 4/30/91, (PB91-212142).
- NCEER-91-0010 "Phase Wave Velocities and Displacement Phase Differences in a Harmonically Oscillating Pile," by N. Makris and G. Gazetas, 7/8/91, (PB92-108356).
- NCEER-91-0011 "Dynamic Characteristics of a Full-Size Five-Story Steel Structure and a 2/5 Scale Model," by K.C. Chang, G.C. Yao, G.C. Lee, D.S. Hao and Y.C. Yeh, "7/2/91, (PB93-116648).
- NCEER-91-0012 "Seismic Response of a 2/5 Scale Steel Structure with Added Viscoelastic Dampers," by K.C. Chang, T.T. Soong, S-T. Oh and M.L. Lai, 5/17/91, (PB92-110816).
- NCEER-91-0013 "Earthquake Response of Retaining Walls; Full-Scale Testing and Computational Modeling," by S. Alampalli and A-W.M. Elgamal, 6/20/91, to be published.
- NCEER-91-0014 "3D-BASIS-M: Nonlinear Dynamic Analysis of Multiple Building Base Isolated Structures," by P.C. Tsopelas, S. Nagarajaiah, M.C. Constantinou and A.M. Reinhorn, 5/28/91, (PB92-113885).
- NCEER-91-0015 "Evaluation of SEAOC Design Requirements for Sliding Isolated Structures," by D. Theodossiou and M.C. Constantinou, 6/10/91, (PB92-114602).
- NCEER-91-0016 "Closed-Loop Modal Testing of a 27-Story Reinforced Concrete Flat Plate-Core Building," by H.R. Somaprasad, T. Toksoy, H. Yoshiyuki and A.E. Aktan, 7/15/91, (PB92-129980).
- NCEER-91-0017 "Shake Table Test of a 1/6 Scale Two-Story Lightly Reinforced Concrete Building," by A.G. El-Attar, R.N. White and P. Gergely, 2/28/91, (PB92-222447).

- NCEER-91-0018 "Shake Table Test of a 1/8 Scale Three-Story Lightly Reinforced Concrete Building," by A.G. El-Attar, R.N. White and P. Gergely, 2/28/91, (PB93-116630).
- NCEER-91-0019 "Transfer Functions for Rigid Rectangular Foundations," by A.S. Veletsos, A.M. Prasad and W.H. Wu, 7/31/91.
- NCEER-91-0020 "Hybrid Control of Seismic-Excited Nonlinear and Inelastic Structural Systems," by J.N. Yang, Z. Li and A. Danielians, 8/1/91, (PB92-143171).
- NCEER-91-0021 "The NCEER-91 Earthquake Catalog: Improved Intensity-Based Magnitudes and Recurrence Relations for U.S. Earthquakes East of New Madrid," by L. Seeber and J.G. Armbruster, 8/28/91, (PB92-176742).
- NCEER-91-0022 "Proceedings from the Implementation of Earthquake Planning and Education in Schools: The Need for Change The Roles of the Changemakers," by K.E.K. Ross and F. Winslow, 7/23/91, (PB92-129998).
- NCEER-91-0023 "A Study of Reliability-Based Criteria for Seismic Design of Reinforced Concrete Frame Buildings," by H.H.M. Hwang and H-M. Hsu, 8/10/91, (PB92-140235).
- NCEER-91-0024 "Experimental Verification of a Number of Structural System Identification Algorithms," by R.G. Ghanem, H. Gavin and M. Shinozuka, 9/18/91, (PB92-176577).
- NCEER-91-0025 "Probabilistic Evaluation of Liquefaction Potential," by H.H.M. Hwang and C.S. Lee," 11/25/91, (PB92-143429).
- NCEER-91-0026 "Instantaneous Optimal Control for Linear, Nonlinear and Hysteretic Structures Stable Controllers," by J.N. Yang and Z. Li, 11/15/91, (PB92-163807).
- NCEER-91-0027 "Experimental and Theoretical Study of a Sliding Isolation System for Bridges," by M.C. Constantinou, A. Kartoum, A.M. Reinhorn and P. Bradford, 11/15/91, (PB92-176973).
- NCEER-92-0001 "Case Studies of Liquefaction and Lifeline Performance During Past Earthquakes, Volume 1: Japanese Case Studies," Edited by M. Hamada and T. O'Rourke, 2/17/92, (PB92-197243).
- NCEER-92-0002 "Case Studies of Liquefaction and Lifeline Performance During Past Earthquakes, Volume 2: United States Case Studies," Edited by T. O'Rourke and M. Hamada, 2/17/92, (PB92-197250).
- NCEER-92-0003 "Issues in Earthquake Education," Edited by K. Ross, 2/3/92, (PB92-222389).
- NCEER-92-0004 "Proceedings from the First U.S. Japan Workshop on Earthquake Protective Systems for Bridges," Edited by I.G. Buckle, 2/4/92, (PB94-142239, A99, MF-A06).
- NCEER-92-0005 "Seismic Ground Motion from a Haskell-Type Source in a Multiple-Layered Half-Space," A.P. Theoharis, G. Deodatis and M. Shinozuka, 1/2/92, to be published.
- NCEER-92-0006 "Proceedings from the Site Effects Workshop," Edited by R. Whitman, 2/29/92, (PB92-197201).
- NCEER-92-0007 "Engineering Evaluation of Permanent Ground Deformations Due to Seismically-Induced Liquefaction," by M.H. Baziar, R. Dobry and A-W.M. Elgamal, 3/24/92, (PB92-222421).
- NCEER-92-0008 "A Procedure for the Seismic Evaluation of Buildings in the Central and Eastern United States," by C.D. Poland and J.O. Malley, 4/2/92, (PB92-222439).
- NCEER-92-0009 "Experimental and Analytical Study of a Hybrid Isolation System Using Friction Controllable Sliding Bearings," by M.Q. Feng, S. Fujii and M. Shinozuka, 5/15/92, (PB93-150282).
- NCEER-92-0010 "Seismic Resistance of Slab-Column Connections in Existing Non-Ductile Flat-Plate Buildings," by A.J. Durrani and Y. Du, 5/18/92.

- NCEER-92-0011 "The Hysteretic and Dynamic Behavior of Brick Masonry Walls Upgraded by Ferrocement Coatings Under Cyclic Loading and Strong Simulated Ground Motion," by H. Lee and S.P. Prawel, 5/11/92, to be published.
- NCEER-92-0012 "Study of Wire Rope Systems for Seismic Protection of Equipment in Buildings," by G.F. Demetriades, M.C. Constantinou and A.M. Reinhorn, 5/20/92.
- NCEER-92-0013 "Shape Memory Structural Dampers: Material Properties, Design and Seismic Testing," by P.R. Witting and F.A. Cozzarelli, 5/26/92.
- NCEER-92-0014 "Longitudinal Permanent Ground Deformation Effects on Buried Continuous Pipelines," by M.J. O'Rourke, and C. Nordberg, 6/15/92.
- NCEER-92-0015 "A Simulation Method for Stationary Gaussian Random Functions Based on the Sampling Theorem," by M. Grigoriu and S. Balopoulou, 6/11/92, (PB93-127496).
- NCEER-92-0016 "Gravity-Load-Designed Reinforced Concrete Buildings: Seismic Evaluation of Existing Construction and Detailing Strategies for Improved Seismic Resistance," by G.W. Hoffmann, S.K. Kunnath, A.M. Reinhorn and J.B. Mander, 7/15/92, (PB94-142007, A08, MF-A02).
- NCEER-92-0017 "Observations on Water System and Pipeline Performance in the Limón Area of Costa Rica Due to the April 22, 1991 Earthquake," by M. O'Rourke and D. Ballantyne, 6/30/92, (PB93-126811).
- NCEER-92-0018 "Fourth Edition of Earthquake Education Materials for Grades K-12," Edited by K.E.K. Ross, 8/10/92.
- NCEER-92-0019 "Proceedings from the Fourth Japan-U.S. Workshop on Earthquake Resistant Design of Lifeline Facilities and Countermeasures for Soil Liquefaction," Edited by M. Hamada and T.D. O'Rourke, 8/12/92, (PB93-163939).
- NCEER-92-0020 "Active Bracing System: A Full Scale Implementation of Active Control," by A.M. Reinhorn, T.T. Soong, R.C. Lin, M.A. Riley, Y.P. Wang, S. Aizawa and M. Higashino, 8/14/92, (PB93-127512).
- NCEER-92-0021 "Empirical Analysis of Horizontal Ground Displacement Generated by Liquefaction-Induced Lateral Spreads," by S.F. Bartlett and T.L. Youd, 8/17/92, (PB93-188241).
- NCEER-92-0022 "IDARC Version 3.0: Inelastic Damage Analysis of Reinforced Concrete Structures," by S.K. Kunnath, A.M. Reinhorn and R.F. Lobo, 8/31/92, (PB93-227502, A07, MF-A02).
- NCEER-92-0023 "A Semi-Empirical Analysis of Strong-Motion Peaks in Terms of Seismic Source, Propagation Path and Local Site Conditions, by M. Kamiyama, M.J. O'Rourke and R. Flores-Berrones, 9/9/92, (PB93-150266).
- NCEER-92-0024 "Seismic Behavior of Reinforced Concrete Frame Structures with Nonductile Details, Part I: Summary of Experimental Findings of Full Scale Beam-Column Joint Tests," by A. Beres, R.N. White and P. Gergely, 9/30/92, (PB93-227783, A05, MF-A01).
- NCEER-92-0025 "Experimental Results of Repaired and Retrofitted Beam-Column Joint Tests in Lightly Reinforced Concrete Frame Buildings," by A. Beres, S. El-Borgi, R.N. White and P. Gergely, 10/29/92, (PB93-227791, A05, MF-A01).
- NCEER-92-0026 "A Generalization of Optimal Control Theory: Linear and Nonlinear Structures," by J.N. Yang, Z. Li and S. Vongchavalitkul, 11/2/92, (PB93-188621).
- NCEER-92-0027 "Seismic Resistance of Reinforced Concrete Frame Structures Designed Only for Gravity Loads: Part I -Design and Properties of a One-Third Scale Model Structure," by J.M. Bracci, A.M. Reinhorn and J.B. Mander, 12/1/92, (PB94-104502, A08, MF-A02).

- NCEER-92-0028 "Seismic Resistance of Reinforced Concrete Frame Structures Designed Only for Gravity Loads: Part II -Experimental Performance of Subassemblages," by L.E. Aycardi, J.B. Mander and A.M. Reinhorn, 12/1/92, (PB94-104510, A08, MF-A02).
- NCEER-92-0029 "Seismic Resistance of Reinforced Concrete Frame Structures Designed Only for Gravity Loads: Part III -Experimental Performance and Analytical Study of a Structural Model," by J.M. Bracci, A.M. Reinhorn and J.B. Mander, 12/1/92, (PB93-227528, A09, MF-A01).
- NCEER-92-0030 "Evaluation of Seismic Retrofit of Reinforced Concrete Frame Structures: Part I Experimental Performance of Retrofitted Subassemblages," by D. Choudhuri, J.B. Mander and A.M. Reinhorn, 12/8/92, (PB93-198307, A07, MF-A02).
- NCEER-92-0031 "Evaluation of Seismic Retrofit of Reinforced Concrete Frame Structures: Part II Experimental Performance and Analytical Study of a Retrofitted Structural Model," by J.M. Bracci, A.M. Reinhorn and J.B. Mander, 12/8/92, (PB93-198315, A09, MF-A03).
- NCEER-92-0032 "Experimental and Analytical Investigation of Seismic Response of Structures with Supplemental Fluid Viscous Dampers," by M.C. Constantinou and M.D. Symans, 12/21/92, (PB93-191435).
- NCEER-92-0033 "Reconnaissance Report on the Cairo, Egypt Earthquake of October 12, 1992," by M. Khater, 12/23/92, (PB93-188621).
- NCEER-92-0034 "Low-Level Dynamic Characteristics of Four Tall Flat-Plate Buildings in New York City," by H. Gavin, S. Yuan, J. Grossman, E. Pekelis and K. Jacob, 12/28/92, (PB93-188217).
- NCEER-93-0001 "An Experimental Study on the Seismic Performance of Brick-Infilled Steel Frames With and Without Retrofit," by J.B. Mander, B. Nair, K. Wojtkowski and J. Ma, 1/29/93, (PB93-227510, A07, MF-A02).
- NCEER-93-0002 "Social Accounting for Disaster Preparedness and Recovery Planning," by S. Cole, E. Pantoja and V. Razak, 2/22/93, (PB94-142114, A12, MF-A03).
- NCEER-93-0003 "Assessment of 1991 NEHRP Provisions for Nonstructural Components and Recommended Revisions," by T.T. Soong, G. Chen, Z. Wu, R-H. Zhang and M. Grigoriu, 3/1/93, (PB93-188639).
- NCEER-93-0004 "Evaluation of Static and Response Spectrum Analysis Procedures of SEAOC/UBC for Seismic Isolated Structures," by C.W. Winters and M.C. Constantinou, 3/23/93, (PB93-198299).
- NCEER-93-0005 "Earthquakes in the Northeast Are We Ignoring the Hazard? A Workshop on Earthquake Science and Safety for Educators," edited by K.E.K. Ross, 4/2/93, (PB94-103066, A09, MF-A02).
- NCEER-93-0006 "Inelastic Response of Reinforced Concrete Structures with Viscoelastic Braces," by R.F. Lobo, J.M. Bracci, K.L. Shen, A.M. Reinhorn and T.T. Soong, 4/5/93, (PB93-227486, A05, MF-A02).
- NCEER-93-0007 "Seismic Testing of Installation Methods for Computers and Data Processing Equipment," by K. Kosar, T.T. Soong, K.L. Shen, J.A. HoLung and Y.K. Lin, 4/12/93, (PB93-198299).
- NCEER-93-0008 "Retrofit of Reinforced Concrete Frames Using Added Dampers," by A. Reinhorn, M. Constantinou and C. Li, to be published.
- NCEER-93-0009 "Seismic Behavior and Design Guidelines for Steel Frame Structures with Added Viscoelastic Dampers," by K.C. Chang, M.L. Lai, T.T. Soong, D.S. Hao and Y.C. Yeh, 5/1/93, (PB94-141959, A07, MF-A02).
- NCEER-93-0010 "Seismic Performance of Shear-Critical Reinforced Concrete Bridge Piers," by J.B. Mander, S.M. Waheed, M.T.A. Chaudhary and S.S. Chen, 5/12/93, (PB93-227494, A08, MF-A02).

- NCEER-93-0011 "3D-BASIS-TABS: Computer Program for Nonlinear Dynamic Analysis of Three Dimensional Base Isolated Structures," by S. Nagarajaiah, C. Li, A.M. Reinhorn and M.C. Constantinou, 8/2/93, (PB94-141819, A09, MF-A02).
- NCEER-93-0012 "Effects of Hydrocarbon Spills from an Oil Pipeline Break on Ground Water," by O.J. Helweg and H.H.M. Hwang, 8/3/93, (PB94-141942, A06, MF-A02).
- NCEER-93-0013 "Simplified Procedures for Seismic Design of Nonstructural Components and Assessment of Current Code Provisions," by M.P. Singh, L.E. Suarez, E.E. Matheu and G.O. Maldonado, 8/4/93, (PB94-141827, A09, MF-A02).
- NCEER-93-0014 "An Energy Approach to Seismic Analysis and Design of Secondary Systems," by G. Chen and T.T. Soong, 8/6/93, (PB94-142767, A11, MF-A03).
- NCEER-93-0015 "Proceedings from School Sites: Becoming Prepared for Earthquakes Commemorating the Third Anniversary of the Loma Prieta Earthquake," Edited by F.E. Winslow and K.E.K. Ross, 8/16/93.
- NCEER-93-0016 "Reconnaissance Report of Damage to Historic Monuments in Cairo, Egypt Following the October 12, 1992 Dahshur Earthquake," by D. Sykora, D. Look, G. Croci, E. Karaesmen and E. Karaesmen, 8/19/93, (PB94-142221, A08, MF-A02).
- NCEER-93-0017 "The Island of Guam Earthquake of August 8, 1993," by S.W. Swan and S.K. Harris, 9/30/93, (PB94-141843, A04, MF-A01).
- NCEER-93-0018 "Engineering Aspects of the October 12, 1992 Egyptian Earthquake," by A.W. Elgamal, M. Amer, K. Adalier and A. Abul-Fadl, 10/7/93, (PB94-141983, A05, MF-A01).
- NCEER-93-0019 "Development of an Earthquake Motion Simulator and its Application in Dynamic Centrifuge Testing," by I. Krstelj, Supervised by J.H. Prevost, 10/23/93, (PB94-181773, A-10, MF-A03).
- NCEER-93-0020 "NCEER-Taisei Corporation Research Program on Sliding Seismic Isolation Systems for Bridges: Experimental and Analytical Study of a Friction Pendulum System (FPS)," by M.C. Constantinou, P. Tsopelas, Y-S. Kim and S. Okamoto, 11/1/93, (PB94-142775, A08, MF-A02).
- NCEER-93-0021 "Finite Element Modeling of Elastomeric Seismic Isolation Bearings," by L.J. Billings, Supervised by R. Shepherd, 11/8/93, to be published.
- NCEER-93-0022 "Seismic Vulnerability of Equipment in Critical Facilities: Life-Safety and Operational Consequences," by K. Porter, G.S. Johnson, M.M. Zadeh, C. Scawthorn and S. Eder, 11/24/93, (PB94-181765, A16, MF-A03).
- NCEER-93-0023 "Hokkaido Nansei-oki, Japan Earthquake of July 12, 1993, by P.I. Yanev and C.R. Scawthorn, 12/23/93, (PB94-181500, A07, MF-A01).
- NCEER-94-0001 "An Evaluation of Seismic Serviceability of Water Supply Networks with Application to the San Francisco Auxiliary Water Supply System," by I. Markov, Supervised by M. Grigoriu and T. O'Rourke, 1/21/94.
- NCEER-94-0002 "NCEER-Taisei Corporation Research Program on Sliding Seismic Isolation Systems for Bridges: Experimental and Analytical Study of Systems Consisting of Sliding Bearings, Rubber Restoring Force Devices and Fluid Dampers," Volumes I and II, by P. Tsopelas, S. Okamoto, M.C. Constantinou, D. Ozaki and S. Fujii, 2/4/94, (PB94-181740, A09, MF-A02 and PB94-181757, A12, MF-A03).
- NCEER-94-0003 "A Markov Model for Local and Global Damage Indices in Seismic Analysis," by S. Rahman and M. Grigoriu, 2/18/94.

- NCEER-94-0004 "Proceedings from the NCEER Workshop on Seismic Response of Masonry Infills," edited by D.P. Abrams, 3/1/94, (PB94-180783, A07, MF-A02).
- NCEER-94-0005 "The Northridge, California Earthquake of January 17, 1994: General Reconnaissance Report," edited by J.D. Goltz, 3/11/94, (PB193943, A10, MF-A03).
- NCEER-94-0006 "Seismic Energy Based Fatigue Damage Analysis of Bridge Columns: Part I Evaluation of Seismic Capacity," by G.A. Chang and J.B. Mander, 3/14/94, (PB94-219185, A11, MF-A03).
- NCEER-94-0007 "Seismic Isolation of Multi-Story Frame Structures Using Spherical Sliding Isolation Systems," by T.M. Al-Hussaini, V.A. Zayas and M.C. Constantinou, 3/17/94, (PB193745, A09, MF-A02).
- NCEER-94-0008 "The Northridge, California Earthquake of January 17, 1994: Performance of Highway Bridges," edited by I.G. Buckle, 3/24/94, (PB94-193851, A06, MF-A02).
- NCEER-94-0009 "Proceedings of the Third U.S.-Japan Workshop on Earthquake Protective Systems for Bridges," edited by I.G. Buckle and I. Friedland, 3/31/94, (PB94-195815, A99, MF-MF).
- NCEER-94-0010 "3D-BASIS-ME: Computer Program for Nonlinear Dynamic Analysis of Seismically Isolated Single and Multiple Structures and Liquid Storage Tanks," by P.C. Tsopelas, M.C. Constantinou and A.M. Reinhorn, 4/12/94.
- NCEER-94-0011 "The Northridge, California Earthquake of January 17, 1994: Performance of Gas Transmission Pipelines," by T.D. O'Rourke and M.C. Palmer, 5/16/94.
- NCEER-94-0012 "Feasibility Study of Replacement Procedures and Earthquake Performance Related to Gas Transmission Pipelines," by T.D. O'Rourke and M.C. Palmer, 5/25/94, (PB94-206638, A09, MF-A02).
- NCEER-94-0013 "Seismic Energy Based Fatigue Damage Analysis of Bridge Columns: Part II Evaluation of Seismic Demand," by G.A. Chang and J.B. Mander, 6/1/94, (PB95-18106, A08, MF-A02).
- NCEER-94-0014 "NCEER-Taisei Corporation Research Program on Sliding Seismic Isolation Systems for Bridges: Experimental and Analytical Study of a System Consisting of Sliding Bearings and Fluid Restoring Force/Damping Devices," by P. Tsopelas and M.C. Constantinou, 6/13/94, (PB94-219144, A10, MF-A03).
- NCEER-94-0015 "Generation of Hazard-Consistent Fragility Curves for Seismic Loss Estimation Studies," by H. Hwang and J-R. Huo, 6/14/94, (PB95-181996, A09, MF-A02).
- NCEER-94-0016 "Seismic Study of Building Frames with Added Energy-Absorbing Devices," by W.S. Pong, C.S. Tsai and G.C. Lee, 6/20/94, (PB94-219136, A10, A03).
- NCEER-94-0017 "Sliding Mode Control for Seismic-Excited Linear and Nonlinear Civil Engineering Structures," by J. Yang, J. Wu, A. Agrawal and Z. Li, 6/21/94, (PB95-138483, A06, MF-A02).
- NCEER-94-0018 "3D-BASIS-TABS Version 2.0: Computer Program for Nonlinear Dynamic Analysis of Three Dimensional Base Isolated Structures," by A.M. Reinhorn, S. Nagarajaiah, M.C. Constantinou, P. Tsopelas and R. Li, 6/22/94.
- NCEER-94-0019 "Proceedings of the International Workshop on Civil Infrastructure Systems: Application of Intelligent Systems and Advanced Materials on Bridge Systems," Edited by G.C. Lee and K.C. Chang, 7/18/94.
- NCEER-94-0020 "Study of Seismic Isolation Systems for Computer Floors," by V. Lambrou and M.C. Constantinou, 7/19/94, (PB95-138533, A10, MF-A03).

- NCEER-94-0021 "Proceedings of the U.S.-Italian Workshop on Guidelines for Seismic Evaluation and Rehabilitation of Unreinforced Masonry Buildings," Edited by D.P. Abrams and G.M. Calvi, 7/20/94, (PB95-138749, A13, MF-A03).
- NCEER-94-0022 "NCEER-Taisei Corporation Research Program on Sliding Seismic Isolation Systems for Bridges: Experimental and Analytical Study of a System Consisting of Lubricated PTFE Sliding Bearings and Mild Steel Dampers," by P. Tsopelas and M.C. Constantinou, 7/22/94, (PB95-182184, A08, MF-A02).
- NCEER-94-0023 "Development of Reliability-Based Design Criteria for Buildings Under Seismic Load," by Y.K. Wen, H. Hwang and M. Shinozuka, 8/1/94.
- NCEER-94-0024 "Experimental Verification of Acceleration Feedback Control Strategies for an Active Tendon System," by S.J. Dyke, B.F. Spencer, Jr., P. Quast, M.K. Sain, D.C. Kaspari, Jr. and T.T. Soong, 8/29/94.
- NCEER-94-0025 "Seismic Retrofitting Manual for Highway Bridges," Edited by I.G. Buckle and I.F. Friedland, to be published.
- NCEER-94-0026 "Proceedings from the Fifth U.S.-Japan Workshop on Earthquake Resistant Design of Lifeline Facilities and Countermeasures Against Soil Liquefaction," Edited by T.D. O'Rourke and M. Hamada, 11/7/94.
- NCEER-95-0001 "Experimental and Analytical Investigation of Seismic Retrofit of Structures with Supplemental Damping: Part 1 - Fluid Viscous Damping Devices," by A.M. Reinhorn, C. Li and M.C. Constantinou, 1/3/95.
- NCEER-95-0002 "Experimental and Analytical Study of Low-Cycle Fatigue Behavior of Semi-Rigid Top-And-Seat Angle Connections," by G. Pekcan, J.B. Mander and S.S. Chen, 1/5/95.
- NCEER-95-0003 "NCEER-ATC Joint Study on Fragility of Buildings," by T. Anagnos, C. Rojahn and A.S. Kiremidjian, 1/20/95.
- NCEER-95-0004 "Nonlinear Control Algorithms for Peak Response Reduction," by Z. Wu, T.T. Soong, V. Gattulli and R.C. Lin, 2/16/95.
- NCEER-95-0005 "Pipeline Replacement Feasibility Study: A Methodology for Minimizing Seismic and Corrosion Risks to Underground Natural Gas Pipelines," by R.T. Eguchi, H.A. Seligson and D.G. Honegger, 3/2/95.

ł.

L